# Site Audit

# Part 2, 100-120 King Street, Randwick NSW

for Sir Moses Montefiore Jewish Home October 2006

J0807.15R-rev0

CMJA

C. M. Jewell & Associates Pty Ltd

#### Site Audit Part 2, 100-120 King Street, Randwick NSW

October 2006

#### J0807.15R-rev0

Controlled Copy 1	<i>Mr David Freeman</i> Sir Moses Montefiore Jewish Home 120 High Street HUNTERS HILL NSW 2110
2	Ms Fran Mitchell URS Australia Limited Level 3, 116 Miller Street NORTH SYDNEY NSW 2060
3	The General Manager Randwick City Council 30 Frances Street RANDWICK NSW 2031
4	CMJA Library

C. M. Jewell & Associates Pty Ltd ABN: 54 056 283 295 1/13 Kalinda Road, Bullaburra, NSW 2784 PO Box 10, Wentworth Falls, NSW 2782 Phone (02) 4759 3251 Fax (02) 4759 3257 www.cm-jewell.com.au postie@cm-jewell.com.au

Mul

CHRIS JEWELL Auditor

This report has been prepared specifically for Sir Moses Montefiore Jewish Home. This report, the original data contained in the report, and its findings and conclusions remain the intellectual property of C. M. Jewell & Associates Pty Ltd.

This report should not be used by other persons or for other purposes than those for which it was commissioned, and should not be reproduced except in full and with the permission of C. M. Jewell & Associates Pty Ltd. C. M. Jewell & Associates Pty Ltd will not support and will bear no liability for unauthorised use of this report.

# NSW Site Auditor Scheme SITE AUDIT STATEMENT



A site audit statement summarises the findings of a site audit. For full details of the site auditor's findings, evaluations and conclusions, refer to the associated site audit report.

This form was approved under the Contaminated Land Management Act 1997 on 21 February 2005. For more information about completing this form, go to Part IV.

# PART I: Site audit identification

# Site audit statement no: SA183/3

This site audit is a statutory audit/non-statutory audit\* within the meaning of the *Contaminated Land Management Act 1997.* 

Site auditor details (as accredited under the Contaminated Land Management Act 1997)

Name: Christopher Jewell

Address: 1/13 Kalinda Road, Bullaburra, NSW 2784

*Phone*: 4759 3251

Fax: 4759 3257

Site details

Address: 100-120 King Street, Randwick, NSW 2031

#### Property description:

The site is identified as Part 2 of Lot 202 in DP879576, in the Parish of Alexandria, County of Cumberland, as per attached survey plan.

Local Government Area: Randwick

#### Area of site: 2293 m<sup>2</sup>

Current zoning: 2B and 2C Residential

To the best of my knowledge, the site is/is not\* the subject of a declaration, order, agreement or notice under the *Contaminated Land Management Act 1997* or the *Environmentally Hazardous Chemicals Act 1985*.

#### Declaration/Order/Agreement/Notice\* no(s)

.....

\* Strike out as appropriate

C M Jewell and Associates Pty Ltd

# Site audit commissioned by

Name: Mr David Freeman

Company: Sir Moses Montefiore Jewish Home

Address: 120 High Street, Hunters Hill, NSW 2110

Phone: 02 9879 2715

Fax: 02 9871 2700

Name and phone number of contact person (if different from above)

# Purpose of site audit

A. To determine land use suitability (please specify intended use[s])

OR

terre a series de la companya de la company

B(i) To determine the nature and extent of contamination, and/or

B(ii) To determine the appropriateness of an investigation/remedial action/ management plan\*, and/or

·. ....

B(iii) To determine if the land can be made suitable for a particular use or uses by implementation of a specified **remedial action plan/management plan\*** (please specify intended use[s])

Information sources for site audit

Consultancy(ies) which conducted the site investigation(s) and/or remediation

URS Australia Limited

Title(s) of report(s) reviewed:

URS Australia Pty Ltd, Environmental Management Plan, Part 2 Lot 202 King Street, Randwick, dated 21 September 2006.

URS Australia Pty Ltd, Remediation Validation, Part 2 Area Lot 202 King Street, Randwick. Report No. 51072-001-558, Revision 0, dated 5 June 2006.

URS Australia Pty Ltd, Letter Report: Lead Remediation Lot 202 King Street, Randwick – RAP Modification, dated 26 August 2003.

URS Australia Pty Ltd, Remediation Action Plan, Lot 202 King Street, Randwick, dated 7 February 2003.

URS Australia Pty Ltd, Data Assessment Report - Lot 202, King St, Randwick. Report No. 51072-001-R001H dated 5 December 2002.

URS Australia Pty Ltd, Remediation Action Plan, Lot 202 King Street, Randwick. Report No. 51072-001 Rap\_Rev1\_Final dated 26 November 2002.

URS Australia Pty Ltd, Supplementary Sampling and Analysis Plan - Lot 202, King Street,

Randwick. Letter/Report dated 12 June 2002.

\* Strike out as appropriate

Other information reviewed (including previous site audit reports and statements relating to the site)

- Regional geological and topographic mapping
- Regional hydrogeological data

#### Site audit report

Title: Part 2, 100-120 King Street, Randwick NSW

Report no: J0807.15R-rev0

Date: 10 October 2006

\* Strike out as appropriate

C M Jewell and Associates Pty Ltd

# PART II: Auditor's findings

Please complete either Section A or Section B, not both. (Strike out the irrelevant section.)

Use Section A where site investigation and/or remediation has been completed and a conclusion can be drawn on the suitability of land use(s).

Use Section B where the audit is to determine the nature and extent of contamination and/or the appropriateness of an investigation or remedial action or management plan and/or whether the site can be made suitable for a specified land use or uses subject to the successful implementation of a remedial action or management plan.

# Section A

- I certify that, in my opinion, the site is SUITABLE for the following use(s) (tick all appropriate uses and strike out those not applicable):
  - Residential, including substantial vegetable garden-and poultry
  - Residential, including substantial vegetable garden, excluding poultry
  - Residential with accessible-soil, including garden (minimal-home-grown produce-contributing less than 10%-fruit and vegetable intake), excluding poultry
  - Day care centre, preschool, primary school
  - B Residential with minimal opportunity for soil access, including units
  - Secondary school
  - Park,-recreational open space, playing field
  - Gommercial/industrial
  - Other (please specify): A composite development as an aged care and community facility comprising areas of residential use with minimal opportunity for soll access, areas of commercial use (cafes) and areas of landscaped open space as shown on Figure 3 of attached Site Audit Report.

subject to compliance with the following environmental management plan in light of contamination remaining on the site:

Final Report, Environmental Management Plan – Part 2, Lot 202 King Street, Randwick (43346065) dated 21 September 2006.

OR

I certify that, in my opinion, the site is NOT SUITABLE for any use due to the risk of harm from contamination.

#### **Overall comments**

The site is considered suitable for the uses identified above subject to compliance with the Environmental Management Plan identified above. As indicated in the Environmental Management Plan and Site Audit report, a Public Positive Covenant under Section 88E of the Conveyancing Act 1919, requiring management of the site in accordance with the Environmental Management Plan and nominating Randwick City Council as the Prescribed Authority should be made on the relevant land title. It is also recommended that Randwick council note the requirement for observance of the Environmental Management Plan on any Planning Certificate issued in respect of the site under Section 149(5) of the Environmental Planning and Assessment Act 1979. Any soil removed from the site should be appropriately classified in accordance with the Environmental Guidelines: Assessment, Classification and Management of Liquid and Non-Liquid Wastes. Any soil imported to the site should be validated as being suitable for use on the site. Due to regional contamination issues and because groundwater quality may change with time, groundwater should not be extracted for any purpose without appropriate assessment.

#### Section B

Purpose of the plan<sup>4</sup>-which is the subject of the audit-....

I-certify that, in-my opinion:

the nature and extent of the contamination-HAS/HAS NOT\* been appropriately determined

AND/OR

the investigation/remedial-action plan/management plan\* IS/IS NOT\* appropriate for the purpose stated above

#### AND/OR

- the site-CAN BE MADE SUITABLE for the following uses (tick all appropriate-uses and strike out those not applicable):
  - Residential, including substantial vegetable garden-and poultry
  - Residential, including substantial vegetable garden, excluding poultry
  - Residential with accessible soil, including garden (minimal-home-grown produce contributing less than 10% fruit and vegetable intake), excluding poultry
  - Day care centre, preschool, primary school
  - Residential with-minimal opportunity for soil access, including units
  - Secondary school
  - Park, recreational open space, playing field
  - Gommercial/industrial
  - Other (please specify) .....

if-the site is remediated/managed\* in accordance with the following remedial action plan/management plan\* (insert title, date and author of plan)

\_\_\_\_\_

subject-to-compliance with the following condition(s):

<sup>1</sup> For simplicity, this statement uses the term 'plan' to refer to both plans and reports. *Strike out as appropriate* 

C M Jewell and Associates Pty Ltd

# **Overall comments**

# PART III: Auditor's declaration

I am accredited as a site auditor by the NSW Environment Protection Authority under the *Contaminated Land Management Act 1997* (Accreditation No: 9810)

I certify that:

- I have completed the site audit free of any conflicts of interest as defined in the Contaminated Land Management Act 1997, and
- with due regard to relevant laws and guidelines, I have examined and am familiar with the reports and information referred to in Part I of this site audit, and
- on the basis of inquiries I have made of those individuals immediately responsible for making those reports and obtaining the information referred to in this statement, those reports and that information are, to the best of my knowledge, true, accurate and complete, and
- this statement is, to the best of my knowledge, true, accurate and complete.

I am aware that there are penalties under the *Contaminated Land Management Act 1997* for wilfully making false or misleading statements.

Date 10 Ochober Look Signed ...

# PART IV: Explanatory notes

To be complete, a site audit statement form must be issued with all four parts.

#### How to complete this form

**Part I** identifies the auditor, the site, the purpose of the audit and the information used by the auditor in making the site audit findings.

**Part II** contains the auditor's opinion of the suitability of the site for specified uses or of the appropriateness of an investigation, or remedial action or management plan which may enable a particular use. It sets out succinct and definitive information to assist decision-making about the use(s) of the site or a plan or proposal to manage or remediate the site.

The auditor is to complete either Section A or Section B of Part II, not both.

In Section A the auditor may conclude that the land is *suitable* for a specified use(s) OR *not suitable* for any beneficial use due to the risk of harm from contamination.

By certifying that the site is *suitable*, an auditor declares that, at the time of completion of the site audit, no further remediation or investigation of the site was needed to render the site fit for the specified use(s). Any **condition** imposed should be limited to implementation of an environmental management plan to help ensure the site remains safe for the specified use(s). The plan should be legally enforceable: for example a requirement of a notice under the *Contaminated Land Management Act 1997* (CLM Act) or a development consent condition issued by a planning authority. There should also be appropriate public notification of the plan, e.g. on a certificate issued under s.149 of the *Environmental Planning and Assessment Act 1979*.

Auditors may also include **comments** which are key observations in light of the audit which are not directly related to the suitability of the site for the use(s). These observations may cover aspects relating to the broader environmental context to aid decision-making in relation to the site.

In Section B the auditor draws conclusions on the nature and extent of contamination, and/or suitability of plans relating to the investigation, remediation or management of the land, and/or whether land can be made suitable for a particular land use or uses upon implementation of a remedial action or management plan.

By certifying that a site *can be made suitable* for a use or uses if remediated or managed in accordance with a specified plan, the auditor declares that, at the time the audit was completed, there was sufficient information satisfying guidelines made or approved under the CLM Act to determine that implementation of the plan was feasible and would enable the specified use(s) of the site in the future.

For a site that *can be made suitable*, any **conditions** specified by the auditor in Section B should be limited to minor modifications or additions to the specified plan. However, if the auditor considers that further audits of the site (e.g. to validate remediation) are required, the auditor must note this as a condition in the site audit statement.

Auditors may also include **comments** which are observations in light of the audit which provide a more complete understanding of the environmental context to aid decision-making in relation to the site.

In **Part III** the auditor certifies his/her standing as an accredited auditor under the CLM Act and makes other relevant declarations.

# Where to send completed forms

In addition to furnishing a copy of the audit statement to the person(s) who commissioned the site audit, statutory site audit statements must be sent to:

# Department of Environment and Conservation (NSW)

Contaminated Sites Section PO Box A290, SYDNEY SOUTH NSW 1232Fax: (02) 9995 5930

AND

the local council for the land which is the subject of the audit.



1.0	INTR	ODUCTION	.1
	1.1	Background	.1
	1.2	Involvement of Auditor	.1
	1.3	Scope and Structure of Review Report	.2
	1.4	Limitations and Intellectual Property Matters	
2.0	OITE		
2.0	2.1	Site Identification and Location.	
	2.1	Site Setting	
	2.2	Topography and Drainage	
	2.3 2.4		
		Geology	
	2.5	Hydrogeology	
	2.6	Site History	
	2.7	Site Contamination	
	2.8	Contaminants of Concern	
	2.9	Proposed Development	
	2.10	Assessment Criteria	
		2.10.1 Soils	.6
3.0	INVE	STIGATION/S, REMEDIATION AND VALIDATION	.9
	3.1	Assessments Prior to 2002	.9
	3.2	URS, Data Assessment Report (2002)	.9
		3.2.1 Objectives and Scope	.9
		3.2.2 Results	10
		3.2.3 Hibbs & Associates, Asbestos Survey (2002)	10
		3.2.4 URS's Conclusions and Recommendations	
	3.3	URS, Remedial Action Plan (2003)	11
	3.4	URS, Pre-Remediation Works (2003)	11
		3.4.1 Objectives and Scope	11
		3.4.2 Results	12
	3.5	URS, Remediation and Validation (2006)	12
		3.5.1 Objectives and Scope	12
		3.5.2 Results	13
		3.5.3 URS's Conclusions	14
		3.5.4 Auditor's Evaluation of Remediation	14
		3.5.5 Auditor's Evaluation of Validation Results	15
	3.6	Auditor's Evaluation of Adherence to DEC Guidelines	15
	3.7	URS, Environmental Management Plan (2006)	16
4.0		ITOR'S ASSESSMENT OF THE ADEQUACY OF THE CONSULTANT'S WOR	
		DRTING STANDARDS	
AND			
5.0	AUD	ITOR'S ASSESSMENT OF SITE CONDITION	18
	5.1	Risks to Human Health	18
	5.2	Risk to Structures	
	5.3	Risk to the Environment	18
	5.4	Regulatory Compliance	18
	5.5	Potential Contaminant Migration	19
	5.6	Groundwater Issues	
	5.7	Aesthetic Issues and Odours	19
	5.8	Chemical Mixtures	19
	5.9	Long-Term Management	19

6.0	AUD	ITOR'S CONCLUSIONS AND RECOMMENDATIONS	. 19
	6.1	Adequacy of Investigation, Remediation and Validation	. 19
	6.2	Suitability of Site for Proposed Use	. 19
RFF	FRFN	ICES	20

#### Important Information About Your Site Audit

# TABLES

Table 1	Assessment Criteria – Soils
Table 2	Usable Historical Samples
Table 3	Summary Statistics of Validation Data

# FIGURES

Figure 1	Site Location and Setting
Figure 2	Site Layout and Investigation Sample Locations (Lead)
Figure 3	Site Land Use Design and Containment Cell Locations
Figure 4	Lead Delineation Samples – Locations and Concentrations
Figure 5	Sample Locations – Asbestos Validation
Figure 6	Sample Locations – Lead Excavation Validation

# APPENDICES

- Appendix B Communications with the Auditor
- Appendix C Information Relied Upon by the Auditor
  Appendix C1 Summary of Laboratory Results
  Appendix D Data Quality Objectives
  Appendix E Quality Assurance/Quality Control
- Appendix F Environmental Management Plan

# **ASSOCIATED REPORTS**

URS Australia Pty Ltd, Environmental Management Plan, Part 2 Lot 202 King Street, Randwick, dated 21 September 2006.

URS Australia Pty Ltd, Remediation Validation, Part 2 Area Lot 202 King Street, Randwick. Report No. 51072-001-558, Revision 0, dated 5 June 2006.

URS Australia Pty Ltd, Letter Report: Lead Remediation Lot 202 King Street, Randwick – RAP Modification, dated 26 August 2003.

URS Australia Pty Ltd, Remediation Action Plan, Lot 202 King Street, Randwick, dated 7 February 2003.

URS Australia Pty Ltd, Data Assessment Report - Lot 202, King St, Randwick. Report No. 51072-001-R001H dated 5 December 2002.

URS Australia Pty Ltd, Remediation Action Plan, Lot 202 King Street, Randwick. Report No. 51072-001 Rap\_Rev1\_Final dated 26 November 2002.

URS Australia Pty Ltd, Supplementary Sampling and Analysis Plan – Lot 202, King Street, Randwick. Letter/Report dated 12 June 2002.



#### Measures

modouroo	
μg/L	micrograms per litre
km	kilometre
L	litre
	metre
$m m^2$	
	square metre
µS/cm	microsiemens per centimetre
mS/cm	millisiemens per centimetre
mg/kg	milligrams per kilogram
mg/L	megalitres per litre
mm	Millimetre
General	
AHD	Australian Haight Datum
	Australian Height Datum
AMG	Australian Map Grid
ANZECC	Australian and New Zealand Environment and Conservation Council
AST	above-ground storage tank
CLM Act	Contaminated Land Management Act
CMJA	C. M. Jewell & Associates Pty Ltd
COPC	contaminants of potential concern
DA	development application
DEC	Department of Environment and Conservation
DECC	Department of Environment and Climate Change
DLWC	Department of Land and Water Conservation
DNAPL	dense aqueous-phase liquid
DNR	Department of Natural Resources
DP	deposited plan
DQO	data quality objectives
EPA	Environment Protection Authority
ESA	Environmental Site Assessment
HDPE	high-density polyethylene
NATA	National Association of Testing Authorities
NEPM	National Environment Protection Measure
PID	photoionisation detector
	•
PQL	practical quantitation limit
ppmv	parts per million volume
PSH	phase-separated hydrocarbons
QA	quality assurance
QC	quality control
RAP	remediation action plan
RL	relative level
RPD	relative percentage difference
TCLP	Toxicity Characteristic Leaching Procedure
THI	target hazard index
TOC	top of casing
TWA	time weighted average
UCL	upper confidence limit
UST	underground storage tank



# Analytes – Organic

-	+
BaP	benzo(a)pyrene
BTEX	benzene, toluene, ethylbenzene, xylene
OCP	organochlorine pesticides
OPP	organophosphorus pesticides
PAH	polycyclic aromatic hydrocarbons
PCB	polychlorinated biphenyls
SVOC	semivolatile organic compounds
TPH	total petroleum hydrocarbons
VHC	volatile halogenated compounds
VOC	volatile organic compounds

# Analytes – Inorganic

As	Arsenic
Cd	Cadmium
Cr	Chromium
Cu	Copper
Fe	Iron
Hg	Mercury
Mn	Manganese
Ni	Nickel
Pb	Lead
Zn	zinc

# 1.0 INTRODUCTION

## 1.1 Background

This Site Audit Report relates to a portion of land within the property located at 100-120 King Street, Randwick, in New South Wales.

Specifically, the Site Audit relates to the validation work completed by URS Australia Limited (URS) on behalf of Sir Moses Montefiore Jewish Home (and as described in URS's Remediation Validation report dated June 2006).

The Site Audit that this report describes was requested by Mr David Freeman of Sir Moses Montefiore Jewish Home on 27 May 2002, for the purpose of complying with the conditions of the Notice of Determination of Development Application (DA) 02/00551/GI granted by Randwick City Council on 29 October 2002 under the Environmental Planning and Assessment Act 1979. Further conditions were imposed by Council in a letter dated 20 December 2002. In September 2003 Council granted an application to modify the development consent, allowing construction works to proceed within the Part 1 area. The Site Audit is thus a Statutory Site Audit under the provisions of Part 4 of the Contaminated Land Management Act 1997.

The site that is the subject of this audit report is the area designated as 'Part 2' of the overall property. The 'Part 1' area was the subject of a previous audit report and site audit statement, issued on 17 November 2003 by the Auditor, Christopher Jewell.

The audit was conducted for the purpose of determining

- (i) the nature and extent of any contamination of the land,
- (ii) the nature and extent of the investigation or remediation,
- (iia) whether the land is suitable for any specified use or range of uses,
- (iii) the suitability and appropriateness of a plan of remediation, a long-term management plan, a voluntary investigation proposal or a remediation proposal.

The Site Audit Report has been prepared in accordance with the guidelines issued by the NSW Department of Environment and Conservation (DEC) (*Guidelines for the NSW Site Auditor Scheme*, 2nd edition, 2006). It has been prepared by Christopher Jewell, who is a Site Auditor accredited under the NSW Contaminated Land Management Act 1997.

# 1.2 Involvement of Auditor

The Auditor became involved in this project at the start of URS's engagement in 2002, and has had input into the scope and planning of the assessment, remediation and validation works. He had no involvement in work previously undertaken on the site by others.

After reviewing URS's Draft Data Assessment Report (September 2002), the Auditor concluded that residual contamination of the site was such as to preclude development of the site without prior remediation. Accordingly, on 30 September 2002, the Auditor issued Site Audit Statement SA183, indicating that the site was not suitable for any beneficial use owing to risk of harm from contamination. A process of remediation planning, remediation and validation was recommended in the Site Audit Statement. The Auditor reviewed the work undertaken on the area now known as Part 1 and issued Site Audit Statement SA183/2 and a Summary Site Audit Report in November 2003.

The Auditor has visited the site on five occasions to observe and verify, as far as practicable, the site conditions and the progress of the work audited. A compliance checklist has been completed and is held on file.

# 1.3 Scope and Structure of Review Report

Section 2 of this report sets out basic identification and location information concerning the site, and briefly describes the site's topography, geology and hydrogeological setting. An indication of the site's history and an outline of the proposed future use and the associated assessment criteria are also provided. A list of identified contaminants of potential concern is provided in Appendix A.

Section 3 sets out a summary of the environmental assessment/s, remediation and validation undertaken on the site by URS, and includes the Auditor's evaluation of the work's adherence to DEC (2006) guidelines.

Section 4 of this report presents an audit of the completeness and adequacy of the environmental assessments, remediation and validation works that have been completed. The audit was carried out against the criteria established by the NSW DEC publication, *Guidelines for the NSW Site Auditor Scheme* 2nd edition (2006), but also incorporates the reviewer's own judgement; reference has been made to other guideline publications issued or endorsed by the NSW EPA, including *Guidelines for Consultants Reporting on Contaminated Sites* (1997), *Guidelines for Assessing Service Station Sites* (1994), *Sampling Design Guidelines* (1995) and the *National Environmental Protection (Assessment of Site Contamination) Measure* (1999), as appropriate.

Section 5 provides the Auditor's assessment of the site condition, including an assessment of risks to human health, the environmental and structures, groundwater and aesthetic issues and any long-term management that may be required.

Section 6 outlines the Site Auditor's conclusions including the suitability of the site for its intended use.

Throughout this report, extensive use has been made of the site assessment and validation reports prepared by URS; sections of those reports have been adopted for use in this report.

Communications with the Auditor that have ongoing relevance are attached as Appendix B; information relied upon by the Auditor is included in Appendix C.

The use of data quality objectives is detailed in Appendix D; quality assurance/quality control details are provided in Appendix E.

Appendix F provides the Environmental Management Plan developed by URS.

#### 1.4 Limitations and Intellectual Property Matters

This report has been prepared by C. M. Jewell & Associates Pty Ltd for the use of the client and local or state government agencies identified in Section 1.1, for the specific purpose described in that section.

The work has been carried out, and this report prepared, utilising the standards of skill and care normally expected of a site auditor practising in New South Wales under the requirements of the Contaminated Land Management Act 1997. The level of confidence of the conclusions reached is governed, as in all such work, by the scope of the investigation carried out and by the availability and quality of the data. The Auditor has satisfied himself that the available data are adequate to support the conclusions he has reached, and comply with the minimum requirements indicated in the guideline documents specified for the NSW Site Auditor Scheme. Where limitations or uncertainties in conclusions are known, they are identified in this report. However, no liability can be accepted for failure to identify conditions or issues which arise in the future and which could not reasonably have been assessed or predicted using the site information and analytical data available for review.

Data collected by others have, of necessity, been used to support the conclusions of this report. Those data have been subjected to reasonable scrutiny but have essentially, and necessarily, been used in good faith. Liability cannot be accepted for errors in data collected by others where such errors could not have been detected by reasonable scrutiny of the data and supporting information supplied to or requested by the Auditor.

This report, any original data contained in the report, and its findings and conclusions remain the intellectual property of C. M. Jewell & Associates Pty Ltd. A licence to use the report for the specific purpose identified in Section 1.1 is granted to the persons identified in that section on the condition of receipt of full payment for the services involved in the preparation of the report.

It is recommended that this report should not be used by other persons or for other purposes than those identified in Section 1.1 without prior reference to the Auditor. The report must not be reproduced except in full and with the permission of C. M. Jewell & Associates Pty Ltd.

## 2.0 SITE INFORMATION

#### 2.1 Site Identification and Location

The property is located at 100-120 King Street, Randwick, New South Wales, as shown on Figure 1. Australian Map Grid Zone 56H co-ordinates of the centre of the property are approximately 33700E 6246650N. At the date of this report, the site was owned by the Honorary Board of Management of the Sir Moses Montefiore Jewish Home.

The property lies within Randwick local government area and is currently zoned 2B and 2C Residential. The property is identified as Lot 202 in DP879576, in the Parish of Alexandria, County of Cumberland. The property comprises Part 1 and Part 2 for the purposes of the remedial and validation works. The area that is the subject of this Site Audit Report is known as Part 2, and hereafter is referred to as 'the site'. As the site is part of Lot 202, a surveyed plan showing its layout within the property and lot is attached to the Site Audit Statement and is also reproduced on Figure 2.

The overall property is rectangular and has an area of approximately 29,400 m<sup>2</sup>. The site is L-shaped, with an area of 2293 m<sup>2</sup>, and is situated in the property's south-eastern corner and along much of its eastern boundary.

#### 2.2 Site Setting

The site is located in a predominantly residential area, bounded as outlined below.

To the north	The Part 1 area (of the overall property), which is bounded by Govett Lane, then residential properties.
To the east	Dangar Street, then residential properties.
To the west	Part 1 and the Moriah Daycare facility situated in the corner of the property which is bounded by Centennial residential apartments, then a NSW STA bus depot.
To the south	King Street, then residential properties.

Centennial Park is located approximately 300 metres from the site, to its north and west, and Randwick Racecourse is approximately 330 metres to the south-west.

# 2.3 Topography and Drainage

The site is approximately 40 metres above Australian Height Datum (AHD).

Prior to the remediation and development works on Part 1, the property's surface had a gentle slope to the west. A cutting with a concrete retaining wall constructed in the eastern portion of the property divided the property into a higher eastern section and a lower western section.

Demolition works and remedial and bulk excavations have been conducted across the property. During and following these works, sediment, runoff and erosion controls were put in place. These included drainage swales, sediment filters and traps, siltation fences, and sediment/erosion and detention ponds.

The site surface has been levelled in accordance with the specifications for the proposed development, which incorporates engineered stormwater drainage systems.

# 2.4 Geology

Reference to the 1:100,000-scale Sydney Geological Sheet (9130, Edition 1, 1983) indicates that the site is situated on the Holocene sediments of the Botany Basin. Sediment thickness ranges up to 35 metres, but is commonly of the order of 15 metres in the north of the basin. The sediments predominantly consist of well-sorted quartz sands interbedded with minor clay, peat and ironstone lenses. Most of the formation overlies the eroded bedrock surface of the Triassic Hawkesbury Sandstone, consisting of medium to coarse grained quartz with very minor shale and laminate lenses, although moderately to highly weathered horizons of Ashfield Shale have been observed to underlie the Botany Sands in the north and west of the basin.

Site observations during assessment and validation have confirmed the presence of Holocene Sands to the maximum depth of investigation.

Reference to the Sydney 1:100,000 Soils Landscape Series Sheet (9130) indicates that the site is underlain by aeolian soil of the Newport soil landscape. The typical characteristics are shallow (less than 0.5 metre), well-sorted siliceous sands, overlying moderately deep (less than 1.5 metres), buried soils including yellow podzolic soils, with sandy topsoils on crests and gentle slopes. Deep podzols are found on steep slopes, on lower slopes, and in depressions. Limitations of this soil landscape group include very high soil erosion hazard, localised steep slopes, non-cohesive topsoil, and very low soil fertility.

The fill layer across the property varies in depth from 100 millimetres up to several metres. It consists of material similar to the natural underlying material. Below the fill the soils are natural Botany Sands – unconsolidated sand deposits of Quaternary age.

#### 2.5 Hydrogeology

The Quaternary alluvial, estuarine and, in places, aeolian sediments are known locally as the Botany Sands. The Botany Sands host an unconfined aquifer that has in the past been used extensively for water supply purposes. The aquifer is still used for industrial and irrigation purposes.

Groundwater movement within the Botany Sands occurs via primary porosity (i.e. intergranular flow). The hydraulic conductivity of the cleaner sands ranges up to 30 metres per day (m/d), with 10-15 m/d more typical for the clayey and peaty sands in the area. Yields obtained from the Botany Sands are generally moderate to high, usually of the order of 10 litres per second (L/s), although yields of up to 35 L/s have been obtained from the aquifer for industrial purposes.

- - -

4

The groundwater in the Botany Basin is naturally of good and generally potable quality. It has a low salinity, typically less than 200 milligrams per litre (mg/L), although higher concentrations of up to 4000 mg/L occur. The pH of the water is generally low, usually of the order of 5.0 to 6.0 pH units, allowing for the enhanced solubility of most metals in the aquifer, including iron. Currently, abstractions from two large areas of the Botany Sands aquifer are restricted due to the presence of chemical contamination.

Intrusive investigations at the site indicated the presence of groundwater at depths of less than 3 metres. It was assessed that the hydraulic gradient at the site was towards the west, although the regional gradient in the Botany Aquifer is generally towards the south-west.

The nearest receptors of groundwater flowing from the site are likely to be irrigation wells at Randwick Racecourse. At its closest point, Randwick Racecourse is approximately 330 metres southwest of the site.

#### 2.6 Site History

A 1990 review of site history found that the area now comprising Lots 201 and 202 had been used for tram and bus servicing from 1881. Documented activities and uses of the building and land of Lot 202 included the following:

- Foundry
- Stores and air compressor room
- Blacksmith, boiler and welding shop
- Woodworking shop, car body repair shop, trades shop
- Oil and paint store
- Paint shop and bus maintenance garage
- Laboratory
- Timber store
- Bus parking area, washing station, offices and fuelling area
- Mechanics locker room
- · Canteen and recreational building
- Tennis courts and administration building

A remediation program was undertaken between 1991 and 1995, while the depot was operational. In 1997 demolition works were completed, so that Lots 202 and 201 could be subdivided and divested by the then site owner, the NSW State Transit Authority (STA). Further demolition works were conducted in 1997, and in 1998 the current site owner purchased the property. At that time a number of derelict buildings remained on site. The owner leased a portion of the site in the south-west corner to Moriah College, which established a day-care centre for pre-school children.

In 2002, site preparation works were initiated, including demolition of the remaining buildings, and investigation and remediation works. By 2003 the site comprised the day-care centre, a number of partially demolished buildings, and stockpiles of demolition rubble and soil.

In 2003 Council granted permission for construction works on Part 1 of the property. At the date of this report, construction was continuing on this portion of the site.

## 2.7 Site Contamination

On the basis of the site history, the potential sources of site contamination were considered to be:

- petroleum products and paints associated with tram and bus maintenance, storage and refuelling;
- fill;
- stockpiles; and
- asbestos-containing building materials.

#### 2.8 Contaminants of Concern

For the property as a whole, the contaminant groups of concern were identified as:

- heavy metals;
- monocyclic aromatic hydrocarbons, particularly benzene, toluene, ethyl benzene, and xylenes (BTEX);
- total petroleum hydrocarbons (TPH);
- polycyclic aromatic hydrocarbons (PAH); and
- asbestos.

Investigations on the Part 2 area, however, indicated that the contaminants of concern were asbestos and lead.

The individual compounds that make up these contaminant groups are listed in Appendix A.

#### 2.9 Proposed Development

The property is being redeveloped as an aged-care residential facility comprising 277 beds, a synagogue, a function room, a day-care centre and associated car parking.

At the date of this report, construction had commenced on the Part 1 area. The Part 2 area is designated as landscaped open space and paved areas. URS informed the Auditor that at the date of the validation report, the design of the site had not been finalised.

The current design is shown on Figure 3. Any alteration to this design must ensure that the surface areas of the containment cells are either paved or part of a building footprint. Other areas must be either paved, part of a building footprint or landscaped.

# 2.10 Assessment Criteria

The criteria adopted by the Auditor to assess the data contained in URS's reports are listed in Table 1.

#### 2.10.1 Soils

The appropriate soil investigation criteria are the guideline levels set out in Columns 2 and 3 of the table: 'Soil Investigation Levels for Urban Redevelopment Sites in NSW', in the NSW DEC's *Guidelines for the NSW Site Auditor Scheme* 2nd edition (2006).

Criteria derived from Column 2 are health-based soil investigation levels for residential settings where there is minimal opportunity for soil access, originally developed by Imray and Langley in 1996, and currently reissued as Imray and Langley (1999): *Health-Based Soil Investigation Levels, National Environment Protection (Assessment of Site Contamination) Measure* (the NEPM), Schedule B, Guideline 7A. These soil investigation levels are also listed in Column D of Table 5-A, Schedule B(1) of the NEPM.

Criteria derived from Column 3 are health-based soil investigation levels for parks, recreational open space and playing fields including secondary schools, originally developed by Imray and Langley in 1996, and currently reissued as Imray and Langley (1999): *Health-Based Soil Investigation Levels, National Environment Protection (Assessment of Site Contamination) Measure* (the NEPM), Schedule B, Guideline 7A. These soil investigation levels are also listed in Column E of Table 5-A, Schedule B(1) of the NEPM.

Criteria derived from Column 5 – provisional phytotoxicity-based investigation levels (PPBILs) – are usually applied to open-space land use. The Auditor nevertheless accepts URS's and Council's use of Column 3 criteria as appropriate for this development.

The rationale for that decision is that PPBILs are intended to provide screening levels to assess whether adverse impacts on plant growth are likely. Where land may be on-sold for residential or open space use, it is considered appropriate that future owners and users be able to grow the range of plant species that would normally thrive under the climatic and natural soil conditions prevalent in an area.

However, when land it to be used by a known owner for a known purpose for the life of a particular development, it is reasonable for that owner to accept, by consent, some limitation on plant species that may be successfully grown, as in an open space area under professional management, there are many alternative planting options.

Criteria for TPH and BTEX are those published in the NSW EPA's *Guidelines for Assessing Service Station Sites* (1994) and listed in its Table 3 – 'Threshold Concentrations for Sensitive Land Use – Soils'.

With regard to asbestos, advice provided to auditors by the NSW EPA is that 'no asbestos in soil at the surface is permitted'.

TABLE 1 Assessment Criteria – Soils (mg/kg)			
Analyte	DEC Column 2* (NEPM Column D) Residential with Minimal Access to Soil	DEC Column 3* (NEPM Column E) Parks, Recreational Open Space, Playing Fields Including Secondary Schools	Sensitive Land Use – Soils <sup>†</sup>
Metals and Metalloids			
Arsenic (total)	400	200	-
Beryllium	80	40	
Cadmium	80	40	-
Chromium (III)	48%	24%	-
Chromium (VI)	400	200	
Cobalt	400	200	
Copper	4000	2000	-
Lead	1200	600	-
Manganese	6000	3000	
Methyl mercury	40	20	
Mercury (inorganic)	60	30	-
Nickel	2400	600	-
Zinc	28,000	14,000	-
Organics			
Aldrin + Dieldrin	40	20	
Chlordane	200	100	-
DDT + DDD + DDE	800	400	-
Heptachlor	40	20	-
PAHs (total)	80	40	-
Benzo(a)pyrene	4	2	-
Phenol	34,000	1700	
PCBs (total)	40	20	-
Petroleum Hydrocarbon Cor	nponents		•
TPH C6-C9	-	-	65
TPH C <sub>10</sub> -C <sub>40</sub>	-	-	1000
TPH C <sub>16</sub> -C <sub>35</sub> (aromatics) <sup>‡‡</sup>	360	180	-
TPH C <sub>16</sub> -C <sub>35</sub> (aliphatics)	22,400	11,200	-
TPH >C <sub>35</sub> (aliphatics)	224,000	112,000	-
Monocyclic Aromatic Hydro	carbons		
Benzene	-	-	1
Toluene	-	-	1.4‡/130§
Ethylbenzene	-	-	3.1 /50 ¶
Total xylenes	-	-	14**/25¶
Other			
Boron	12,000	6000	
Cyanides (complex)	2000	1000	
Cyanides (free)	1000	500	

Notes: \* NSW DEC (2006)

<sup>##</sup> The carbon number is an 'equivalent carbon number', based on a method that standardises according to boiling point. It is a method used by some analytical laboratories to report carbon numbers for chemicals evaluated on a boiling point GC column.

<sup>†</sup> and notes below: NSW EPA (1994) <sup>‡</sup> The toluene threshold concentration

The toluene threshold concentration is the Netherlands MPC to protect terrestrial organisms in soil. This value is obtained by applying a US EPA assessment factor to terrestrial chronic No Observed Effect Concentration (NOEC) data. The MPC is an 'indicative' value (Van de Plassche et al. 1993; Van de Plassche & Bockting 1993).

Ituman health and ecologically based protection level for toluene. The threshold concentration presented here is the Netherlands intervention value for the protection of terrestrial organisms. Other considerations such as odours and the protection of groundwater may require a lower remediation criterion.

The ethylbenzene threshold concentration is the Netherlands MPC for the protection of terrestrial organisms in soil. No terrestrial ecotoxicological data could be found for use in the Netherlands criteria derivation. Therefore, equilibrium partitioning has been applied to the MPC for water to obtain estimates of the MPC for soil. The MPC for water has been derived from aquatic ecotoxicological data (Van de Plassche et al. 1993; Van de Plassche & Bockting 1993).

<sup>1</sup> Human health based protection level for ethylbenzene or total xylenes as shown. The threshold concentration presented here is the Netherlands intervention value. Other considerations such as odours and the protection of groundwater may require a lower remediation criterion.

The xylene threshold concentration is the Netherlands MPC for the protection of terrestrial organisms in soil. No terrestrial ecotoxicological data could be found for use in the Netherlands criteria derivation. Therefore, equilibrium partitioning has been applied to the MPC for water to obtain an estimate of the MPC for Water has been derived from aquatic ecotoxicological data. The concentration shown applies to total xylenes and is based on the arithmetic average of the individual xylene MPCs (Van de Plassche et al. 1993; Van de Plassche & Bockting 1993).

#### 3.0 INVESTIGATION/S, REMEDIATION AND VALIDATION

#### 3.1 Assessments Prior to 2002

At the Auditor's request, URS evaluated the quality and usability of pre-existing data at the beginning of its involvement. Only data of acceptable quality were to be included in the data set used by URS.

URS reviewed the following reports:

- Sinclair Knight Partners (SKP, 1991) Preliminary Geotechnical and Environmental Investigation of the State Transit Authority's Randwick Bus Depot;
- Dames & Moore (D&M, 1991) Feasibility/Design Study for a Site Remediation Program at the Randwick Bus Depot;
- D&M (1992) EPA Compliance Report, Site Remediation Program: Phase I, Randwick Bus Depot;
- D&M (1994) EPA Compliance Report, Initial Site Validation Program, Randwick Bus Depot;
- D&M (1995) EPA Compliance Report, Final Site Validation Program, Randwick Bus Depot; and
- D&M (1998) EPA Compliance Report Post Demolition Site Validation Program Lot 202 King and Dangar Streets Randwick.

URS reviewed the data presented in these reports, concluding that all data prior to and including 1994 were unusable for reasons such as lack of laboratory certificates and inadequate field and/or laboratory quality assurance/quality control (QA/QC).

Table 2 shows details of historical data (1995–1998) that URS considered suitable for incorporation into the data set used to assess the property (Parts 1 and 2).

TABLE 2 Usable Historical Samples	
Analytes	No. of Retained Samples
Heavy metals	30
TPH/BTEX	54
TPH only	7
PAHs	10

# 3.2 URS, Data Assessment Report (2002)

#### 3.2.1 Objectives and Scope

The stated objective was to:

Combine the results of recent investigations (July, August and September 2002) and historic investigations (1995 and 1998) in order to characterize the site condition for review by the site auditor and preparation of a SAS.

In June 2002, before the property was divided into Parts 1 and 2, URS developed a sampling and analysis plan for the additional assessment works at the site, with the objective of filling the data gaps identified in the above review. The plan was submitted to the Site Auditor for approval. The selected contaminants of concern were asbestos, metals, TPH/BTEX and PAHs. While historical sampling data could be retained, the intention of the supplementary works was to achieve better site coverage.

The scope of work for the overall property comprised:

- the excavation of twenty-eight test pits (identified as TP1 to TP28) across the property on a grid pattern, with an approximate spacing of 40 metres;
- the installation of four groundwater monitoring wells;
- investigation and sampling of the stockpiles present (total volume of 9000 m<sup>3</sup>);
- analysis of selected samples for the contaminants of concern.

Four sample locations from the 1995 investigation and two test pits (TP02 and TP06) were within the site that is the subject of this audit (Part 2). These locations are shown on Figure 2.

# 3.2.2 Results

#### Soils

Analytical results for samples obtained within the site indicated that the concentrations of contaminants of concern were within the assessment criteria, with the exception of lead in sample TP02/0.2\_0.4 (1880 mg/kg). Asbestos was not detected in any of the test pit samples.

Other relevant results are shown below.

#### Dames & Moore 1995 Investigation

• Analytical results for samples Z7-2, Z7-3, Z7-4 and Z7-8 showed that lead concentrations exceeded the PQL in two samples (one at 5 mg/kg and the other at 6 mg/kg).

#### URS 2002 Investigation

- The sole TP06 sample was not tested for metals; TPH fractions and PAH analytes were all below the PQL.
- In TP02\_0.2-0.4, the total PAH concentration was 8.2 mg/kg.

The summary of laboratory results for these samples (for lead only) is included with the delineation and validation data provided in Appendix C1.

#### Groundwater

Four wells were installed at the property during the URS investigation. None of these were within the Part 2 area, and so the results were relevant to the investigation, remediation and validation of the Part 1 area. In summary, significant concentrations of TPH were detected in samples from three wells. The groundwater remediation work was described in URS's validation report and in the Summary Site Audit Report for Part 1.

#### 3.2.3 Hibbs & Associates, Asbestos Survey (2002)

In 2002, Hibbs & Associates Pty Ltd (Hibbs) undertook an asbestos inspection of the property. The subsequent letter report was included as an appendix to the URS 2002 report described above.

The survey identified the following issues relating to the site that is the subject of this report:

- Fragments of bonded asbestos cement sheeting were noted at various locations around the site including within the assessment zones No. 1, 4, 5 . . . .[part of Zone 1 was within the area subject to this audit]
- Some demolition rubble mixed with soil was spread across the eastern area of the site (grid location A3 – A5) adjacent to the site boundary [within the site that is the subject of this report]. Cement bonded asbestos fragments were observed in this material.

The asbestos impact identified within the site (Part 2), was an area approximately 50 by 15 metres along the boundary of Parts 1 and 2, parallel to Dangar Street. The area is shown on Figure 5.

#### 3.2.4 URS's Conclusions and Recommendations

URS's conclusions and recommendations largely concerned the contamination and remediation of the Part 1 area and the groundwater. The lead concentration within the sample from TP02 was not considered to represent a hotspot, as it did not exceed 250 per cent of the guideline criterion adopted at that time (from Column 2 – residential with minimal access to soil).

With regard to asbestos, URS recommended manual removal of fragments, with a follow-up inspection by Hibbs.

#### 3.3 URS, Remedial Action Plan (2003)

The Remedial Action Plan (RAP) produced by URS outlined the proposed works for the whole property. The works relevant to the site that is the subject of this site audit report are described below.

#### Lead Remediation Program

Because the lead concentration identified in the sample from TP02 exceeded 250 per cent of the newly adopted criterion (from Column 3 - parks, recreational open space), the location was considered a hotspot, requiring remediation. The proposed works involved excavation of the area, off-site disposal of the soil, and collection of validation samples from the excavation.

#### Asbestos Remediation Program

To remediate the asbestos-impacted areas identified during the 2002 inspection, URS proposed the following.

- Excavation, removing a minimum of 300 millimetres of the surface soil from the impacted areas, and off-site disposal.
- A visual inspection of the scraped surfaces. If the surface was deemed clear of asbestos, validation sampling was to be conducted on a 20-by-20-metre grid.

The works were to be conducted under the supervision of an environmental consultant and an asbestos specialist, using appropriate environmental and occupational health and safety controls.

Following the delineation works described below in Section 3.4, a 'RAP modification' letter report was produced, outlining the chosen remedial option for the site. This involved excavation of the soil where exceedances of the Column 3 criterion for lead were identified. This material was to be placed in containment cells beneath areas designated for building or paving footprints, and capped. Validation sampling of the walls and base of the excavated area was to be conducted on a 10-by-10-metre grid basis.

#### 3.4 URS, Pre-Remediation Works (2003)

#### 3.4.1 Objectives and Scope

Between 12 and 24 April 2003, during the remediation of Part 1, URS conducted additional works in order to further delineate the extent of the hydrocarbon impact (in the Part 1 area) and the lead impact (in the Part 2 area). The hydrocarbon delineation is not relevant to the site that is the subject of this audit, and it will therefore not be summarised or reviewed.

Delineation sampling of the area previously identified as lead-affected (TP02) was carried out at thirty-two locations (identified as TP02, HA201 to HA209, TP201 to TP212, TP301 to TP304, TP306, TP307, TP309, and TP311 to TP313). Some of these locations extended into the Part 1 area. A total

of eighty-six samples collected from these sampling locations were analysed for lead; their locations and the analytical results are shown on Figure 4.

#### 3.4.2 Results

Laboratory analysis detected numerous exceedances of the adopted criteria for lead within the site; fourteen of DEC Column 3 (600 mg/kg) and eight of the less stringent Column 2 (1200 mg/kg). The impact was associated with a strip of fill material adjacent to Dangar Street, and also with fill material sourced on site and placed in the area of a building footprint at depths up to 1.0 metre.

A summary of laboratory results is provided in Table 1 of Appendix C1.

Because these works were described in the validation report, together with results, the Auditor's evaluation of adherence to DEC guidelines is presented in Section 3.6.

#### 3.5 URS, Remediation and Validation (2006)

During 2003 the site was divided into two parts for the purpose of remediation and validation. Development of the larger Part 1 could then be initiated before remediation of Part 2 had been completed.

#### 3.5.1 Objectives and Scope

URS outlined the objectives of the Part 2 remediation as follows:

- To make Part 2 of the site suitable for the following proposed landuses;
  - Paved areas, which are considered, for the purpose of this report, as residential use with minimal opportunities for access to soil; and
  - Landscaped areas surrounding the building and paved areas, which are considered, for the purpose of this report, as open space land areas.
- To protect human health (residents, site workers, off-site workers and off-site residents) and the environment from hazards relating to remediation works;
- To demonstrate that lead and asbestos impacted soil identified during environmental assessments are remediated to a level appropriate for the proposed land use of the site;
- To comply with regulatory and legislative requirements; and
- To comply with RCC [Council] Contaminated Land policy, 1999.

The remediation methodology for the site was detailed in the RAP (URS, February 2003) and the RAP Modification (URS, August 2003). The proposed and actual remedial and validation works are summarised below.

#### Asbestos-Impacted Materials

The preferred and undertaken strategy was the excavation and off-site disposal of the fill and building materials containing asbestos. The extent of the excavation was to be based on visual observations. Appropriate environmental and occupational health and safety controls, including air monitoring, were implemented during the works. Validation sampling was conducted within the excavation.

Hibbs oversaw the removal and disposal of asbestos pieces and impacted material, and validation of the impacted area. Its report documenting the works and providing clearance certificates was provided as an appendix to the URS report, and is reproduced as Appendix C2 of this site audit report. The area within the site, described by Hibbs as 'the footprint of the former building located between grid co-ordinates A2 and A5', is shown on Figure 5.

Demolition fill material containing asbestos was excavated to a depth of between 300 millimetres and 1.0 metre along the eastern site boundary. The volume of material was approximately  $600 \text{ m}^3$ . The excavation extended beneath the fill to the natural Botany Sands.

The asbestos-containing material was transported off site to Penrith Waste Services, and documentation provided to the Auditor.

#### Lead-Impacted Materials

The remediation strategy for the delineated lead-impacted area involved excavating the material and collecting validation samples from the walls and base of the excavation. The proposed remediation area is shown on Figure 4.

The area identified as impacted included all locations from which samples had exceeded one or both adopted criteria, with two exceptions: samples from TP206 and TP202, at depths of 0.5-0.6 metre below ground level, with concentrations of 2330 mg/kg and 1390 mg/kg respectively. Remediation was not considered necessary at these two locations because of their proximity to the edge of a proposed building, which rendered the soils effectively inaccessible. The Column 2 criterion (1200 mg/kg) was therefore appropriate, and the concentrations were less than 250 per cent of that criterion.

The excavated area was roughly rectangular, as shown on Figure 6. Its final dimensions were approximately 49 metres in length by a maximum width of 9 metres; the depth (intruding into Botany Sands) ranged from 0.9 metre to 1.3 metres.

The project managers confirmed that, at the date of this site audit report, the excavated area had been reinstated during additional excavation and backfilling works conducted at the site as part of site development.

#### **Containment Cells**

The excavated material was placed in two prepared containment cells on site, together with leadimpacted material excavated from Part 1. The cells were lined with a geo-fabric marker layer, filled with the contaminated material, and covered with another geo-fabric layer. Cell depth was a minimum of 40 metres above AHD, ensuring that cells were above the groundwater table.

Groundwater monitoring was undertaken by Jeffrey and Katauskas. The August 2003 and July 2003 monitoring reports indicated that the groundwater level in the south-east of the property was approximately 38 metres above AHD. Cell locations are shown on Figure 3.

URS confirmed that the containment cells would be capped with 100-millimetre-thick non-reinforced concrete. At the date of the validation report, URS understood that paving would be laid on top of these concrete slabs.

The Auditor inspected the cells following completion and confirmed the construction to be as specified by URS.

#### 3.5.2 Results

Asbestos was not detected in any of the four samples collected from the asbestos-impacted area. Sample locations are shown on Figure 5.

To validate the excavated lead-impacted area, twenty-five samples were collected on a 10-by-10-metre grid basis. Sample locations are shown on Figure 6. The laboratory results for lead were assessed against the Column 3 criterion (600 mg/kg). Lead concentrations in samples VAL200, VAL203,

VAL209, VAL210 and VAL220 exceeded the criterion, at 2420, 1470, 2270, 650 and 1440 mg/kg respectively, and the excavations at those locations were extended (see Figure 6).

Lead concentrations in samples VAL206, VAL211 and VAL222 (all from the western side of the excavation) exceeded the criterion, at 602, 655 and 902 mg/kg respectively. These concentrations were less than 2.5 times the criterion.

Two significant lead exceedances were detected in samples from the eastern excavation wall – VAL207 (3990 mg/kg) and VAL204 (3790 mg/kg). Chasing out was not possible, however, given that the sample locations were only 200 millimetres from the property boundary and a footpath.

The data sets presented by URS within the validation report included:

- analytical results for the lead delineation samples described in Section 3.4 (URS's Table 1);
- results for lead excavation validation samples (URS's Table 2a);
- data from the remaining Stage 2 area (URS's Table 2b);
- results for the impacted soil that was excavated and placed within the containment cells (URS's Table 3); and
- the asbestos validation results (URS's Table 4).

These tables are provided in Appendix C1.

URS conducted a statistical analysis of the validation sampling, using a 95 per cent upper confidence limit (UCL) of the mean concentrations. The data set excluded the exceedances from the eastern boundary (VAL201, VAL204, VAL207, VAL214 and VAL217) because further excavation was not possible and they were considered to be 'representative of the boundary conditions'. Samples from the western boundary (TP202 and TP206 samples) were also excluded from the data set for the reasons stated in Section 3.5.1 (application of different criterion). The 95 per cent UCL was calculated to be 305.3 mg/kg, meeting the adopted criterion for lead of 600 mg/kg (Column 3).

#### 3.5.3 URS's Conclusions

URS stated :

As a consequence of the remediation and the validation program . . . URS considers that Part 2 of Lot 202 . . . is suitable for the land use as paved or built areas with minimal opportunity access to soils and open space landscaped area in accordance with the current DEC guidelines. The physical definition of the two land use areas is presented in [Figure 3 of this report].

A post remediation Environmental Management Plan (EMP) will be prepared to address future management of Part 2 of the Site with regard to placement of restrictions on excavation in the containment areas such that they are undertaken in accordance with specific requirements.

URS produced an EMP dated 21 September 2006, which is discussed further in Section 3.7 below.

#### 3.5.4 Auditor's Evaluation of Remediation

The Auditor was consulted about the proposed remediation prior to implementation and considered it to be technically feasible and environmentally justifiable.

Although the encapsulation option is not high on the DEC-endorsed ANZECC – NHMRC preferred order of remediation options, it is a reasonable approach for a highly managed site such as this. Neither on-site nor off-site treatment is applicable to relatively small volumes of lead-contaminated soil, so the only available alternative would have been landfill disposal.

The strategy proposed and implemented does ensure long-term stability of the capping (by use of reinforced concrete) and similarly minimises the potential for leachate formation. The lead contamination is not volatile.

No structures will be erected on the capped area, and a management plan will be implemented.

#### 3.5.5 Auditor's Evaluation of Validation Results

Summary statistics of the validation data are provided in Table 3. This data set includes validation data from the lead excavation and investigation data from all sampling locations that remain on site, with the exception of those on the eastern boundary (VAL201, VAL204, VAL207, VAL214 and VAL217) and those on the western boundary (TP202 and TP206).

TABLE 3 Summary Statistics of Validation Data (mg/kg)									
Analyte		PQL	Set	Min.	Max.	Mean	Standard Deviation	C.V	95% UCL <sup>*</sup>
Lead		1	32	<pql< td=""><th>902</th><td>227</td><td>254</td><td>1.1</td><td>303</td></pql<>	902	227	254	1.1	303
Notes:	PQL Set Min. Max. Mean C.V. UCL	laboratory practical quantitation limit number of samples in data set minimum concentration maximum concentration arithmetic mean coefficient of variation (standard deviation/mean) upper confidence limit calculated via US EPA Non-parametric Jack-knife							

The Auditor has reviewed the validation data. He notes that the lead concentrations are very variable. There were three exceedances of the 600 mg/kg validation criterion. The 95% UCL is well below the criterion, but the standard deviation of the concentrations is high – greater than the mean, but nevertheless less than half the criterion.

#### 3.6 Auditor's Evaluation of Adherence to DEC Guidelines

This evaluation has been conducted using the information and data presented in URS's validation report, which incorporates investigation sampling within the site area and validation sampling.

#### Data Quality Objectives (DQOs)

URS appropriately adopted the data quality objectives (DQO) process endorsed by DEC (2006).

A detailed DQO checklist is included as Appendix D of this site audit report.

#### Data Quality Indicators (DQIs)

Data quality indicators (DQIs) were not used to assess field procedures and analytical results, but much of the information required in a DQI assessment was provided in another form.

This part of the auditing checklist is included as Appendix E.

#### **QA/QC** Evaluation

The field and laboratory QA/QC measures presented have been reviewed and are considered to comply with DEC guidelines and to be adequate to ensure the integrity of the data set used to assess the site.

Specifically, a detailed QA/QC checklist is provided in Appendix E.

The QA/QC criteria list examined in this review included:

- Precision
- Accuracy
- Sensitivity
- Representativeness
- Comparability
- Completeness
- Holding times
- Blanks

URS adopted and described an appropriate sampling plan, sample handling, and sample collection and transport processes.

URS's sampling procedures, as outlined in the reports, have been reviewed. These procedures are considered to substantially comply with DEC guidelines and to be adequate to ensure the integrity of the data set used to assess contamination on this site. A QC sampling issue was noted: wash and trip blanks were used, but not at an acceptable frequency.

URS collected blind intra-laboratory duplicates at the standard frequency of 1 in 10. Two significant RPD outliers were attributed to the heterogeneous nature of the lead-impacted material. Interlaboratory duplicates were collected, but not at the required frequency.

Also reviewed were laboratory QA/QC procedures and results. The NATA-accredited laboratory identified methods used, and provided satisfactory reporting limits. QA/QC procedures comprised method blanks, matrix spikes and calculated recoveries, laboratory control samples, duplicates with RPDs calculated, and surrogates.

The review confirmed that targeted frequencies and control limits were met, that method blanks were free of contamination, and that duplicate RPDs were within control limits.

The Auditor considers that the overall quality of data and their presentation are of an adequate standard to support the conclusions he has reached.

#### 3.7 URS, Environmental Management Plan (2006)

The objective of the Environmental Management Plan (EMP) is

to provide a framework for the management of contaminated materials that have been retained on the site following the completion of remediation and validation work.

The EMP is required to provide a framework for monitoring and managing the containment cells into the future, and as a reference for future activities undertaken on the site that may expose the impacted materials within the containment cells. Montefiore (the site owner) is responsible for implementing and adhering to the EMP.

It is noted that the EMP is intended to address minor maintenance works only. Construction or excavation work requiring planning consent is not covered, and would require the preparation of a more detailed, specific EMP.

URS made the following recommendations:

- a Public Positive Covenant under Section 88E of the Conveyancing Act 1919 is made to the property title, with Randwick Council as the prescribed authority;
- that Council place a notation on the Planning Certificate (Section 149 (2) certificate), addressing the restrictions relating to works on the site, and
- that, where a DA is submitted, Council enforce implementation of the EMP through the conditions of consent.

The Auditor has reviewed the EMP and endorses those recommendations.

The EMP outlined procedures and protocols for management of any future minor intrusive works conducted at the site. URS stated:

Procedures for current and future site occupants/workers to ensure contact with the contaminated fill and/or soils during maintenance of the site is in accordance with suitable occupational health and safety (OH&S) environmental controls.

This will include the identification of potential occupant/worker exposure pathways and methods for minimising occupant/workers exposure to the contaminated fill and/or soils.

The Auditor's review indicates that the EMP is consistent with the requirements of *Guidelines for the NSW Site Auditor Scheme*, 2nd edition (2006).

The EMP is provided as Appendix F of this report.

## 4.0 AUDITOR'S ASSESSMENT OF THE ADEQUACY OF THE CONSULTANT'S WORK AND REPORTING STANDARDS

The Auditor has assessed the adequacy of URS's work and the assessment and validation reports.

The following information was provided and considered to be adequate for the purposes of this audit:

- site location and description of site
- · review of site history, including potentially contaminating activities
- identification of potential contaminants of concern
- description of soil stratigraphy and hydrogeology
- discussion of groundwater issues
- outline of actual or potential contamination
- investigation and remediation works
- quality assurance and quality control plan
- discussion of analytical results
- environmental quality criteria
- assessment of risks to human health
- recommendations for ongoing management of residual contamination
- recommendations and conclusions

The following information was not provided, but was not considered relevant by the Auditor, given the results of the investigation:

- assessment of chemical mixtures
- discussion of evidence of migration of contaminants
- assessment of aesthetic issues

Overall the standard of reporting presented within URS's report is considered satisfactory and to comply with the NSW EPA's *Guidelines for Consultants Reporting on Contaminated Sites* (1997) and the NSW DEC's *Guidelines for the NSW Site Auditor Scheme* 2nd edition (2006).

The Auditor considers that there has been compliance with requirements imposed by the planning consent authority.

#### 5.0 AUDITOR'S ASSESSMENT OF SITE CONDITION

#### 5.1 Risks to Human Health

The assessment and validation of site soils and groundwater indicated that all identifiable contamination risks to human health remaining at the site will be effectively managed under the proposed management plan.

The remediation and validation of the site ensured that the contaminants of concern (namely asbestos and lead) were removed from the site or appropriately contained, in order to ensure that the risk to human health is minimal. The EMP was designed to ensure ongoing management of the impacted soils held within the containment cells, and appropriate management of any exposure, in order to reduce the potential risk to human health.

#### 5.2 Risk to Structures

The assessment of site soils and groundwater indicated that there are no identifiable contamination risks to structures at the site.

# 5.3 Risk to the Environment

The assessment of site soils and groundwater indicated that there are no identifiable contamination risks to the environment at the site.

#### 5.4 Regulatory Compliance

The remediation of the site was defined as Category 2 under SEPP 55. As required under this regulation, notice was provided to Council at least 30 days before remediation commenced. Council responded with several requirements, which were addressed in the RAP outlined in Section 3.3.

After delineation works had been completed, URS submitted its RAP modification letter report to Council. In response, Council's letter of 5 September 2003 imposed a number of additional conditions relating to adherence to the conditions of consent dated 22 October 2002, and development of an Auditor-approved EMP addressing management of the containment cells. Specifically, Council stipulated that the Site Audit Statement should include a reference to the EMP and the covenant on the title.

Other regulatory requirements identified by URS and addressed in the RAP (2003) were those imposed by Council's Contaminated Land Policy (1999). They included environmental controls;

access and signage requirements; OH&S measures; the requirement for Category 2 notifications; provision of investigation and RAP reports prior to remediation; and provision of a validation report.

## 5.5 Potential Contaminant Migration

The Auditor does not consider that the potential for off-site migration of contamination originating from the site is an issue of concern because remaining lead-impacted soils have been contained in accordance with regulations and current industry-endorsed practice. As long as the EMP is adhered to, the risk of leaching or migration of the contaminated contained material is low.

#### 5.6 Groundwater Issues

The Auditor considers that groundwater issues have been effectively addressed and are no longer of concern on the site.

#### 5.7 Aesthetic Issues and Odours

The Auditor does not consider that aesthetic issues or odours are of concern on this site.

#### 5.8 Chemical Mixtures

On the basis of the data that he has reviewed, the Auditor does not consider that the potential for chemical mixtures to be present is an issue of concern on this site.

# 5.9 Long-Term Management

The EMP developed by URS (as outlined in Section 3.7 and provided in Appendix F) adequately addresses long-term management of the impacted soil remaining on site within the containment cells.

#### 6.0 AUDITOR'S CONCLUSIONS AND RECOMMENDATIONS

#### 6.1 Adequacy of Investigation, Remediation and Validation

The Auditor considers that the investigation, remediation and validation works were adequate and conducted substantially in accordance with the appropriate guidelines, as demonstrated earlier in this Site Audit Report.

# 6.2 Suitability of Site for Proposed Use

The Auditor considers that the site has been validated to the required standard, and that analysis of validation samples demonstrates that concentrations of contaminants of concern remaining on the site are within the criteria applicable to residential use with minimal access to soil with open space areas or are contained on site in such a way as not to impact upon that use.

The Auditor has thus concluded that it is appropriate to issue a Site Audit Statement which indicates that the site is suitable for a composite development as an aged-care and community facility including areas of residential use with minimal opportunity for soil access, areas of semi-commercial use as publicy-accessible cafes and areas of open space use, as indicated on Figure 3 of this report, subject to a condition requiring implementation of the environmental management plan that is attached as Appendix F of this report, and subject to the adoption of measures to ensure that the environmental management plan continues to be applied in the future, even if there are successive changes in ownership of the land, or changes to the land title details. It is also appropriate to ensure that potential purchasers of the land are aware of these requirements.

These requirements will be met by placing a Public Positive Covenant under Section 88E of the Conveyancing Act 1919, on the land title. The covenant should require that the site be managed in accordance with the EMP, and should nominate Randwick City Council as the Prescribed Authority.

# REFERENCES

**ANZECC 2000,** *Australian and New Zealand Guidelines for Fresh and Marine Water Quality.* National Water Quality Management Strategy, Australian and New Zealand Environment and Conservation Council/Agriculture and Resource Management Council of Australia and New Zealand.

**ANZECC 1992,** Australian and New Zealand Guidelines for the Assessment and Management of Contaminated Sites, Australian and New Zealand Environment and Conservation Council/National Health and Medical Research Council.

**Imray, Paula and A. Langley 1999,** *Health-Based Soil Investigation Levels, National Environment Protection (Assessment of Site Contamination) Measure, Schedule B, Guideline 7A.* 

**NSW DEC 2006**, *Guidelines for the NSW Site Auditor Scheme*, 2nd edition, NSW Department of Environment and Conservation, Sydney NSW.

**NSW EPA 1994,** *Guidelines for Assessing Service Station Sites,* NSW Environment Protection Authority, Chatswood NSW.

NSW EPA 1995, *Sampling Design Guidelines*, NSW Environment Protection Authority, Chatswood NSW.

**NSW EPA 1997**, *Guidelines for Consultants Reporting on Contaminated Sites*, NSW Environment Protection Authority, Chatswood NSW.

**NSW EPA 1999,** Environmental Guidelines: Assessment, Classification and Management of Liquid and Non-Liquid Wastes, NSW Environment Protection Authority, Chatswood NSW.

**Taylor, Roscoe and A. Langley 1999,** *Exposure Scenarios and Exposure Settings, National Environment Protection (Assessment of Site Contamination) Measure, Schedule B, Guideline 7B.* 

Van de Plassche, E. J. and G. J. M. Bockting 1993, *Towards Integrated Environmental Quality Objectives for Several Volatile Compounds*. National Institute of Public Health and Environmental Protection, Bilthoven, The Netherlands, Report No. 679101 011.

Van de Plassche, E. J., M. D. Polder and J. H. Canton 1993, *Derivation of Maximum Permissible Concentrations for Several Volatile Compounds for Water and Soil.* National Institute of Public Health and Environmental Protection, Bilthoven, The Netherlands, Report No. 679101 008.



# Important Information About Your Site Audit Report

These notes will help you to interpret your Site Audit report. They are based on guidelines prepared by the NSW Department of Environment and Conservation.

# Introduction to the NSW Site Auditor Scheme Objectives

The objectives of the NSW Site Auditor Scheme are to:

- ensure that public health and the environment are protected through proper management of contaminated sites, particularly during changes of land use
- improve access to technical advice on contaminated sites for planning authorities and the community by establishing a pool of accredited site auditors
- provide greater certainty for planning authorities and the community through the independent review by those auditors of contaminated site assessment and remediation reports, and reports that validate the successful completion of the assessment of remediation.

#### Background

In Australia, the use of accredited auditors to review work conducted by contaminated site consultants was first introduced in Victoria in 1989 through the Victorian EPA's Environmental Auditor (Contaminated Land) Scheme.

In 1998, NSW commenced its own Site Auditor Scheme under the *Contaminated Land Management Act 1997* (CLM Act). The scheme is administered by the Department of Environment and Conservation (DEC).

The CLM Act empowers DEC to accredit individuals as site auditors and to establish guidelines for them.

The Contaminated Land Management Regulation 1998 (CLM Regulation) specifies some of the procedural requirements of the scheme.

#### Site Audits in Relation to Contaminated Sites

Site auditors review the work of contaminated site consultants. The CLM Act calls these reviews 'site audits' and defines a site audit as an independent review:

a) that relates to investigation or remediation carried out (whether under the CLM Act or otherwise) in respect of the actual or possible contamination of land, and

- b) that is conducted for the purpose of determining any one or more of the following matters:
  - i) the nature and extent of any contamination of the land
  - ii) the nature and extent of the investigation or remediation
  - iii) whether the land is suitable for any specified use or range of uses
  - iv) what investigation or remediation remains necessary before land is suitable for any specified use or range of uses
  - v) the suitability and appropriateness of a plan of remediation, a long-term management plan, a voluntary investigation proposal or a remediation proposal.

The main products of a site audit are a 'site audit statement' and a 'site audit report'.

A **site audit statement** is the written opinion by a site auditor, on a DEC-approved form, of the essential findings of a site audit. It includes, where relevant, the auditor's conclusions regarding the suitability of the site for its current or proposed use.

Before issuing a site audit statement, the site auditor must prepare and finalise a detailed **site audit report**. The report must be clearly expressed and presented and contain the information, discussion and rationale that support the conclusions in the site audit statement.

In some circumstances a site audit is required by law. These audits are known as '**statutory site audits**' and may be carried out only by site auditors accredited under the CLM Act. A statutory site audit is one that is required by:

- a regulatory instrument issued under the CLM Act, including DEC agreements issued by DEC to voluntary proposals.
- the *Environmental Planning and Assessment Act 1979*, including an environmental planning instrument or development consent condition
- any other Act.

# Role of Site Auditors

The services of a site auditor can be used by anyone who needs an independent and authoritative review of information relating to possible or actual contamination of a site. The review may involve independent expert technical advice or 'sign-off' of contaminated site assessment, remediation or validation work conducted by a contaminated site consultant.

#### Site Assessment and Audit Process

The usual stages in the assessment, remediation and validation of a contaminated site, and in the audit of those activities, are as follows:

#### Consultant is Commissioned to Assess Contamination

In most cases, a site owner or developer engages a contaminated site consultant to assess a site for contamination and, where required, to develop a remediation plan, implement the plan and validate the remediation.

The contaminated site consultant designs and undertakes the site assessment and, where required, all remediation and validation activities to achieve the objectives specified by the owner or developer.

#### Site Auditor Reviews the Consultant's Work

The site owner or developer commissions the site auditor to review the consultant's work. The auditor prepares a site audit report and a site audit statement at the conclusion of the review, which are given to the owner or developer.

Where the local planning authority or DEC uses its legal powers to require the carrying out of a site audit, the site owner or developer must commission a site auditor accredited under the CLM Act to perform this task. This is known as a 'statutory' audit. The CLM Act requires that an auditor must notify DEC when he or she has been commissioned by anyone other than DEC to perform a statutory site audit. The auditor is also required to furnish the local authority and DEC with a copy of the completed site audit statement.

In some cases, the site owner or developer may wish to have a site audit undertaken although it is not a legal requirement. The audit is termed 'nonstatutory'. If their intention is to obtain a site audit statement, they must commission a site auditor accredited under the CLM Act to perform this task. This is because only a site auditor so accredited can issue a site audit statement and they are obliged to issue one at the end of any site audit. For nonstatutory audits, the site auditor must give a copy of the site audit report to the local authority or DEC, or both, on request.

As required by the CLM Act, DEC maintains a record of all statutory site audit statements issued in relation to land that is the subject of a regulatory instrument under the CLM Act. Copies are available for public inspection through DEC's website at www.environment.nsw.gov.au. If the local council

receives a copy of a site audit statement, it must list the statement on any certificate it issues under section 149 of the *Environmental Planning and Assessment Act 1979* in relation to the land concerned.

#### Limitations of Your Site Audit Report

The following notes have been added by the Auditor who prepared this report, to highlight some important limitations on the use of this report.

This report has been prepared by C. M. Jewell & Associates Pty Ltd for the use of the client who commissioned it, and relevant government agencies, for the specific purpose described in the report.

Consistently with the objectives of the NSW Site Auditor Scheme, it may be appropriate for others to rely upon this report in some circumstances.

However, the original purpose of this report and the site conditions prevailing at the time the report was prepared – as described in the report – should be considered first.

If you are not the person for whom the report was prepared, or you wish to use it for a different purpose to that for which it was prepared, or site conditions appear to differ from those described in this report, or a significant period of time has elapsed since the report was prepared, then PLEASE CONSULT THE SITE AUDITOR BEFORE RELYING UPON THE REPORT.

It is also important to recognise that a site audit is primarily a review of work carried out by other companies and individuals.

The site auditor has checked data and interpretations, ascertained whether or not appropriate guidelines have been followed, and satisfied himself that the available data are adequate to support the conclusions he has reached.

However, all environmental sampling programs have an inherent degree of uncertainty. Even when sampling fully complies with guidelines, it is possible for areas of contamination to remain undetected, but be revealed by more extensive excavations during site redevelopment. This risk is usually quantified using statistical confidence limits.

The site audit report identifies data limitations and uncertainties where these are recognised, but users must accept the finite and unavoidable risk that some contamination may remain undetected during even a diligent site assessment and audit process.

If there is a need to copy this report, it must be reproduced in full. No reliance whatsoever should be placed upon partial copies of a site audit report.




Report Ref: J0807.15R Rev: 0 Rev Date: 10/02/2006 Author: CS

# Figure 1

Site Location and Setting

#### Site Audit – Part 2, 100-120 King Street, Randwick



Figure 2



C.M. Jewell & Associates Pty Ltd

. Rev: 0 Rev Date: 27/06/2006 Author: CS

Site Landuse Design and Containment Cell Locations





Report Ref: J0807.15R Rev: 0 Rev Date: 27/06/2006 Author: CS



Lead Delineation Sample Locations and Concentrations



Source: URS, 2006



Report Ref: J0807.15R Rev: 0 Rev Date: 5/07/2006 Author: CS

## Figure 5

Sample Locations Asbestos Validation

C.M. Jewell & Associates Pty Ltd



Source: URS, 2006



Report Ref: J0807.15R Rev: 0 Rev Date: 5/07/2006 Author: CS

## Figure 6

Sample Locations Lead Excavation Validation

C.M. Jewell & Associates Pty Ltd



APPENDIX A Contaminant Groups

#### POLYCYCLIC AROMATIC HYDROCARBONS

Naphthalene Acenaphthylene Acenaphthene Fluorene Phenanthrene Anthracene Fluoranthene Pyrene Benz(a)anthracene Chrysene Benzo(b) & (k) fluoranthene Benzo(a)pyrene Indeno(1,2,3-cd)pyrene Dibenz(a,h)anthracene Benzo(g,h,i)perylene

#### TOTAL PETROLEUM HYDROCARBONS

 $C_6$  -  $C_9$  fraction  $C_{10}$  -  $C_{14}$  fraction  $C_{15}$  -  $C_{28}$  fraction  $C_{29}$  -  $C_{36}$  fraction

#### MONOCYCLIC AROMATIC HYDROCARBONS

Benzene Toluene Ethylbenzene meta- & para-xylene ortho-xylene Styrene

#### **HEAVY METALS**

Arsenic (As) Cadmium (Cd) Chromium (Cr) Copper (Cu) Lead (Pb) Mercury (Hg) Nickel (Ni) Zinc (Zn)



# APPENDIX B Communications with the Auditor



## C. M. Jewell & Associates Pty Ltd

Water and Environmental Management

, ,	02 4759 3251 02 4759 3257	International International	61 2 4759 3251 61 2 4759 3257		FAX
Ref:	J0807.18F			Date:	18 September 2006
To:	URS			C.C.	McLachlan Lister Pty Ltd
Attention	: Lachlan Wo	ood		Attention:	Tim Greenaway
Fax no:	8925 5555			Fax no:	9310 0952
From:	Chris Jewel			Total pages	including this page: 2

Site Audit - 100-120 King Street, Randwick

) Lachlan

I refer to your fax of 5 September 2006 and attached changes to the EMP.

I have no objection to any of the changes suggested by Tim.

However, I do suggest that the plan incorporate a requirement for provision of an annual compliance statement to Council. All that is required is a one-sentence letter stating that during the previous 12 months, the site has been managed in accordance with the EMP.

I think that it is important to have some requirement to do something positive in relation to the EMP each year, to ensure that, over time, it is not forgotten by Council or Montefiore.

Please note that this communication has been provided as interim advice only. Where applicable, the information provided is consistent with NSW DEC guidelines and policies. The advice does not constitute a site audit report or site audit statement and does not pre-empt the conclusions, which will be drawn at the end of the audit process. A site audit report and site audit statement will be issued when the audit process has been completed.

For and on behalf of C. M. JEWELL & ASSOCIATES PTY LTD

CHRIS JEWELL

This facsimile is confidential and may be legally privileged. If you are not the intended recipient, please notify us by return fax and destroy this message. You must not copy, disclose or use the content in any way. Thank you. P.O. Box 10, Wentworth Falls, NSW 2782, Australia 1/13 Kalinda Road, Bullaburra, NSW 2784, Australia Email postie@cm-jewell.com.au ABN 54 056 283 295

160

Alahan da kata watazi k

ł.

## and the second of the second o ManagementLimitations

pathways and methods for minimising occupants/worker's exposure to the contaminated fill and/or

. . . .

#### EMP Implementation 1.4

It is the intention that the implementation of this EMP will be the responsibility of Montaflore, as the site owner, and should be enforced by Randwick City Council (Council). It is recommended that a Public Positive Covenant under Section 88E of the Conveyencing Act 1919 is made to the Property Title, nominating Randwick Connoil as the Prescribed Authority. In addition, it is also considered appropriate that Council place a notation on the Planning Certificate for the site under Section 149 (2) of the Environmental Planning and Assessment Act 1979 (EP&A Act). This notation would address the restrictions for intrusive works on the site. Where a development application (DA) is required, Council will also be able to enforce the implementation of the EMP, through the conditions of consent for the DA. Intrusive works conducted on the site shall only be conducted by contractors/individuals who have read

The following table provides a summary of the responsibilities for the implementation and management. of the BMP. These responsibilities do not replace any other regulatory responsibilities of the parties in undertaking works at the site.

Organisation	Responsibilities
Randwick City Council	Note EMP on appropriate planning instrument
Montellore	
	<ul> <li>Organise revision of Property Title to note that the site is subject to implementation of this EMP.</li> </ul>
	<ul> <li>Nominate a representative of Montefiore to undertake annual Inspections of the capping area. Document and report the inspections to the Montefiore Board on an annual basis.</li> </ul>
	Inplant of the state
	moondance with provide are being undertakenter I - from de Council
	Notify Council of any non-conformance with the EMP and Statement of Compliance with the EMP and Statement of Compliance with and as built drawings.

..... Table 1 EMP Implementation Responsibilities

1-2

 $\nu \nu \sigma \nu$ 13120

## Re: FW: Montefiore Home Randwick Chris Jewell

From: To:

Date sent: Subject: Send reply to: Priority: "Chris Jewell" <chris@cm-jewell.com.au> "Tim Greenaway" <TimG@McLachlanlister.com.au>, <Francene\_Mitchell@URSCorp.com> Tue, 22 Aug 2006 23:32:40 +1000 Re: FW: Montefiore Home Randwick chris@cm-jewell.com.au normal

Tim,

Sorry, this one fell through the cracks.

l suggest a Public Positive Covenant under Section 88E of the Conveyancing Act 1919, nominating Randwick Council as the Prescribed Authority.

The Covenant can state:

"Due to the presence of contaminated soil beneath some parts of the land, the land shall at all times be managed in accordance with the site environmental management plan No. XXXXXXXX prepared by URS Australia and dated XXXXXXXXX, as is required by site audit statement SA XXXXXX issued by Christopher Jewell and dated XXXXXXXXXXXXXXXXX.

The covenant will need to be imposed in the form required by Section 88E (3) or 88E (4) as appropriate.

Chris.

On 22 Aug 2006 at 15:00, Tim Greenaway wrote:

> Chris,

> Could you respond to my email below. Thanks,

>

>

> >

> Chris,

> Further to your fax dated 31 July 2006 regarding the Environmental

Management Plan, I have checked with Council and they require we place

a

> positive covenant on title, refer email below. Do you have some words

> that may be suitable or I will have the Montefiore Home's lawyers

> prepare something for yourself and Council to review?

>

> Regards,

>

>

> Tim Greenaway.
 > Executive Manager - Project D

> Executive Manager - Project Delivery

> McLachlan Lister Pty Limited

> ASN Building

> Level 1

> 1-5 Hickson Road

> THE ROCKS NSW 2000

Printed for Natalie Addison, 23 Aug 2006, 10:18

Page 1 of 5

## C. M. Jewell & Associates Pty Ltd

Water and Environmental Management



Phone Fax	02 4759 3251 02 4759 3257	International International	61 2 4759 3251 61 2 4759 3257	·	<b>F</b>
Ref:	J0807.17F			Date:	31 July 2006
To:	URS			C.C.	McLachlan Lister Pty Ltd
Attentic	n: Fran Mitchel	l		Attention:	Tim Greenaway
Fax no:	8925 5555			Fax no:	9310 0952
From:	Chris Jewell			Total pages	including this page: 1

Site Audit – 100-120 King Street, Randwick

#### Fran

From:

I have reviewed your Draft Environmental Management Plan (EMP) and outline below a number of issues to be addressed. The new draft of the EMP should then be submitted to me and to Randwick City Council for comment.

- As outlined in the DEC's Guidelines for the NSW Site Auditor Scheme, 2nd edition (2006) specifically in 1. Section 3.4.6, all EMPs must include an enforcement mechanism. The options for this site, as I see it, are
  - a) the placement of a positive covenant on the title, or
  - b) a Section 96 application in order to make compliance with the EMP a condition of the existing DA.

Council must agree to the enforcement mechanism chosen and must also agree to notification of the EMP on any planning certificate issued under Section 149 (2) of the EP&A Act for the site.

- 2. In Section 4.1 of the draft EMP, more detail is required concerning the maintenance of the concrete capping. Please specify how frequently inspections should be conducted and by whom. Details of the inspections and their findings must be documented; please explain how this should be done.
- 3. In Section 4.6, please replace the word 'investigation' in the first sentence with 'excavation'.
- 4. In the EMP's list of references, please note that the correct reference for the Guidelines for the NSW Site Auditor Scheme should read: NSW DEC 2006, Guidelines for the NSW Site Auditor Scheme, 2nd edition, NSW Department of Environment and Conservation, Sydney NSW.

I have recently viewed the site but will need to conduct a more detailed inspection of the containment areas during a final site inspection, at a date to be arranged.

Please note that this communication has been provided as interim advice only. Where applicable, the information provided is consistent with NSW DEC guidelines and policies. The advice does not constitute a site audit report or site audit statement and does not pre-empt the conclusions, which will be drawn at the end of the audit process. A site audit report and site audit statement will be issued when the audit process has been completed.

#### For and on behalf of C. M. JEWELL & ASSOCIATES PTY LTD

CHRIS JEWELL

This facsimile is confidential and may be legally privileged. If you are not the intended recipient, please notify us by return fax and destroy this message. You must not copy, disclose or use the content in any way. Thank you. P.O. Box 10, Wentworth Falls, NSW 2782, Australia 1/13 Kalinda Road, Bullaburra, NSW 2784, Australia Email postie@cm-jewell.com.au ABN 54 056 283 295

## C. M. Jewell & Associates Pty Ltd

Water and Environmental Management

Phone Fax		759 3251 759 3257	International International	61 2 4759 3251 61 2 4759 3257		751 <b>FAX</b>
Ref:		J0807.16F			Date:	26 June 2006
To:		URS			C.C.	McLachlan Lister Pty Ltd
Attentio	on	Fran Mitche	əli		Attention:	Tim Greenaway
Fax no		8925 5555			Fax no:	9241- <del>1898</del> <del>ጉ</del> ንጉ እ
From:		Chris Jewe			Total pages	including this page: 2 NA
Site Au	ıdit –	King Street	Randwick			

CMJA

I have reviewed your report Remediation Validation, Part 2 Area, Lot 202 King Street Randwick against the mandatory requirements of the *Guidelines for the NSW Site Auditor Scheme* 2nd edition (2006).

Attached is the relevant checklist section that assesses compliance with the guidelines. The field and laboratory QA/QC section of the checklist is not able to be completed due to the required information outlined below.

The report indicates that work was substantially conducted as outlined in the RAP and as previously agreed. While you may comment on any identified non-compliance and partial compliance items listed, the issues that require clarification areas follows.

- It appears that duplicate identifications have been repeated over time, which makes checking difficult eg. QC 05 is tabulated as a duplicate for VAL214 (2/10/03) however it is not in the laboratory report of that date and a laboratory report has a QC05 dated 23/7/03. In addition you have not tabulated the primary and duplicate samples concentrations or RPD's. Could you please do this for the delineation data set (Table 1) and the validation data set (Tables 2a and 2b)?
- 2. Table 3 lists two TP02-1.0-1.1 samples (with different dates) both with the same two duplicates QC10 and QC11, neither of which are listed as a inter laboratory duplicate. In addition it appears somewhat coincidental that these samples also have exactly the same concentration.
- 3. You have stated that a rinsate blank was identified 'QC03' and a field blank as'QC04', yet you have also listed these as duplicates. Could you please supply information on blanks used i.e. tabulate the analytical results?
- 4. According to your validation data (Table 2b) TP206 samples and two out of three samples from TP202 (one isn't?) and are boundary samples, yet TP01 and TP307 also appear to be located on the boundary.
- 6. It isn't clear as to why samples VAL 200, 203, 209, 210 and 220 are tabulated in Table 3 rather than the excavation validation data. Neither is there reference to these samples within the report nor on a figure.
- 7. There is no mention of the current status of the excavation, i.e was it reinstated?

This facsimile is confidential and may be legally privileged. If you are not the intended recipient, please notify us by return fax and destroy this message. You must not copy, disclose or use the content in any way. Thank you.

P.O. Box 10, Wentworth Falls, NSW 2782, Australia 1/13 Kalinda Road, Bullaburra, NSW 2784, Australia Email postie@cm-jewell.com.au ABN 54 056 283 295 8. Could you please clarify whether the day care centre is still present within the southern western portion of the property?

Please respond to Caitlin Spiller in the first instance.

Please note that this communication has been provided as interim advice only. Where applicable, the information provided is consistent with NSW DEC guidelines and policies. The advice does not constitute a site audit report or site audit statement and does not pre-empt the conclusions which will be drawn at the end of the audit process. A site audit report and site audit statement will be issued when the audit process has been completed.

For and on behalf of C. M. JEWELL & ASSOCIATES PTY LTD

CHRIS JEWELL

ى مياويون <u>سيونيا بروكي محيد توهيجنا أ</u>لاك تخطيف المحد الح

静和风气。	<u>li COM</u>	nkole.	
Control No:	012	-7	
Date received:	10.0	1.03	
Job Ne:	0807		
Routing:	CMJ	15/	
		l .	
,	675 49	16/	

9 September 2003 Project No. 51072-001

CM Jewell & Associates PO Box 10 Wentworth Falls NSW 2782

Attention: Chris Jewell NSW EPA accredited Site Auditor

Dear Chris,

#### Subject: Proposed RAP Modification – Lot 202 King St - RCC Conditional Approval

Please find enclosed Randwick City Council's conditional approval of the RAP Modification (URS, 26 August 2003).

In relation to the conditions listed by Randwick City Council (RCC), the following is noted for your information:

- The Site Management Plan (SMP) will be prepared as part of the documentation presented to yourself for review and approval.
- An appropriate Covenant will be placed on the title to the satisfaction of RCC.
- The SAS is to include appropriate conditions regarding the SMP and the Covenant, with the concurrence from RCC.

In addition and in response to the 3<sup>rd</sup> condition noted by RCC, URS considers that the validation of the remediation methodology (i.e. cap and contain) will be documented in the Part 2 Validation Report, rather than in the SMP. The SMP will include details on the construction of the containment area for the purpose of recording these details should disturbance of the area be necessary in the future.

Details of the remediation planning for the selected Option C, have been previously summarised in the RAP Modification (URS, 26 August 2003) and include:

• Excavation of material which exceeds the NEPM 'E' Guideline Level from the open space areas of the Part 2 area and replace this material beneath the building or paving footprints.

URS Australia Pty Ltd (ABN 46 000 691 690) Level 3, 116 Miller Street North Sydney, NSW 2060 Australia Tel: 61 2 8925 5500 Fax: 61 2 8925 5555





Chris Jewell CM Jewell & Associates 9 September 2003 Page 2

- The building or paving would provide a physical barrier to prevent access to the impacted material. No material would be disposed off-site under this option.
- This approach would most likely result in the loss of the trees within the remediation area, and would require prior Council approval to allow for the trees to be removed.
- The containment area for the excavated material beneath the building or paving footprint will not meet the NEPM 'D' Guideline Level and a conditional SAS would be anticipated for this area. A SMP would be required to appropriately manage the contaminated material beneath the buildings or paving and a notation on the property title would be enacted.
- The material to be replaced under either paving or building foundations has been assessed to be suitable in that the material will be replaced above the water table. Groundwater level monitoring has been recently undertaken by Jeffrey & Katauskas (J&K). The J&K reports (4 August 2003 and 18 July 2003) indicate that the standing groundwater level in the north east of the site is approximately 38m AHD, although it is noted that a fluctuation of 1 m-2 m in level is possible in the Botany Basin. Assuming a 2m rise, it is proposed that materials will be placed above 40m AHD.
- The impacted soil materials will be underlain and overlain by a clearly visible geofabric marker layer. In the areas where paving is proposed, a 100mm sand cover will be placed to overly the geofabric and subsequently a blinding layer of concrete prior to the interlocking paving bricks. It is anticipated that the paving will be of low permeability, given the placement of the blinding concrete layer. In the areas where material is retained under a building slab, a similar placement methodology will be implemented including marking of the impacted materials with a high visibility geofabric marker layer, placement of a 100 mm sand cover layer, followed by a blinding concrete layer and then concrete slab construction.
- A SMP will be prepared to detail the protocols which will be implemented to restrict access to the impacted material. The SMP will include procedures for the maintenance and inspection of the containment.

In summary, URS consider that the "cap & contain" method of managing the contamination on-site will:

- Adequately contain the contaminated material by the construction of the containment area as described above;
- Restrict unreasonable interference with the contaminated material as the material will be generally inaccessible beneath the paving slab. In addition, the SMP will provide protocols for the restriction of excavations and the maintenance and inspection of the capping;





Chris Jewell CM Jewell & Associates 9 September 2003 Page 3

• Prevent potential leaching of the contaminants to the underlying groundwater table as the contaminated material will be placed above the estimated groundwater level and beneath low permeability paving

Further to your e-mail dated 28 August 2003, could you please indicate if the proposed "cap & contain" remediation methodology and associated SMP is appropriate to ensure that the contaminated material will not be unreasonably interfered with or migrate with respect to groundwater.

McLachlan Lister/Montefiore would like to proceed with the works as a matter of priority such that the remediation works are completed prior to intensive construction activities being undertaken on the Part 1 land parcel of Lot 202.

Yours sincerely, URS AUSTRALIA PTY LTD

Fran Mitchell Associate Engineer

Martin Howell Senior Principal

Enclosures:

)

RCC, 5 September 2003, Amendment of Remediation Action Plan

cc: Tim Greenaway McLachlan Lister



.d03	MON	18:	35	1.4

002

#### -5 September 2003

Dur Ref:

McLachlan Lister Pty Ltd

68 Cloucester Streat

The Rocks NSW 2000

Attention: Tim Greenaway

Dear Tim

Re: Sir Moses Monteflore Jewish Home 100 - 120 King Street, Randwick Amendment Of Remediation Action Plan

C/Q563/2003, P/000869---

Thank you for your recent correspondence in relation to the amendment of the Remediation Action Plan (RAP) for the Sir Moses Montefiore Jewish Home located at King Street, Randwick

The information has been reviewed and no objection is raised to accepting the amended RAP, subject to the following conditions:

Conditions of Council consent No. 02/00551/G1 approved on 22 October 2002 being adhered to.

A Site Management Plan (SMP) being prepared in accordance with the site auditors specifications, obtain the approval of the site auditor and the Council prior to the implementation of the site management plan.

It is to be clearly demonstrated in the SMP that the "cap and contain" method of managing the contamination onsite will contain the contaminated material and ensure it will-not be unreasonably interfered with or migrated (in particular by groundwater), to the satisfaction of the auditor.

It is to be clearly identified in the SMP what and where the contaminated material is located.

The approved site management plan be implemented prior to the occupation of the site.

An appropriate covenant be place on the title to the satisfaction of Council. Appropriate conditions regarding the SMP and the Covenant on the land title are to be

Appropriate conditions regarding the SMP and the Opverland on the land the are to be included on the Site Audit Statement accordingly, with the concurrence from Council.

Should you require any further details, please do not hesitate to contact Ms Joanne Brown, Senior Environmental Health Officer, on 9399 during business hours Monday to Friday.

Yours faithfully

Roman Wereszczynski Manager Building Certification Services

Per. !

Joanne Brown Senior Environmental Health Officer G:\TOWN\WP\TLET\Tle:2003\100KingStJB9-5-ss.doo

#### Jennifer Tully

3° a

Date sent: From: Subject: To: Copies to: Send reply to: Organization: Thu, 11 Sep 2003 00:18:07 +1000 Chris Jewell <cm\_jewell@hotmail.com> Re: Montefiore, King Street, Randwick RAP Modification Francene\_Mitchell@URSCorp.com Jenny@cm-jewell.com.au Chris Jewell <chris@cm-jewell.com.au> C.M.Jewell & Associates Pty Ltd 8765

#### Dear Fran,

Further to your letter of 9th September, our meeting last week and my previous email (28th August), I confirm that the proposed approach to containment of lead contaminated soils and associated SMP will be satisfactory, and that groundwater-related issues have been adequately considered.

Chris.

----- Original Message -----From: <<u>Francene\_Mitchell@URSCorp.com</u>> To: <<u>Chris@cm-jewell.com.au</u>> Sent: Tuesday, September 09, 2003 4:59 PM Subject: RAP Modification - RCC response

>			
> Francene Mitchell		A local statement of the second statement of the second statement of the	····
> Associate Engineer			9
> URS Australia Pty Ltd		Cant of Ma-	10142
> Phone: +61 (02) 8925 5500	Fax: +61 (02) 8925 5555	Direct merereived:	9.9.03
> Dial: +61 (02) 8925 5712		Job No:	0807
> 116 Miller St, North Sydney 206	0	Mouting:	0m7
>			
> This e-mail together with any att	achments is confidential and m	hay be the	Production from the second
> subject of legal privilege. If you	are not the intended recipient (	please	
> contact us by return email and d	estroy this message. You are	not have been and the second second	
permitted	• –		and the second
> to copy, disclose or use the cont	tent in any way. Thank you.		

Printed for Jennifer Tully, 11 Sep 2003, 9:18 Page 1 of 2

	>
	>
	>
	> Forwarded by Francene Mitchell/Sydney/URSCorp on 09-09-03 05:01 PM
	>
	> Francene Mitchell
	<u>Chris@cm-jewell.com.au</u>
	> 09-09-03 03:43 PM cc:
	tim@mclachlanlister.com, Martin Howell/Sydney/URSCorp@URSCorp
	> Subject RAP
	Modification - RCC response
	> second seco
	>
	> Chris,
	> Randwick City Council (RCC) have responded to URS' RAP Modification
)	> regarding the proposed lead remediation works in Part 2 of the Lot 202
	King
	> St site.
	>
	> Please find attached the RCC conditional approval (5 September 2003).
	>
	> In response to the RCC conditional approval, URS has drafted the attached
	> letter to yourself to clarify the proposed remediation methodology of the
	> containment area and its associated integrity.
	> Can you please review the URS letter and provide comment in relation to
·	
<u></u>	> RCC condition.
	> Regards,
	> Fran
	> (See attached file: URS 9 September 2003 re conditional approval.pdf)(See
	> attached file: RCC response 5 September 2003.pdf)
1	>
1	

.

÷.

. •



C. M. Jewell & Associates Pty Ltd Water and Environmental Management A.B.N. 54 056 283 295

# fax transmission

### P.O. Box 10, Wentworth Falls, NSW 2782, Australia 1/13 Kalinda Road, Bullaburra, NSW 2784, Australia



 Phone (02) 4759 3251
 Email postie@cm-jewell.com.au
 Fax (02) 4759 3257

 (International +61 247 59 3251)
 (International +61 247 59 3257)

Ref: J0807.7	Date: 20 December 2002	Time: 12:50
To: McLachlan Lister Pty Ltd		c.c. URS Australia Pty Limited
Attention: Tim Ereenaway		Aftention Frank Mindlell)
Fax No 3241 1898	From: Chris Jewell	Fax No. 8925-5555
Original to follow: No		Total pages including cover: 1

#### Subject:

Dear Tim and Fran,

Further to our site meeting and inspection yesterday, I am satisfied that URS and Hibbs Associates have effectively managed the asbestos issues on the site, and subject to Hibbs' clearance certificate for asbestos and the results of URS's validation sampling, I would consider that the asbestos has effectively been remediated.

With regard to future management during my absence over the following few months, we agreed the following:

- 1. McLachlan Lister/URS will keep me informed of progress on the site by email. Emails will be copied to Fiona Keserue-Ponte at this office.
  - 2. Fiona will visit the site on 1 or 2 occasions during remedial works, the preferable times for her to visit would be when there is maximum exposure of the site following completion of demolition and pavement removal and, secondly, when remediation is nearly complete, the later may coincide with my return during the last 2 weeks of February and if that is the case I will carry out a further site visit myself.

I wish you and your families and all at URS a happy Christmas and a peaceful new year.

For and on behalf of <u>C. M. JEWELL & ASSOCIATES PTY LTD</u>

CHRIS JEWELL

Administrative Centre 30 Frances St Randwick 2031 Tel: 02 9399 0999 Fax: 02 9319 1510 general.manager@randwick.nsw.gov.au



INCORPORATED AS A MUNICIPALITY 22 FEBRUARY 1859 PROCLAIMED AS A CITY 1 JULY 1990

ABN: 77 362 848/2/G P/000869 (Contact Officer: Joanne Brown – 9399 0879)

20 December 2002

McLachlan Lister Pty Ltd 68 Gloucester Street THE ROCKS NSW 2000

Attention: Tim Greenaway

DOCUM	ENT CONTE	IOL
Control No:	9245	
Date received:	24-12	2-02
Job No:	0807	
Routing:	CMT	61
	Fi F	6/

Dear Mr Greenaway

## Re: 100 – 120 King Street, Randwick

Reference is made to the Remediation Action Plan submitted to Council on 2 December 2002, together with a letter by C M Jewell & Associates Pty Ltd dated 27 November 2002.

Following discussions with the site auditor you are advised of the following:

- The land where buildings will be located (minimal direct access to soil) are to be
  remediated to NEHF D Criteria
- The recreational gardens, and unpaved open space areas are to be remediated to NEHF
   E Criteria
- The Statutory Site Audit Statement (SAS) and Summary Site issued Audit Report (SSAR) issued to Council shall be unconditional, in that it requires no further monitoring, ongoing review or remedial actions, beyond the stage of issuing the Construction Certificate, and shall cover both land and groundwater. The SAS should clearly state, the remediated land is at an asbestos free level or to a level where no unacceptable health risk remains.

Should you require further details or information please contact our Senior Environmental Health Officer, Joanne Brown, on 9399 0879 during business hours.

Yours faithfully,

Roman Wereszczynski Manager, Environmental Health and Building Services

Per: ..... Joanne Brown Senior Environmental Health Officer

Cc C M Jewell & Associates Pty Ltd PO Box 10 WENTWORTH FALLS NSW 2782

> Fran Mitchell URS Australia Pty Ltd Level 3, 116 Miller Street NORTH SYDNEY NSW 2060

G:\TOWN\WP\TLET\Tlet2002\JB12-20.1V.doc



## C.M. Jewell & Associates Pty Ltd A.C.N. 056 283 295 A.B.N. 54 056 283 295

Water and Environmental Management

My Ref: J0807.6 27 November 2002 CMJ:na

1/13 Kalinda Road, Bullaburra, NSW 2784, Australia P.O. Box 10, Wentworth Falls, NSW 2782 Phone: (02) 4759 3251 Fax: (02) 4759 3257 Email: postie@cm-jewell.com.au

McLachlan Lister Pty Limited 68 Gloucester Street THE ROCKS NSW 2000



Attention: Mr Tim Greenaway

Dear Tim,

<u>ن</u> ک

## RE: Remedial Action Plan - Sir Moses Montefiore Jewish Home

As the NSW EPA accredited site auditor appointed for the site, I have reviewed the final remedial action plan (RAP) (26 November 2002) prepared by URS Australia in respect of the Sir Moses Montefiore Jewish Home development in Randwick.

I have previously reviewed a draft of the RAP; my comments on the draft have been incorporated into the final document.

The RAP generally complies with the recommendations of the EPA *Guidelines for Consultants Reporting on Contaminated Sites* and with Australian Industry practice. I consider that it also meets the site-specific requirements of this project and forms an appropriate basis for site remediation.

I consider that if the site is remediated and validated in accordance with the RAP, it should then be suitable for the proposed use (residential with minimal opportunity for soil access) and I should be able to issue a site audit statement to that effect.

For and on behalf of C. M. JEWELL & ASSOCIATES PTY LTD

CHRIS JEWELL

cc: Fran Mitchell, URS Australia Pty Ltd



## C.M. Jewell & Associates Pty Ltd A.C.N. 056 283 295 A.B.N. 54 056 283 295

Water and Environmental Management

1/13 Kalinda Road, Buliaburra, NSW 2784, Australia P.O. Box 10, Wentworth Falls, NSW 2782 Phone: (02) 4759 3251 Fax: (02) 4759 3257

My Ref: J0807.5 21 November 2002 CMJ:mp

McLachlan Lister Pty Limited 68 Gloucester Street THE ROCKS NSW 2000



Email: postie@cm-jewell.com.au

Attention: Mr Tim Greenaway

Dear Tim,

## RE: Remedial Action Plan - Sir Moses Montefiore Jewish Home

As the NSW EPA accredited site auditor appointed for the site, I have reviewed the draft remedial action plan (RAP) (6 November 2002) prepared by URS Australia in respect of the Sir Moses Montefiore Jewish Home development in Randwick.

The RAP generally complies with the recommendations of the EPA *Guidelines for Consultants Reporting on Contaminated Sites* and with Australian Industry practice. I consider that it also meets the site-specific requirements of this project and forms an appropriate basis for site remediation.

I have the following comments on three specific aspects of the RAP.

## Section 3.4.2 Page 3-12 and corresponding part of Section 3.5, Page 3-14

- 1. Excavation below the water table in the sands present below this part of the site is likely to encounter stability problems unless some dewatering is carried out. Measures for dealing with this issue should be detailed in the RAP. In particular, it may prove difficult to collect the validation samples specified in Section 3-5.
- 2.
- 3. I think the word 'months' has been omitted from line 3 of the last paragraph of Section 3.4.2.

Section 3.4.4 Page 3-13

Paragraph numbered 3.

A visual inspection of the entire surface should also be included.

#### Section 3.6 Page 3-14

3. The reference to 'No more than 20,000 T of soils classified as Inert Waste ...' relates to the "Waste Facilities" heading of Schedule 1 of the Protection of the Environment (Operations) Act 1997. The exemption given under part (f) (iii) of this section does not apply to this site because it is caught by the following part (g) as the development is classed as residential.

Thus only VENM can be used.

-

Page 3-15

4

Sampling at a rate of 1 sample per 1000 cubic metres is inadequate - the rate should be 1 per 100 to 1 per 250 cubic metres, depending upon the volume imported from one source.

For and on behalf of <u>C. M. JEWELL & ASSOCIATES PTY LTD</u>

CHRIS JEWELL

a de la companya da la companya da comp

c.c. Fran Mitchell, URS Australia Pty Ltd



C. M. Jewell & Associates Pty Ltd Water and Environmental Management A.B.N. 54 056 283 295

P.O. Box 10, Wentworth Falls, NSW 2782, Australia 1/13 Kalinda Road, Bullaburra, NSW 2784, Australia fax transmission

FAXED

 Phone (02) 4759 3251
 Email postie@cm-jewell.com.au
 Fax (02) 4759 3257

 (International +61 247 59 3251)
 (International +61 247 59 3257)

Ref: J0807.2	Date: 26 June 2002	Time: 12:02	
To: URS		c.c. McLachlan Lister	
Attention: Fran Mitchell		Attention: Simon Magri	
Fax No. 8925 5555	From: Chris Jewell	Fax No. 9241 1898	
Original to follow: No		Total pages including cover: 1	

## Subject: Supplementary Sampling and Analytical Plan, Lot 202 King Street, Randwick

Fran,

**К**)

Further to our telephone conversation today, I confirm that I have reviewed your supplementary sampling and analytical plan and am generally happy with the approach that you have proposed.

There are two areas where some modification is appropriate. These are:

- 1) Comparability of data sets. Unless you are sure that you can demonstrate consistency of sampling and analytical methodology, allowing the original and supplementary data sets to be combined, then it would be worthwhile including some overlap sampling to allow a direct comparison.
- 2) Asbestos. I would like you to follow the approach to asbestos assessment outlined in the ACLCA Code of Practice (February 2002 draft), taking into account the comments in enHealth's review of that document. I note that you intend to sample from test pits, and concur with that approach. I would like to see your sampling program supplemented, and placed in context, by a walk-over inspection of the whole site by an asbestos specialist.

I don't think that either of these suggestions will result in significant extra costs or delays, and should help ensure that we end up with a validation data set with which we are all comfortable.

For and on behalf of C. M. JEWELL & ASSOCIATES PTY LTD

CHRIS JEWELL



# APPENDIX C Information Relied Upon by the Auditor



# APPENDIX C1 Summary of Laboratory Results

e suro report de la uny voy transmurant en anomena annome Soil Analytical Results - Lead Impact Delineation Results Lot 202 - Part 2 Validation

AND STREET

11.4

•. .

	T31         193         194 <th>Sample ID</th> <th>Sample Date</th> <th>Primary Laboratory</th> <th>Primary Batch Nu</th> <th>(84/8ill) pearl</th> <th>QC Sample II)</th> <th>QC Sample Type</th> <th>Secondary QC Batch No</th> <th>QC concentrațion Lead (mg/kg)</th> <th>(%)</th>	Sample ID	Sample Date	Primary Laboratory	Primary Batch Nu	(84/8ill) pearl	QC Sample II)	QC Sample Type	Secondary QC Batch No	QC concentrațion Lead (mg/kg)	(%)
12.33         1935         Emeriant         20013         6	213         1933         Entiment         20103         5         10000         100	Z7-2	1995	Enviromet	9505153	\$					
ZrA         1950         Berforment         9900013         6.4         9900013         6.4         9900013         6.4         9900013         7.4         9900013         9900013         9900013         9900013         9900013         9900013         9900013         9900013         9900013         9900013         9900013         9900013         9900013         9900013	TAI         1981         Entriment         2000.01         Control         Con	Z7-3	1995	Enviromet	9505153	Ş					
TY. 12.81         MURDE	TYL         MARN	Z7-4	1995	Enviromet	9505153	9					
Statut         Statut<	10.300         200,000         A.33         SERIED (ACC)         Control (ACC)         Contro (ACC)         Contro (ACC)         Co	Z7-8	1995 14 Internor	Enviromet	LIVEYEDE	C C	DI IBO2	field dunlicate	ALS - ES 34736	3	200
Nonline         Als         Stability	Nonl 10.51         Symon         All         Symon	11/02 0.2-0.4	24(U // 2002	ALS	ES39497	618					
Wares         0.000         A.3         ENSOF         Main         Description         A.3         ENSOF	Wares         0.33         BESNOF         M.34         ENSOF         M.35         <	HA201 0.4-0.5	30/04/2003	ALS	ES39497	484					
Mum         Ord         Mis         Entrol         Mis         Mis <td>Manne         Manne         <th< td=""><td>HA202 0.2-0.3</td><td>30/04/2003</td><td>ALS</td><td>ES39497</td><td>-47-749 - 0.5</td><td></td><td></td><td></td><td></td><td></td></th<></td>	Manne         Manne <th< td=""><td>HA202 0.2-0.3</td><td>30/04/2003</td><td>ALS</td><td>ES39497</td><td>-47-749 - 0.5</td><td></td><td></td><td></td><td></td><td></td></th<>	HA202 0.2-0.3	30/04/2003	ALS	ES39497	-47-749 - 0.5					
Image:	Handle (14)         Mass         Entroper (1	HA202 0.4-0.5	30/04/2003	ALS	ES39497	2940					
Rooted cost         11877001         ALS         Bennin	Index         Index <th< td=""><td>HA203_0.4-0.5</td><td>30/04/2003</td><td>ALS</td><td>ES39497</td><td>181</td><td></td><td></td><td></td><td></td><td></td></th<>	HA203_0.4-0.5	30/04/2003	ALS	ES39497	181					
Mont 0,4ct         Mont 0,	Mach (1-6,1)         Threfforthold         Als         Entrolic	HA204 0.2-0.3	11/07/2003	ALS	ES41010	1430					
Model (je,1)         Differention         All         Bistion         State         Constant	Mask (jc)1         Different         Als         Bittion         State         Constrained         Als         Bittion         Total         Constrained         Als         State         Constrained         Constrained         Constrained         Constrained         Constrained         Constrained         Constrained         Constrained         Constrained         Constrained <th< td=""><td>HA204_0.6-0.7</td><td>11/07/2003</td><td>ALS</td><td>ES41010</td><td>2460</td><td></td><td></td><td></td><td></td><td></td></th<>	HA204_0.6-0.7	11/07/2003	ALS	ES41010	2460					
Fixed 0.2.13         Fixed 0.2.13<	Monto 0,453         Minition         All         Bennin         263           Monto 0,451         11/07/060         All         Bennin         263           Monto 1,411         11/07/060         All         Bennin         263           Monto 1,611         11/07/060         All         Bennin         263           Monto 1,611         11/07/060         All         Bennin         264           Monto 1,611         11/07/060         All         Bennin         266           Monto 1,611         11/07/060	HA204 1.0-1.1	11/07/2003	ALS	ES41010	2101				-	
Motor         Motor <th< td=""><td>Holdson         Holdson         Add         Estimation         Add         Add         Add         <th< td=""><td>HA205 0.2-0.3</td><td>11/07/2003</td><td>ALS</td><td>ES41010</td><td>263</td><td></td><td></td><td></td><td></td><td></td></th<></td></th<>	Holdson         Holdson         Add         Estimation         Add         Add         Add <th< td=""><td>HA205 0.2-0.3</td><td>11/07/2003</td><td>ALS</td><td>ES41010</td><td>263</td><td></td><td></td><td></td><td></td><td></td></th<>	HA205 0.2-0.3	11/07/2003	ALS	ES41010	263					
MASS 15-15         INTOR06         ALS         FS4100         FS600	Musice (3-1)         Imprise         Als         Estinol         Procession         Procession         Als         Estinol         Procession         Als         Estinol         Procession         Als         Estinol         Procession         Als         Estinol         Procession	HA205 0.6-0.7	11/07/2003	ALS	ES41010	344					
HURDON         ALS         ESATION         ALS	MACRO         Col         MAC         Station         400         Col         Tool         Tool <th< td=""><td>HA205 1.0-1.1</td><td>11/07/2003</td><td>ALS</td><td>ES41010</td><td>666 Tak</td><td></td><td></td><td></td><td></td><td></td></th<>	HA205 1.0-1.1	11/07/2003	ALS	ES41010	666 Tak					
Hoade Get-1         Internation         Alia         Encino         Encino         Encino         Encino         Encino<	H.M.G. 66-7	HA206_0.2-0.3	11/07/2003	ALS	ES41010	430					0.000
HAOR         Link         BSA100         160         10	HAOO         Link         Estimo         160         16	HA206_0.6-0.7	11/07/2003	VIS	ES41010	764	QC12	field duplicate	ALS-ES41010		Ye count
HART 0.54.3         1107203         ALS         ESH010         FG3         FS4010	HAOT         CAL         HINOR         ALS         ESA1010         FAO         ALS	HA206_1.0-1.1	11/07/2003	ALS	ES41010	169					
HAND         Link         Estinition         HAND         Link         Link         Link         Link         Link         Link         Link         Link         Link         Link <thlink< th=""> <thlink< th=""> <thlink< th=""></thlink<></thlink<></thlink<>	HAMP 0.64/1         11007000         ALS         Estinio         1430         HAMP 0.64/1         11007000         ALS         Estinio         5400         6401         HAMP 0.64/1         HU072005         ALS         Estinio         5400         FAU         HAMP 0.64/1         HU072005         ALS         Estinio         5400         FAU         HAMP 0.64/1         HU072005         ALS         Estinio         5400         FAU         HAMP 0.64/1         HU072005         ALS         Estinio         250         Hu010         FAU         Hu100         Hu100         Hu100	HA207_0.2-0.3	11/07/2003	ALS	ES41010	1 160					
HAGOD (16.1)         11070003         ALS         EN1000         610         600         610         610         610         610         610         610         610         610         610         610         610         610         610         610         610         610         610	Hadron Lot         Hadron		11/07/2003	ALS	ES41010	1340					
11072003         ALS         ESA1010         STAGE         ALS         SEA1010         STAGE         ALS         SEA1010         STAGE         ALS         SEA1010         STAGE         ALS         SEA1010         STAGE	1107/2003         ALS         ESA1010         STAGE         ALS         STAGE		11/07/2003	STY	ES41010	5130					
II         III         MIS         E40100 $3730$ MIS         E401010 $3560/7090$ $3300/7090$	1107/2003         ALS         ESA1010         S73         Contrinue         ALS         ESA103         ALS         ESA103 </td <td>HA208 0.6-0.7</td> <td>11/07/2003</td> <td>ALS</td> <td>ES41010</td> <td>. 3370</td> <td></td> <td>-</td> <td></td> <td></td> <td></td>	HA208 0.6-0.7	11/07/2003	ALS	ES41010	. 3370		-			
II/077003         AIS         E841010         230         Colory (Color)         230         Colory (Color)         230         Colory (Color)         230         Colory (Color)         2300/7000         573         E841010         230         Colory (Color)         2300/7000         533         Colory (Color)	II(07)2003         AIS         E841010         230         AIS         E841010         AIS         E841010         230         AIS         E841010         AIS         AIS         AIS         AIS         AIS         AIS         AIS         AIS         AIS <t< td=""><td>FIA208 1.0-1.1</td><td>11/07/2003</td><td>ALS</td><td>ES41010</td><td>716</td><td></td><td></td><td></td><td></td><td></td></t<>	FIA208 1.0-1.1	11/07/2003	ALS	ES41010	716					
Interfactors         ALS         ESA1010         573         QCI0/QCI1         Aniplicate         ALS-B541010         5960/7000           110072003         ALS         ES41190         2600         703         ALS-B541010         5960/7000         5960/7000           230072003         ALS         ES41189         41         1<	I107/2003         ALS         E84100         573         OCIO/CELL         Alded duplicate         ALS         E84100         593         OCIO/C000         593         OCIO/C001         ALS         E84100         5960/7000<	FIA209 0.6-0.7	11/07/2003	ALS	ES41010	230					
11/07/2003         ALS         Estility         Cd0         QCI0/QCI1         And enginering         ALS-ES41010         5960/7000           23/07/2003         ALS         ES41189         1         Acs         ES41189         1         Acs         ES41010         5060/7000         5050/7000	1107000         ALS         ES1101         AGD         QCI0/QCI1         Ando component         ALS-5541010         560/7000           230772003         ALS         ES1189         1         ALS         ES1189         1         960/7000         960/7000           230772003         ALS         ES1189         1         ALS         ES1189         1         960/7000         960/700           230772003         ALS         ES1189         1         ALS         ES1189         1	HA209 1.0-1.1	11/07/2003	ALS	ES41010	573					
110/12003         Al.3         Esting         Cold         Octive         Al.3         Esting         Cold	1100003         Alis         Estility         Calibrations         Alis         Estility				-				AT 0 7011010	2000 1 2000	79.10.202
23072003         ALS         E841189 $263$ 23072003         ALS         E841189 $1$ 23072003         ALS         E841189 $1$ 23072003         ALS         E841189 $1$ 23072003         ALS         E841189 $263$ 23072003         ALS         E841189 $263$ 23072003         ALS         E841189 $263$ 23072003         ALS         E841189 $263$ 23077003         ALS         E841189 $266$ 23077003         ALS         E841189 $266$ 23077003         ALS         E841189 $260$ 23077003         ALS         E841189 $266$ 23077003         ALS         E841189 $260$ <td>23077003         ALS         ES41189         4           2307703         ALS         ES41189         1           23077033         ALS         ES41189         1           230772033         ALS         ES41189         1           230772033         ALS         ES41189<!--</td--><td>TP02_1.0-1.1</td><td>11/07/2003</td><td>ALS</td><td>ES41010</td><td>66010</td><td>0010100</td><td>/IIIbiicate</td><td>ALS *ES#1ULU</td><td>nent tobec</td><td>B/7'N1 - 7'I</td></td>	23077003         ALS         ES41189         4           2307703         ALS         ES41189         1           23077033         ALS         ES41189         1           230772033         ALS         ES41189         1           230772033         ALS         ES41189 </td <td>TP02_1.0-1.1</td> <td>11/07/2003</td> <td>ALS</td> <td>ES41010</td> <td>66010</td> <td>0010100</td> <td>/IIIbiicate</td> <td>ALS *ES#1ULU</td> <td>nent tobec</td> <td>B/7'N1 - 7'I</td>	TP02_1.0-1.1	11/07/2003	ALS	ES41010	66010	0010100	/IIIbiicate	ALS *ES#1ULU	nent tobec	B/7'N1 - 7'I
	1 $23072003$ A.I.S         E841189         4rd         N	TP201 0.5-0.6	23/07/2003	ALS	ES41189	263					
230772003         ALS         ES41189 $1$ $   -$	23(0712003         A1S         E841189         1         ·	TP201 1.0-1.1	23/07/2003	ALS	ES41189	484					
230732003         ALS         E841189 $4.00$ $230772003$ ALS         E84189 $<1$ $230772003$ ALS         E84189 $<1$ $230772003$ ALS         E84189 $<1$ $230772003$ ALS         E84189 $<0$ $230772003$ ALS         E84189 $36 230772003         ALS         E84189         96 230772003         ALS         E84189         96 230772003         ALS         E84189         96 230772003         ALS         E84189         90 230772003         ALS         E84189         90 230772003         ALS         E84189         3 230772003         ALS         E84189         3 230772003         ALS         E84189         3 230772003         ALS         E84189         506 230772003         ALS         E84189         50 230772003         ALS         E84189         50 230772003         ALS         E84189   $	2307/2003         ALS         E841189         2400           2307/2003         ALS         E841189         E64100           2307/2003         ALS         E841189         505           2307/2003         ALS         E841189         505           2307/2003         ALS         E841189         505           2307/2003         ALS         E841189         505           2307/2003         ALS         E841189         606           2307/2003         ALS         E841189         506           2307/2003         ALS         E841189         50           2307/2003         ALS         E841189         50           2307/2003         ALS         E841189         50           2307/2003         ALS         E841189         50	TP201_2.0-2.1	23/07/2003	ALS	ES41189	1			-	-	
230712003         ALS         E841189         651% $230712003$ ALS         E841189 $16$ $230712003$ ALS         E841189 $365$ $230772003$ ALS         E841189 $16$ $230772003$ ALS         E841189 $16$ $230772003$ ALS         E841189 $16$ $230772003$ ALS         E841189 $16060$ $230772003$ ALS         E841189 $2601$ $230772003$	23/07/2003         ALS         ES41189         661         1	TP202_0.5-0.6	23/07/2003	ALS	ES41189	2:330					
23/07/2003         ALS         E84/189         <1          <1          <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1	237072003         ALS         E541189         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1         <1	TP202 1.0-1.1	E002/L0/EZ	ALS	ES41189						
23/07/2003         ALS         E841180         505	231072003         ALS         E841180         505         <	TP202 2.5-2.6	23/07/2003	ALS	ES41189	₽					
23/07/2003         ALS         E34/180         C <thc< th="">         C         C</thc<>	230712003         ALS         E341189         -1	TP203_0.5-0.6	23/07/2003	ALS	ES41189	505					
230772003         ALS         E841189         <1           230772003         ALS         E841189         (660)           230772003         ALS         E841189         (660)           230772003         ALS         E841189         (660)           230772003         ALS         E841189         (66)           230772003         ALS         E841189         (70)	230772003         ALS         E641189         C600	TP203 1.0-1.1	23/07/2003	ALS	ES41189	346					
23/07/2003         ALS         E34189         1660           23/07/2003         ALS         E34189         8           23/07/2003         ALS         E34189         8           23/07/2003         ALS         E34189         8           23/07/2003         ALS         E34189         8           23/07/2003         ALS         E34189         956           23/07/2003         ALS         E34189         956           23/07/2003         ALS         E34189         3           23/07/2003         ALS         E34189         3           23/07/2003         ALS         E34189         3           23/07/2003         ALS         E34189         3           23/07/2003         ALS         E34189         2           23/07/2003	2370772003         ALS         E841189         field           239772003         ALS         E341189         8         9           239772003         ALS         E341189         9         95         96           239772003         ALS         E341189         9         95         966         966           239772003         ALS         E341189         9         95         960         966         96           239772003         ALS         E341189         7         9         966         966           239772003         ALS         E341189         2         9         9         9         9           239772003         ALS         E341189         2         9	TP203 2.2-2.3	23/07/2003	ALS	ES41189	⊽					
23/07/2003         ALS         E341189         1660           23/07/2003         ALS         E341189         8         6           23/07/2003         ALS         E341189         651         6         6           23/07/2003         ALS         E341189         651         6         6         6           23/07/2003         ALS         E341189         750         0         6         6         6           23/07/2003         ALS         E341189         750         0         0         1         6         6           23/07/2003         ALS         E341189         750         0         0         1	23/07/2003         ALS         E341189         1060         1           2307/2003         ALS         E341189         8         8         9         2         1	TP204_0.5-0.6	23/07/2003	ALS	ES41189	1660					
237712003         ALS         E341189         8         0         1	237072003         ALS         ES41189         8         0         1	TP204 1.0-1.1	23/07/2003	ALS	ES41189	1080					
23/07/2003         ALS         E841189         2661 (50)         ALS         E841189         2601         field duplicate         ALS-ES 41180         506 <td>23/07/2003         ALS         E341189         661</td> <td>TP204 2.2-2.3</td> <td>23/07/2003</td> <td>ALS</td> <td>ES41189</td> <td>80</td> <td></td> <td></td> <td></td> <td></td> <td></td>	23/07/2003         ALS         E341189         661	TP204 2.2-2.3	23/07/2003	ALS	ES41189	80					
23/07/2003         ALS         E841189         495         QC01         field dupficate         ALS-E8 41189         506           23/07/2003         ALS         E841189         3         QC01         field dupficate         ALS-E8 41189         506           23/07/2003         ALS         E841189         2         3         QC01         field dupficate         ALS-E8 41189         506           23/07/2003         ALS         E841189         2         3         QC01         field dupficate         ALS-E8 41189         506           23/07/2003         ALS         E841189         2	23/07/2003         ALS         E841189         495         QC01         field diplicate         ALS-E8 41189         506           23/07/2003         ALS         E841189         3         QC01         field diplicate         ALS-E8 41189         506           23/07/2003         ALS         E841189         2	TP205 0.5-0.6	23/07/2003	ALS	ES41189	614					
23/07/2003         AL.S         ES41189         3           23/07/2003         AL.S         ES41189         1091           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         500           23/07/2003         AL.S         ES41189         500           23/07/2003         AL.S         ES41189         500           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         2           31/07/2003         AL.S         ES41189         2           31/07/2003         AL.S         ES41146         5           31/07/2003         AL.S         ES41346         57           31/07/2003         AL.S         ES41346         57           31/07/2003         AL.S         ES41346         57           31/07/2003         AL.S         ES41346         57	23/07/2003         AL.S         ES41189         3           23/07/2003         AL.S         ES41189         1090           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         500           23/07/2003         AL.S         ES41189         500           23/07/2003         AL.S         ES41189         500           23/07/2003         AL.S         ES41189         500           23/07/2003         AL.S         ES41189         2           23/07/2003         AL.S         ES41189         2           31/07/2003         AL.S         ES41366         6           31/07/2003         AL.S         ES41366         6           31/07/2003         AL.S         ES41366         6           31/07/2003         AL.S         ES41366         6           31/07/2003         AL.S         ES41366         5	TP205 1.0-1.1	23/07/2003	ALS	ES41189	495	QC01	field duplicate	ALS - ES 41189		(, 2297)
Z307/2003         ALS         ES41189         T590           Z3107/2003         ALS         ES41189         Z           Z3107/2003         ALS         ES41169         Z           Z3107/2003         ALS         ES41169         Z           Z3107/2003         ALS         ES41366         Z           Z3107/2003         ALS         ES41346         J           Z3107/2003         ALS         ES41367         Z	23/07/2003         ALS         ES41189         1290           23/07/2003         ALS         ES41189         2           23/07/2003         ALS         ES41189         2           23/07/2003         ALS         ES41189         500           23/07/2003         ALS         ES41189         500           23/07/2003         ALS         ES41189         500           23/07/2003         ALS         ES41189         2           31/07/2003         ALS         ES41189         2           31/07/2003         ALS         ES41366         5           31/07/2003         ALS         ES41366         5	TP205 2.23	23/07/2003	ALS	ES41189	m					
23/07/2003         A.I.S         ES41189         2           23/07/2003         A.I.S         ES41189         2           23/07/2003         A.I.S         ES41189         500           23/07/2003         A.I.S         ES41189         500           23/07/2003         A.I.S         ES41189         500           23/07/2003         A.I.S         ES41189         2           23/07/2003         A.I.S         ES4139         2           31/07/2003         A.I.S         ES41366         6           31/07/2003         A.I.S         ES41366         15	23/07/2003         A.S.         ES4[189         2           23/07/2003         A.S.         ES4[189         2           23/07/2003         A.S.         ES4[189         500           23/07/2003         A.S.         ES4[189         50           23/07/2003         A.S.         ES4[189         2           23/07/2003         A.S.         ES4[189         2           31/07/2003         A.S.         ES4[139         2           31/07/2003         A.S.         ES4[136         5	20.50 Ancarr	23/07/2003	ALS	ES41189	0661					
Z2107/2003         ALS         ES41189         <1           Z2407/2003         ALS         ES41189         500           Z2407/2003         ALS         ES41189         500           Z2407/2003         ALS         ES41189         510           Z3407/2003         ALS         ES41189         2           Z107/2003         ALS         ES41189         2           Z107/2003         ALS         ES4136         2           Z107/2003         ALS         ES4136         5           Z107/2003         ALS         ES4136         5           Z107/2003         ALS         ES4136         5           Z107/2003         ALS         ES4136         5	Z310712003         ALIS         ES41189         <1           Z310712003         ALIS         ES41189         <1		CONTROLL	ALC	FS41180	6					
Z310712003         ALS         E541180         500           Z310712003         ALS         E541189         500           Z310712003         ALS         E541189         500           Z310712003         ALS         E541189         2           Z310712003         ALS         E341189         2           Z310712003         ALS         E341366         5           310712003         ALS         E341366         5           310712003         ALS         E341366         15	23/07/2003         ALS         ES41189         500           23/07/2003         ALS         ES41189         500           23/07/2003         ALS         ES41189         500           23/07/2003         ALS         ES41189         500           31/07/2003         ALS         ES41189         5           31/07/2003         ALS         ES41366         6           31/07/2003         ALS         ES41366         6	TT-DT DDZJT	EQUICITAL EC	ATE	ECALL 80	V					
2470112003         ALS         ESA1189         2440           23(07)2003         ALS         ES41189         2           31(07)2003         ALS         ES41346         357           31(07)2003         ALS         ES41346         6           31(07)2003         ALS         ES41346         15           31(07)2003         ALS         ES41346         15	Za107/2003         ALS         ES41189         AAA           Za107/2003         ALS         ES41189         2           Za307/2003         ALS         ES41189         2           J107/2003         ALS         ES41366         557           J107/2003         ALS         ES41366         5           J107/2003         ALS         ES41346         15	11200 200 210 210 210 210 2	CUUCIEN/CE	ALC	p541189	0.5					
z4001/2003         ALS         ES41102         Za           23(07/2003         ALS         ES4136         2           31(07/2003         ALS         ES4136         5           31(07/2003         ALS         ES4136         6           31(07/2003         ALS         ES4136         15           31(07/2003         ALS         ES41346         15	z4101/2003         ALS         ES41189         Z           23(07/2003         ALS         ES41345         2           31(07/2003)         ALS         ES41346         6           31(07/2003)         ALS         ES41346         6           31(07/2003)         ALS         ES41346         15           31(07/2003)         ALS         ES41346         15	D.0-C.0 212.11			1041100	2720					
z40072003         ALS         ES41102         2           310772003         ALS         ES41346         57           310772003         ALS         ES41346         15           310772003         ALS         ES41346         15	z4/01/2003         ALS         E341160         2           31/07/2003         ALS         E341346         57           31/07/2003         ALS         E341346         6           31/07/2003         ALS         E341346         15           31/07/2003         ALS         E341346         15	14212 1.0-1.1	CUU2/1/0/62		COLLEGA 100	ł					
31/01/2013         ALS         Estiste         52/           31/07/2003         ALS         Estiste         6           31/07/2003         ALS         Estiste         15	31/01/2003         ALS         Exertise         337           31/07/2003         ALS         E841346         6           31/07/2003         ALS         E841346         15	TP212 1.7-1.8	F007//0/57	ALS	E341165	7 10					
31/072003         ALS         ES41340         0           31/072003         ALS         ES41346         15	31/07/2003         ALS         ES41340         0           31/07/2003         ALS         ES41346         15	TP309_0.5-0.6	31/07/2003	ALS	ES41340	1					
a1/07/2003   ALS   E841346 1200	1 31/072003 1 ALS 1 E341340 1 200	TP309 1.0-1.1	31/07/2003	ALS	ES41346	0					
		TP309 1 5-1 6	31/07/2003	ALS	ES41346	<u>.</u>	And a second of the second of				
						1200		7			

Exceeds NEPM HIL "D" - Investigation levels for land use with minimum opportunity for access to soil and use as open space is something the access to soll and use as open space is a space of the space

Prepared By, NDS Checked By, BG

Page 1 of 4

Audit Damanan Tahlas

Table 2a (Rev 1 - 12 July 06) Soil Analytical Results - Lead Remediation Excavation Validation Lot 202 - Part 2 Validation

	<b>—</b>																					
Čomments		Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Excavation Validation	Eastern property boundary	Eastern property boundary	Eastern property boundary	Eastern property boundary	Eastern property boundary	
ЙРД) (%)			24	5.8								3.5								142		
QC concentration Lead (mg/kg)			14	638								139								96		
Secondary QC Batch No			ALS-ES42626	ALS-ES42626								ALS - ES42668								Amdel - 3E3626		
QC Sample Type			field duplicate	field duplicate								field duplicate								field duplicate		
QC Sample ID	LOR		QC02	aco1					•			QC06								QC05		HIL "E"
Primary Lead Concentration (mg/kg)	•	2	11	00100100000000000000000000000000000000	66	5345545-655545-6454	د د	281	3	535	26	144	151	242-15-902 <u>averation</u>	276	561	4	A. 2010 3790 36 30 36 30 36	24 3990 to 32	557	10	100 Per 100 Per 100
Primary Batch No		ES42626	ES42626	ES42626	ES42626	ES42645	ES42645	ES42645	ES42645	ES42645	ES42668	ES42668	ES42743	ES42743	ES42743	ES42743	ES42626	ES42626	ES42626	ES42645	ES42668	
Primary Laboratory		ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	
Sample Date	lation Samples	31/09/2003	31/09/2003	31/09/2003	31/09/2003	1/10/2003	1/10/2003	1/10/2003	1/10/2003	1/10/2003	2/10/2003	2/10/2003	9/10/2003	9/10/2003	9/10/2003	9/10/2003	31/09/03	31/09/03	31/09/03	1/10/2003	2/10/2003	
Sample ID	Excavation Validation Samples	VAL202	VAL205	VAL206	VAL208	VAL211	VAL212	VAL213	VAL215	VAL216	VAL218	VAL219	VAL221	VAL222	VAL223	VAL224	VAL201	VAL204	VAL207	VAL214	VAL217	

600. Exceeds NEPM HIL, "E" - Investigation levels for land use as open space

Page 2 of 4

,

-------

# <u>Table 2b (Rev I - 12 July 06)</u> Soil Analytical Results - Part 2 Lead Validation Lot 202 - Part 2 Validation

Comment		Previous Investigation	Previous Investigation	Provisional Investigation	Defineation Investigation	Defineation Investigation	Defineation Investigation	Delineation Investigation		Defineation Investigation	Evenation Validation	Excavation Validation	Evention Validation	Freevation Validation	Evenuetion Validation	Excertion Validation	Exceverion Validation	Excavation Validation	Western Part 2 boundary	Western Part 2 boundary	Western Part 2 boundary	Western Part 2 boundary	western Part z boundary	vvestern Part z poundary	Westem Part 2 boundary	western Fait z poundary	Eastern property boundary	Eastern property boundary	Eastern property boundary	Eastern property boundary															
RPD (%)																24	04	0.0							35																			142	
QC concentration Lead (ng/kg)	Į															14	1 063	020					-		139															0 # 0				86	
Secondary GC Batch No																ALC - ECAPEDE		AL.5 - E542020							ALS - FS42668						•								-					Amdel - 3E3626	
QC Sample Type																Eated dunling		field duplicate							field dunlicate															0110hr				field duplicate	
gc Sample (D	<-LOR (mg/kg)												-			0000	מכמק	0001							0008	1000														QC31/ QC32				QC05	-
Primary Lead Concentration (mg/kg)	F	\$	ŝ	9	2	230	5/G	6 007	- -	505	346	Þ	357	g	<del>ئ</del>	54	LL LL	602	00	655	5	281	ä	686	207	424	101 CI	276	561	263 }	484 🛔	1	[1983][1993][1994][1994][1994][1994][1994][1994][1994][1994][1994][1994][1994][1994][1994][1994][1994][1994][19	100 100 100 100 100 100 100 100 100 100	4	12-12-1390-1-1-1	2	চ	250	5	4	54 ST908 State	0666	557	6
Primary Baich No		9505153	9505153	9505153	9505153	ES41010	ES41010		EC41100	EC4100	ES41189	ES41189	ES41346	ES41346	ES41346	ES42626	ES42626	ES42626	ES42626	ES42645	ES42645	ES42645	ES42645	ES42645	ES42666	EC47743	ECA5743	ES42743	ES42743	FS41189	ES41189	ES41189	ES41189	ES41189	ES41189	ES41189	ES41189	ES41189	ES41348	ES41346	ES42626	ES42626	ES42625	ES42645	ES42668
Primary Laboratory		Enviromet	Enviromet	Enviromet	Enviromet	ALS	ALS	ALS	ALG	ALO	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS ALE	VIS	ALS ALS	AIS	AIS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS	ALS
Sample Date		1995	1995	1995	1995	11/07/2003	11/07/2003	23/07/2003	23/07/2003	23/0//2003	20/07/2003	23/07/2003	31/07/2003	31/07/2003	31/07/2003	31/09/2003	31/09/2003	31/09/2003	31/09/2003	1/10/2003	1/10/2003	1/10/2003	1/10/2003	1/10/2003	2/10/2003	2/10/2003	5002/01/R	SUUZUUS	8/10/2013	24/17/20104	23/07/2003	23/07/2003	23/07/2003	23/07/2003	23/07/2003	23/07/2003	23/07/2003	23/07/2003	31/07/2003	31/07/2003	31/09/2003	31/09/2003	31/09/2003	1/10/2003	2/10/2003
Sampte (1)		27-2	Z7-3	Z7-4	Z7-8	HA209_0.6-0.7	HA209_1.0-1.1	TP201_0.5-0.6	TP201_1.0-1.1	TP201 2.0-2.1	1P203 0.5-0.5	TP203 2 2-2 3	TP309_0.5-0.6	TP309 1.0-1.1	TP309_1.5-1.6	VAL202	VAL205	VAL206	VAL208	VAL211	VAL212	VAL213	VAL215	VAL216	VAL218	VALZT8	VALZZT	VALZZZ	VAL223	TDOOL OF AR	TP201 1.0-1.1	TP201 2.0-2.1	TP202 0.5-0.6	TP202 1.0-1.1	TP202 2.5-2.6	TP206_0.5-0.6	TP206 1.0-1.1	TP206 2.0-2.1	TP307 1.0-1.1	TP307 1.5-1.6	VAL201	VAL204	VAL207	VAL214	VAL217

•

Exceeds NEPM HIL "E" - Investigation levels for land use as open space

Audit Response Tables

# Table 3 (Rev 1 - 12 July 06) Soil Analytical Results - Lead Containment Cell Characterisation Results Lot 202 - Part 2 Validation

Sample ID	Sample Date	Laboratory	Batch No	QC Sample ID	Lead (mg/kg)
				LOR->	1
TP02_0.2-0.4	24/07/2002	ALS	ES34741	DUP02	1880
HA201_0.2-0.3	30/04/2003	ALS	ES39497		819
HA201_0.4-0.5	30/04/2003	ALS	ES39497		484
HA202_0.2-0.3	30/04/2003	ALS	E\$39497		749
HA202_0.4-0.5	30/04/2003	ALS	ES39497		2940
HA203_0.4-0.5	30/04/2003	ALS	ES39497		181
HA204_0.2-0.3	11/07/2003	ALS	ES41010		1430
HA204_0.6-0.7	11/07/2003	ALS	ES41010		2460
HA204_1.0-1.1	11/07/2003	ALS	ES41010		5100
HA205_0.2-0.3	11/07/2003	ALS	ES41010		263
HA205_0.6-0.7	11/07/2003	ALS	ES41010		344
HA205_1.0-1.1	11/07/2003	ALS	ES41010		999
HA206_0.2-0.3	11/07/2003	ALS	ES41010		430
HA206_0.6-0.7	11/07/2003	ALS	ES41010	QC12	764
HA206_1.0-1.1	11/07/2003	ALS	ES41010		169
HA207_0.2-0.3	11/07/2003	ALS	ES41010		1460
HA207_0.6-0.7	11/07/2003	ALS	ES41010		1340
HA207_1.0-1.1	11/07/2003	ALS	E\$41010		5130
HA208_0.6-0.7	11/07/2003	ALS	ES41010		3370
HA208_1.0-1.1	11/07/2003	ALS	E\$41010		716
TP02_1.0-1.1	11/07/2003	ALS	ES41010	QC10 / QC11	6600
TP204_0.5-0.6	23/07/2003	ALS	ES41189		1660
TP204_1.0-1.1	23/07/2003	ALS	ES41189		1080
TP204_2.2-2.3	23/07/2003	ALS	ES41189		8
TP205_0.5-0.6	23/07/2003	ALS	ES41189		614
TP205_1.0-1.1	23/07/2003	ALS	ES41189	QC01	506
TP205_2.2-2.3	23/07/2003	ALS	ES41189		3
TP207_0.5-0.6 *1	23/07/2003	ALS	ES41189	QC02/QC03	6460
TP207_1.0-1.1 *1	23/07/2003	ALS	ES41189		1900
TP207_1.5-1.6 *1	23/07/2003	ALS	ES41189		1
TP212 0.5-0.6	23/07/2003	ALS			500
TP212_1.0-1.1	23/07/2003	ALS	ES41189		3730
TP212 1.7-1.8	23/07/2003	ALS	ES41189		2
VAL200	31/09/2003	ALS	E\$42626	·	2420
VAL203	31/09/2003	ALS	ES42626		1470
VAL209	31/09/2003	ALS	ES42626		2270
VAL210	1/10/2003	ALS	ES42645		650
VAL220	2/10/2003	ALS	ES42668		1440
LSP-03 *2	12/08/2003	ALS	ES41573		13
LSP-04 *2	12/08/2003	ALS	ES41573		2200

Notes:

1200

\*1

Exceeds NEPM HIL "D" - Investigation levels for land use with minimum opportunity for access to soil 600 Exceeds NEPM HIL "E" - Investigation levels for land use as open space

Material from TP207 (located in Part 1) was excavated and relocated to the containment cells located in Part 2.

\*2

Stockpile LSP was generated from excavations completed in Part 1 and was relocated to the containment cells located in Part 2.

Table 4 Soil Analytical Results - Part 2 Asbestos Validation Lot 202 - Part 2 Validation

Result	QN	QN	DN	ND	DN	Q
D Sample Description	Mixture of sand, stones, fragments of bitumen and debris	Mixture of sandy soil, stones, plant matter, fragments of plaster, brick and bitumen and debris	Mixture of sand, stones and debris	Mixture of sand, stone, plant matter and debris	Mixture of sand, stones and debris	Mixture of sand, stones and debris
Batch No QC Sample ID					DUP01	
Batch No	ES34741	ES34741	ES37609	ES37609	ES37609	ES37609
Laboratory	ASET	ASET	ASET	ASET	ASET	ASET
Sample Date	23/07/2002	24/07/2002	7/01/2003	7/01/2003	7/01/2003	7/01/2003
Sample ID	TP27_0.3-0.5	TP28_0.3-0.5	AS-Z1-01	AS-Z1-02	AS-Z1-03	AS-Z1-04

ND Not Detected



APPENDIX D Data Quality Objectives

#### 7.0 DATA QUALITY OBJECTIVES CHECKLIST

The Guidelines for the NSW Site Auditor Scheme (2nd edition, April 2006) state that site auditors must check that a consultant has properly addressed and adopted Data Quality Objectives (DQOs) for the investigation or validation program (as outlined in the checklist below), and that the consultant's report includes the following:

- a statement of pre-determined DQOs for field and laboratory procedures, including quantitative DQOs;
- a plan to achieve pre-determined DQOs; and
- procedures to be undertaken if the data do not meet the expected DQOs.

Title of Report: Remediation Validation Report - Part 2 Area

Report Number: J0807

DQO Steps and Outputs	C = Compliance P = Partial Compliance N = Non-Compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NR)
The timing for the various stages of the project must be clearly understood by all parties prior to commencing any work on the project. The DQO process must be commenced before any investigative work begins on the project	С	
Step 1: State the problem		
Has a concise description of the problem been provided?	С	
Has a list of the planning team members and a decision maker been identified?	N	
Is there a summary of available resources and relevant deadlines for the study?	N	
Has a conceptual model of the site based on information prior to the commencement of the investigation/validation been provided?	C	
Step 2: Identify the decisions		
Has a decision statement that links the principal study question to actions that will solve the problem been provided?	С	
Step 3: Identify inputs to decision		
Has a list of informational inputs needed to resolve the decision statement been developed?	С	Reference to information and methodology provided in RAP
Has a list of environmental variables or characteristics to be measured been provided?	Р	ec E6
Is the information required to allow informed decisions to be made to address the decision statements provided?	Р	ec
Have the media (fill, soil, groundwater) to be investigation/validated been identified?	Р	ec ec
Have the criteria for each medium been identified and provided?	Р	Elsewhere in report
Are the analytical methods required for chemicals of concern relative to the site criteria identified?	Р	Elsewhere in report
#### DATA QUALITY OBJECTIVES CHECKLIST 7.0

смја 📕

DQO Steps and Outputs	C = Compliance P = Partial Compliance N = Non-Compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NR)
Have defined concentrations for field screening (and a response if reached) been identified ?	NR	Field screening was not conducted
Has any other information required to make decisions been provided?	Nō	
Step 4: Define the study boundari	es	and an and a subscription of the
Has a detailed description of the spatial and temporal boundaries of the problem been provided?	Ċ	
Have any practical constraints that may interfere with the study been identified?	C	
Step 5: Develop a decision rule		
Have acceptable limits for; chemicals of concern detected in field blanks, recoveries of laboratory spike additions, and RPDs of matrix spike and matrix spike duplicates, been defined?	Р	Provided in QA/QC appendix
Have the statistical parameters of interest that characterise the population been identified (eg 95% UCL)?	С	
Has a statement that the criteria exceed the laboratory reporting limits been provided?	nternet e <b>N</b>	ana na ang sana ang salan na salan na salan na salan sa salan sa sa
Have any contingency measures been developed and provided?	No	
Step 6: Specify limits on decision	errors	
Have decision error rates based on a consideration of the consequences of making an incorrect decision been provided and justified?	N	
Step 7: Optimise the design for ob	taining data	
Was data collection optimised?	С	
Was a sampling analytical and quality plan (SAQP) developed and provided?	Ν	
Does the consultant's site assessment report include a QA/QC narrative describing all information relevant to the site assessment ?	С	

• . •

.



# APPENDIX E Quality Assurance/Quality Control

## 8.0 REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

#### Report Number: J0807 Site Address: Remediation Validation Report - Part 2 Area

This section is based on Appendix V of the Guidelines for the NSW Site Auditor Scheme (2006). It contains the essential issues 'which must be included in the quality assurance program' conducted by a contaminated site consultant during site assessment and remediation processes.

ltem	C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	Comments (comment always required for P and NF		
FIELD				
SAMPLING STRATEGY				
Does the consultant's sampling program include assessment of all relevant environmental media, including:				
Soil	С			
Dust	NR	Not relevant to site or validation		
Surface water	NR	Not relevant to site or validation		
Groundwater	NR	Not relevant to site or validation		
Air	С	Asbestos fibre air monitoring		
Sediments	NR	Not relevant to site or validation		
Biota	NR	Not relevant to site or validation		
Is the sampling strategy clearly defined and justified on the basis of project objectives, site conditions and history?	C			
	ed 🗆 Judgemental 🗆 Con	nbination		
Was the sampling strategy appropriate for the conditions at the site and the nature of the contamination ?	С			
Is the rationale for the strategy described in the consultant's report?	Р	Grid sampling for an excavation is Standard validation strategy		
Does the rationale include:		· · · · · · · · · · · · · · · · · · ·		
Sampling pattern	<u> </u>	As above		
Sampling density	P	Density provided without rationale		
Estimated size of residual hotspots that may remain undetected	С	Elsewhere in report		
Sampling depths	NR	Not necessary for base and walls		
Analytes	Р	Provided without rationale		
Analytical methods	N			
Were all samples analysed for all analytes of concern?	N			
Justification of decisions concerning samples to be analysed and samples not to be analysed	С			
Does the number of sample locations comply with EPA sampling design guidelines?	С	No assessment against EPA guidelines – excavation sampling ;10 x 10 grid – lead and 20 metres apart – asbestos		
Are divergences from guidelines adequately justified?	NR	No relevant to validation sampling		
Is overall coverage of site adequate?	NR	Only excavations required validation		
Are the sampling locations shown on a	C	Not scaled		

#### **REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE** 8.0 AND QUALITY CONTROL

ltem	C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NR)		
scaled site sampling plan ?				
Are sample depths stated?	Р	Investigation sample depths were and validation sample depths were not		
Are borehole/test pit logs provided?	a	For BHs where samples are used for		
		validation data set.		
Was sampling investigation depth sufficient?	С			
Was fill material adequately investigated?	С	a de la francisca de la composición de		
Was number of depth samples sufficient	~			
to give adequate coverage of profile?	C			
Are sample collection, handling and transportation procedures documented and appropriate to meet the project DQOs?	С	As an appendix		
Was sampling representative of site conditions, based on the selection of appropriate number of sampling points and of samples from each relevant strata and material types stated in a site sampling plan to meet the project DQOs?	C	a second a fight diffuier attacks a second		
SAMPLING METHODS				
Are sampling procedures adequately				
described?	<u> </u>			
Are sampling procedures adequate and appropriate for the site?	С			
Was composite sampling used?	No	· · · · · · · · · · · · · · · · · · ·		
Were composite samples laterally adjacent?		· · · · · · · · · · · · · · · · · · ·		
Were composites from the same depth interval?	-			
Were samples for analysis for volatile analytes composited?	-			
Adequate description of investigation / Validation methods?	С			
Sampling equipment description (including drilling plant)	Ν			
Has an assessment of the reliability of field procedures been undertaken by the consultant by using the Data Quality Indicators (DQI) (precision, accuracy, representativeness, completeness and comparability).	Ν	No assessment conducted, the following provides an indication if the required information was provided elsewhere in the report.		
Representativeness				
Tepresentationess				
Were the appropriate media sampled according to SAQP [in this case RAP]?	С			

## 8.0 REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

ltem	C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NR)
Are SOPs appropriate and complied with?	С	
Accuracy (bias)		· ·
Are SOPs appropriate and complied with?	C	
Completeness	L.,	L Martine Martine
Have all critical locations been sampled?	С	
	C	
Have all required samples been collected	C	
(from grid and at depth)?	C	
Are SOPs appropriate?		
Are SOPs complied with?	C	
Was an experienced sampler used on each	С	
occasion?	·	
Is the documentation correct?	С	
Comparability		
Have the same SOPs been used on each occasion?	Unknown	Not stated
Have comparisons been made regarding climatic conditions (temperature, rainfall, wind, etc)	N	
Were the applicability and limitations of field methodology discussed appropriately in the consultant's report?	N	
Has the consultant ensured adequate calibration of instruments?	N	Air monitoring conducted by Sub contractor.
[DEC] Has the consultant's report adequately assessed the significance of the results of field screening methods compared with the results of laboratory analyses, for example that the results reported for field screening using a photo- ionisation detector are compatible with the results reported by the laboratory for volatile organic compounds?	N	
[CMJ] Have the results of screening methods and laboratory analyses been compared and discussed (including an explanation of non-compatibility)?	N	
[CMJ] Where not compatible, has the consultant's report adequately explained this?	NR	All air and soil samples were non detect
Are the applicability and limitations of any screening methodology used by the laboratory appropriately discussed in the consultant's report?	NR	
Is screening method performance known and expressed as a multiple of specific analytical method performance?	Ν	
FIELD QA/QC		
Has a field QA/QC plan been included in	С	

-----

Page 10

#### 8.0 REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

ltem		C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NR)	
	the consultant's report?			•;·· •
	Does the report include details of the sampling team ?	Ν		
	Does the report include details of			
	sampling method(s), including the actual	-	la anna an	
	methods employed for obtaining samples,			
	sample devices and equipment, type(s) of	С		
	sample containers and seal used, order			
	and degree of filling, preservation,			
	labelling, logging, custody?			
	Does the report include details of			
	evidence of appropriate decontamination	С		
	procedures carried out between sampling			
	events?		· · · ·	
	Does the report include details of logs for			
	each sample collected showing time, location, initials of sampler, duplicate			
-	locations, duplicate type, chemical	N	n an Aran an Aran an Aran an Aran an Aran an Aran an an Aran an an an an an an a An an	
	analyses to be performed, site		water a second and the second second second	
	observations and weather conditions?			
	Does the report include details of chain-			
	of-custody documentation fully	·		
	identifying for each sample:	·····		·
	Name of sampler		initials	
	Nature of sample	C		
	Collection date	C		
	Analyses to be performed	C		
	Sample preservation method	C		
	Departure time from the site	<u> </u>		
	Dispatch courier(s)	NR	Were relinquished by URS	
	Condition of samples at dispatch	N		
	Does the report include details of sample		·	
	splitting techniques ?	С		
	Does the report include details of a			
	statement of duplicate frequency for intra-	С		
	laboratory and inter-laboratory duplicate			
ł	samples and duplicate sample results ? Does the report include details of			
	background sample results	Ν	}	
ľ	Does the plan include details of rinsate	~	Provided upon request from Auditor	
	sample results ?	C		
ľ	Does the report include details of			
ļ	laboratory-prepared trip spike results for	NR	Not used	
	volatile analytes ?			
	Does the report include details of trip blank results ?	С	Provided upon request from Auditor	
-	Does the report include details of field instrument calibrations on-site (when		No field screening was conducted by URS	

# 8.0 REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

ltem	C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	Comments (comment always required for P and N		
used)?				
Does the consultant's field QA/QC program include replicate samples split in the field and submitted to two separate laboratories in accordance with the requirements of the National Environment Protection (Assessment of Site Contamination) Measure 1999 ?	С			
LABORATORY				
METHODS AND REPORTING LIMITS	<b>_</b>			
Has an assessment of the reliability of analytical results has been undertaken by the consultant by using the Data Quality Indicators (DQI) (precision, accuracy, representativeness, completeness and comparability).	N	As an assessment of laboratory results and QC tests were not conducted in the form of DQIs, the following is a review of whether an assessment was provided in another form.		
Representativeness				
Have all samples been analysed according to SAQP?	С			
Precision				
Analysis of:				
Field Intra-laboratory and inter- laboratory duplicates with RPDs	С			
Laboratory-prepared volatile trip spikes	NR	Not used – volatile contaminants were not of concern for validation.		
Accuracy (bias)	· · · · · · · · · · · · · · · · · · ·			
Analysis of:				
Field blanks	С			
Rinsate blanks	C			
	NR	Not conducted		
Reagent blanks	C			
Method blanks	C			
Matrix spikes				
Lab duplicates	С			
Matrix spike duplicates	С			
Surrogate spikes	NR	Not relevant for inorganics		
Reference materials (CRM)	NR	Not conducted		
Laboratory control samples	С			
Completeness				
Have all critical samples been analysed according to the SAQP [RAP]?	С			
Have all analytes been analysed according to the SAQP?	NR	No SAQP provided		
Have appropriate methods and PQLs been used?	С			
Is sample documentation complete?	Ň			
Have sample holding times been	С			

\_\_\_\_\_

\_\_\_\_\_

. ....

CMIA

#### 8.0 REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

	Item	C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NR)
	complied with?		
	Comparability		
	Have the same analytical methods been used (including extraction and clean-up)?	P	Only methods confirmed
	Sample PQLs (justify/quantify if different)	С	PQLs the same
	Same laboratories (justify/quantify if different)	C	Different labs used for primary and inter dup, analysis
	Same units (justify/quantify if different)	С	
	Are the analytical methods used for site validation of appropriate precision and accuracy ?	N	Unknown
	Are the sensitivity and selectivity of the analytical methods appropriate for the assessment of the risk?	Ν	Unknown
	Do the precision and accuracy criteria set out in the consultant's QA/QC plan, for a given method and matrix, meet the performance expected of the reference method?	N	
ſ	Has the consultant included in their		······································
-	reports written documentation on quality	······································	
1	of data supplied by the analytical	······································	
	laboratory which meets the objectives of the testing laboratory's quality plan for at least 95% of test results?	N	
t	CONSULTANTS REVIEW OF LA	BORATORY QA/QC	
┢	Does the consultant's report(s) includ		
-	Names of the accredited laboratories used and relevant details of their accreditation	P	No method accreditation verification
L	for each analytical method.		
	A statement that laboratories were accredited for all analyses by the National Association of Testing Authorities (NATA) or an equivalent (government- endorsed provider of accreditation for laboratories).	N	
	A statement that sample analyses use appropriate methodologies for each potential contaminant in the matrix.	N	
	A statement that Practical Quantitation Limits are appropriate for the chemicals of concern for use in the assessment of risk.	N	
	Has a laboratory QA/QC report been p	rovided (by the lab) with	the following information:
-	A copy of signed chain-of-custody forms acknowledging receipt date and time, conditions of samples on receipt and	С	

.

## 8.0 REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

ltem	C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NF		
identity of samples included in shipments.		· · · · · · · · · · · · · · · · · · ·		
Record of holding times and a comparison with method specifications.	Р	No comparison with method specifications		
Analytical methods used.	C			
Laboratory accreditation for analytical methods used.	С			
Laboratory performance in inter- laboratory trials for the analytical methods used, where available.	N			
Acceptance limit(s) for each QC test, such as duplicate relative percentage differences (RPDs) and recoveries for laboratory quality control analyses.	С			
Where used, the origin of certified reference material (CRM), its batch number and the concentrations of the chemicals of potential concern.	NR	Not used		
Results for blind duplicate samples collected from the field.	С			
Description of surrogates and spikes used	С			
Per cent recoveries of spikes and surrogates	С			
Instrument detection limit	N			
Method detection limits	N			
Matrix or practical quantification limits	С			
Standard solution results	N			
Reference (CRM) sample results	Ň			
Daily check sample results	N			
Laboratory (reagent/method) blank results	C			
	N N			
Laboratory standard charts The laboratory specifying compliance with the requirements of the NEPM and equivalence with the reference method or non-standard methods.	N			
QA/QC Documentation	· · · · · · · · · · · · · · · · · · ·			
Does the consultant's site assessment report address all the QA/QC checklist items in the Guidelines for Consultants Reporting on Contaminated Sites (EPA 1997) related to field quality assurance and quality control, laboratory QA/QC and data evaluation QA/QC reporting?	С	Substantially		
Does the consultant's site assessme	ent report include:			
QC results relevant to the sample analysis.	С			
For each sample, the highest	N			

.....

. . . . . . .

------

\_\_\_\_ ....

land po<del>gaga operaties and b</del>iotecome

#### 8.0 REPORTING OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

	ltem	C = Compliance P = Partial Compliance N = Non-compliance NR = Not Relevant	<b>Comments</b> (comment always required for P and NR)	
	measurement result wherever replicate			<u> </u>
	measurements are taken (or all			
	measurement results for each sample).			
	Results for all data tabulated separately			
	according to each type of soil, fill			
÷	materials, groundwaters, surface waters			
	and sediments, with appropriate statistical	NĬ	liller och mader sör i sinder söret.	
	analysis according to the National	14		
·	Environmental Protection (Assessment of			
	Site Contamination) Measure 1999			
	requirements.			

\_\_\_\_\_

#### 9.0 AUDITOR'S REVIEW OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

Is a QA/QC narrative that substantially complies with DEC (2006) and EPA (1997) guidelines included in the report?

Yes	No
1	

#### 9.1 Field Work and Methods

#### Sample Collection

Number of samples collected: Soil: 80 (51 - Delineation (Table 1), +25 - lead excavation (table 2a + excavated val. samples) + 4 - asbestos excavation) Water: 0

Number of days of sampling:

Soil: numerous Water: NA

Number and type of QA/QC	SOIL			WATER			
samples collected:	No.	Frequency	Criterion	No.	Frequency	Criterion	
Inter Laboratory Duplicates	9	<1/10	1/10				
Intra Laboratory Duplicates	2	1/40					
Trip Blanks	1	1/76	Not stated		NA		
Wash Blanks	1	1/76	Not stated				
Other – Trip Spike	0						

Fie	ld duplicates	Contro	l Limits	Yes	No
a.	Were an <u>adequate number</u> of field duplicates collected?			. 🗸	
b.	Were RPDs within control limits?	Min	Max		✓
			<u>+</u> 40%		

**Comments:** Two significant exceedances were recorded for duplicates (lead) one of 200% and one of 142%. The sample location with the 200% RPD was later re-sampled - including a duplicate and a triplicate - which all recorded high concentrations.

These RPDs were attributed to sample heterogeneity.

Trip blanks	Yes	No
a. Were an adequate number of trip blanks collected?		Ń
b. Were the trip blanks free of contaminants? (If no, comment whether the contaminants present are also detected in the samples and whether they are common laboratory chemicals.)	~	
Comments:		

Wash blanksYesNoa. Were an adequate number of wash blanks collected?✓b. Were the wash blanks free of contaminants?<br/>(If no, comment whether the contaminants present were also detected in the<br/>samples and whether they are common laboratory chemicals.)✓Comments:✓Overview: Was field QA/QC satisfactory? Comment as necessary.✓

**Comments:** While insufficient intra laboratory duplicates, wash blanks and trip blanks were prepared/collected, the Auditor considers the overall sampling QA/QC to be adequate.

CM

#### 9.0 AUDITOR'S REVIEW OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

#### 9.2 Sample Handling

[	a.	Are COC forms provided and complete?		· · · · · · · · · · · · · · · · · · ·	<b>.</b>
	b.	Were sample receipts provided?	√		
<u></u>	Ċ,	Did they indicate - proper custody between the field and the laboratory?			
	d.	- adequate sample preservation and condition?	1		
	e.	- that the sample holding times were met?	<ul> <li>✓</li> </ul>		
Į					

a se al successo en se se successo a completar a analizada en al activitada en al seconda en al completar en a

Comments

#### 9.3 Laboratory

#### 9.3.1 Methods and Reporting Limits

	Yes	No
1. Was a NATA-registered laboratory used?		
2.Did reports have the NATA seal and a signature?	. 1	
3. Were laboratory methods identified?	1	
4. Did the laboratory perform the requested tests?	√	
5. Were PQLs/MDLs/LORs for each analyte/matrix combination given?	✓	
6.Are any non-standard/non NATA endorsed methods justified?	NA	
7. Were the reporting limits (LOR) satisfactory?	✓	
8.Were the appropriate test procedures followed?	✓	

#### 9.3.2. Internal Quality Control Procedures

Type and number of QA/QC		SOIL		
	No.	Frequency	Criterion	
Method Blanks/Reagent Blanks	16	<1/20	1/10	
Matrix Spikes/Matrix Spike	14	<1/20 -	1/20	
Duplicates				
Laboratory Control Samples/	17 -	<1/20	1/20	
Certified Reference Material	1	-		
Analysis				
Laboratory Duplicates	9	<1/10	1/10	
Surrogates	1/Org	1/Organic	1/Organic	
		sample	sample	

		Control Limits		Yes	No	
		Min	Max		(Comment Below)	
2.	Were the method/reagent blanks free of contamination?	<pql< td=""><td></td><td><ul> <li>✓</li> </ul></td><td></td></pql<>		<ul> <li>✓</li> </ul>		
3.	Were the spike recoveries within control limits?	70%	130%	~		
4.	Were the RPD's of the laboratory duplicates within control limits?	-30%	+30%	1		
5.	Were the surrogate recoveries within control limits?	70%	130%	$\checkmark$		
б.	Were the origin and batch number of certified reference			$\checkmark$		
7.	Were the Laboratory Control samples within limits?	70%	130%	✓		
8.	Are all QC results provided?	1		$\checkmark$		
9.	Was the overall standard of Laboratory QA/QC adequate?					

Comments:

. .

CMIA

#### 9.0 AUDITOR'S REVIEW OF FIELD & LABORATORY QUALITY ASSURANCE AND QUALITY CONTROL

#### 9.4 DATA USABILITY

Are the field and laboratory analytical data provided of adequate quality for the purpose of this audit?

Yes	No.
$\checkmark$	

Comment further as necessary.

Date Auditor



# APPENDIX F Environmental Management Plan

# FINAL REPORT

Environmental Management Plan Part 2, Lot 202 King Street, Randwick



C/- McLachlan Lister Pty Ltd Level 1, 1 Hickson Road The Rocks, NSW, 2000

21 September 2006 43346065



URS Australia Pty Ltd Prepared By: Level 3, 116 Miller Street Lachlan Wood North Sydney, NSW 2060 Australia Project Engineer

Peer Reviewed & Authorised By:

Francene Mitchell

Principal Engineer

Tel: 61 2 8925 5500 Fax: 61 2 8925 5555

Date:	-21 Septembe	97-2000	)	 	 	 	 	1
Reference:	43346065							
Status:	Final							
A PARTY AND A P	· ·							

1	Intro	oduction	14 او او ها ها ها های او و ها های او او ها ها های او ها ها ها ها و های او های و های او ها های های های های ها و	11
	1.1	Introduction		1-1
	1.2	EMP Objective		1-1
	1.3	EMP Scope		1-1
	1.4	EMP Implementation		1-2
2	Site	Description and Background	به ها به ها ها بر حلول همین از این هم می بر بر بر بولول از با با در هم هو فاقل با به	2-1
	2.1	Site Location and Setting		2-1
	2.2	Site History		2-1
	2.3	Environmental Setting		2-1
		2.3.1 Site Topography & Description		2-1
		2.3.2 Geology		2-2
		2.3.3 Hydrogeology & Hydrology		2-2
	2.4	Previous Environmental Investigations		2-2
	2	2.4.1 Identified Contaminants of Concern		2-3
	2,5	Summary of Remediation Works		2-3
	2,5	2.5.1 Asbestos Remediation		2-3
		2.5.2 Preparation of Containment Cells		2-4
		2.5.3 Lead Remediation Excavation	-	2-4
		2.5.4 Lead Impacted Validation Sampling		2-5
		2.5.5 Site Surface Cover		2-5
3	Ris	k Management Considerations		3-1
	3.1	Nature of Contaminants		3-1
	3.1 3.2	Identification of Contamination		3-1
	2.4	3.2.1 Exposure Pathways		3-1
	3.3	Location of Contaminated Materials		3-1
4	Fut	ure Site Management	ر بر ها ها ها الله الله الله الله الله الله	4-1
	4.1	Maintenance of the Concrete Capping	• • •	4-1
	4.2	Intrusive Site Works - Preliminaries	,	4-1
	4.3	Intrusive Site Works - Earthworks		4-2
	4.4	Reinstatement		4-2
		4.4.1 Environmental Controls		4-2
	4.5	Health and Safety	•	4-3
	4.6	Reporting Intrusive Site Works Activities		4-4
	4.7	Contingency and Emergency Response Planning	· .	4-4
5	Re	ferences	محمد فالافاذ جرم عدم بروان یا جرم بر بر بر این از از از از مربع مربع عدم بر مربع مربع مربع مربع مربع مربع مربع	5-1
6	Lin	nitations	مروحا والمحمد والمحالية و	6-1

URS

i

# List of Tables, Figures, Plates & Appendices

_	
Tables In Tex	t References and extension and the second s
Table 1 Table 2	EMP Implementation Responsibilities Contingency Planning
Tables	
Table 3 Table 4	Containment Cell Soil Characterisation Analytical Results – Lead Soil Validation and Characterisation Analytical Results – Lead
Figures	
Figure 1 Figure 2 Figure 3	Site Location Plan Containment Cell Locations and Proposed Future Surface Coverings Lead Remediation Excavation and Validation/Characterisation Sample Locations
Appendices	
Appendix A	Bulk Earthworks Design Drawing
Appendix B	Plates of Containment Cell
Appendix C	Registered Survey Drawings
Appendix D	Legislation and Guidelines

.

ii

## Introduction

#### 1.1 Introduction

This Environmental Management Plan (EMP) has been prepared by URS Australia Pty Ltd (URS) on behalf of Sir Moses Montefiore Jewish Home (Montefiore) for the redevelopment of Part 2 of Lot 202 King St, Randwick (the site). URS has received instruction in the completion of the project from McLachlan Lister Pty Ltd (McLL) who was engaged by Montefiore to Project Manage the redevelopment works.

This EMP has been prepared to manage the potential risks to human health and the environment posed by lead impacted fill materials, which remain in-situ at the site following the completion of remediation works.

This EMP is strictly for use while the proposed site surface structures and land use remain unchanged. Upon the disturbance or removal of the surface structures, further remediation works may be required. Use of the site for purposes other than those stated would require assessment of the risks associated with those purposes, and may require management measures additional to those described in this EMP.

# 1.2 EMP Objective

The objective of the EMP is to provide a framework for the management of contaminated materials that have been retained on the site following the completion of remediation and validation works.

It is envisaged that the EMP will be referenced during proposed future activities undertaken on the site that are likely to expose residual impacted materials. Montefiore as the owner of the site will be responsible for implementing and administering the EMP.

## 1.3 EMP Scope

The scope of this EMP includes:

- Site description, including environmental setting and structures on the site;
- Summary of the site history and a background of previous environmental investigations at the site;
- Summary of identified soil contamination on the site;
- Summary of remediation works that have been completed at the site; and
- The development of procedures and protocols for the management of contaminated soils, which may be encountered during future intrusive works at the site. These include procedures for current and future site occupants/workers to ensure contact with the contaminated fill and/or soils during maintenance of the site is in accordance with suitable occupational, health and safety (OH&S) and environmental controls. This will include the identification of potential occupant/worker exposure

## Introduction

#### SECTION 1

pathways and methods for minimising occupants/worker's exposure to the contaminated fill and/or

soils

#### 1.4 EMP Implementation

It is the intention that the implementation of this EMP will be the responsibility of Montefiore, as the site owner, and should be enforced by Randwick City Council (Council). It is recommended that a Public Positive Covenant under Section 88E of the Conveyancing Act 1919 is made to the Property Title, nominating Randwick Council as the Prescribed Authority. In addition, it is also considered appropriate that Council place a notation on the Planning Certificate for the site under Section 149 (2) of the Environmental Planning and Assessment Act 1979 (EP&A Act). This notation would address the restrictions for intrusive works on the site. Where a development application (DA) is required, Council will also be able to enforce the implementation of the EMP, through the conditions of consent for the DA. Intrusive works conducted on the site shall only be conducted by contractors/individuals who have read and acknowledged understanding of this EMP.

The following table provides a summary of the responsibilities for the implementation and management of the EMP. These responsibilities do not replace any other regulatory responsibilities of the parties in undertaking works at the site.

Organisation	Responsibilities
Randwick City Council	<ul> <li>Note EMP on appropriate planning instrument.</li> </ul>
Montefiore	<ul> <li>Organise revision of Property Title to note that the site is subject to implementation of this EMP.</li> </ul>
	<ul> <li>Nominate a representative of Montefiore to undertake annual inspections of the capping area. Document and report the inspections to the Montefiore Board on an annual basis.</li> </ul>
	Notify Council of any non-conformance with the EMP and corrective actions.
,	Maintain survey plans of excavations and as-built drawings.
	<ul> <li>Provide Council with an annual statement of compliance with the EMP.</li> </ul>

and the second second second				 
		5	· · · · · · · · · ·	 
Table 1 EMP	Implementation	Responsibilities		
	-			



## 2.1 Site Location and Setting

The site is located at the corner of King Street and Dangar Street, Randwick, NSW and is known as Part 2 of Lot 202 in Deposited Plan (DP) 879576. The site location is shown in Figure 1. Lot 202 was divided into Part 1 and Part 2 for the purposes of remediation and validation in order to allow for the adoption of a staged approach to the development of the aged care facility. Part 1 was previously remediated and validated in 2003 (URS, 2003). The site is within the Local Government Area of Randwick.

The site occupies approximately 2293  $m^2$  in the south-east corner and much of the eastern boundary of Lot 202. The site is currently vacant, however further development is proposed as part of the construction of the aged care facility across Lot 202. The proposed development will include some portions of open space,, including soft landscaping and paved areas. Concrete slabs will also be constructed above the lead impacted materials that have been placed within the containment cells. The proposed future site layout and land use are shown in Figure 2.

## 2.2 Site History

A detailed site historical study was conducted by Doring (1990). The historical data provided in the Doring study indicated that the eastern portion of the original depot, including both Lots 201 and 202, was used for tram and bus mechanical maintenance. Lot 201 is located directly to the west of Lot 202. The study indicated that the area initially commenced operations as a tramway workshop and depot in 1881. The bus depot was operated by the NSW State Transit Authority (STA).

Lots 201 and 202 were divested by the STA in 1998. Sir Moses Montefiore Jewish Home (Montefiore) purchased Lot 202 in 1998. At the time of acquisition, the Site was vacant following partial demolition works and a number of derelict buildings remained on the Site.

## 2.3 Environmental Setting

## 2.3.1 Site Topography & Description

Prior to the remediation works on Part 1, the ground gently sloped in a general westerly direction across Lot 202. During occupancy of the site by STA, a constructed cutting was located in the eastern third of Lot 202 to improve the access for the bus depot operations. This cutting was reinforced by the placement of a concrete retaining wall, which effectively divided the Site into a higher eastern section and a lower western section.

Prior to the 2002/2003 demolition conducted by Montefiore as part of the site preparation works, four brick buildings remained on Lot 202, mostly located in the north east corner and concrete slabs covered some areas. A program of demolition and bulk earthworks was conducted as part of the overall program

# **Site Description and Background**

involving the Part 1 remediation works. The site levels now reflect the required development levels as indicated in the earthworks design drawing provided with Appendix A.

#### 2.3.2 Geology

Across most of Lot 202, investigations indicated a layer of fill, resulting from previous Site activities. This fill varies in depth from 100mm to up to several metres. The fill is generally sandy and similar to the underlying natural material.

Beneath the fill, the Site and surrounding area is underlain by unconsolidated sand deposits of Quaternary age. These sands form part of the Botany Basin deposits. On a regional level the depth of these unconsolidated deposits vary greatly in thickness.

The region is underlain by bedrock units of the Hawkesbury Sandstone Formation (1:250 000 Geological Series Sheet S1 56-5, Sydney). The Hawkesbury Sandstone Formation is dominated by medium to coarse grained quartz with very minor shale and laminate lenses.

#### 2.3.3 Hydrogeology & Hydrology

The Botany Sands, which underlie the site, contain a system of unconfined and semi-confined aquifers that are referred to as the Botany Aquifer. Water in this aquifer is utilised for irrigation in the region. Water in the region flows in a south westerly direction (westerly locally) and the nearest receptors of this water are irrigation wells which are located at the Randwick Racecourse.

## 2.4 Previous Environmental Investigations

The Former Randwick bus depot has been the subject of several site contamination investigations conducted between 1991 and 1995 by both Sinclair Knight Partners (SKP) and Dames & Moore (D&M) as outlined below:

- Sinclair Knight Partners (SKP, 1991) Preliminary Geotechnical and Environmental Investigation of the State Transit Authority's Randwick Bus Depot;
- Dames & Moore (D&M, 1991) Feasibility/Design Study for a Site Remediation Program at the Randwick Bus Depot;
- D&M (1992) EPA Compliance Report, Site Remediation Program: Phase I, Randwick Bus Depot;
- D&M (1994) EPA Compliance Report, Initial Site Validation Program, Randwick Bus Depot; and
- D&M (1995) EPA Compliance Report, Final Site Validation Program, Randwick Bus Depot.

Demolition works were completed by STA in 1997 to allow for subdivision of the depot property into Lots 201 and 202. The 1997 demolition work allowed access to remaining parts of the Site for further



## Site Description and Background

validation assessment, which had not been validated in 1995. This second assessment (D&M (1998) EPA. Compliance Report Post Demolition Site Validation Program Lot 202 King and Dangar Streets Randwick) was completed during 1998.

URS were engaged by Montefiore to conduct further remediation, validation and reinstatement works on the site during 2003. These works are summarised in Section 2.5.

#### 2.4.1 Identified Contaminants of Concern

In the D&M reports from 1994-1998, the various contaminants investigated at Lot 202 have included volatile halogenated compounds (VHCs), cyanide, sulphates, phenols, organochlorine and organophosphate pesticides (OC/OPs), PCBs, metals (including cadmium, chromium, copper, nickel, lead, zinc and mercury), arsenic, total petroleum hydrocarbons (TPH), monocyclic aromatic hydrocarbons (MAHs including benzene, toluene, ethylene and xylene (BTEX)), and polynuclear aromatic hydrocarbons (PAHs). These were selected based on site history including bus and tram operations and maintenance.

The contaminants of concern for the Lot 202 ESA (URS, 5 December 2002) were selected as follows: heavy metals from paint residues and fill materials; TPH/BTEX from bus operations such as refuelling and fuel storage on site; PAHs from fill encountered throughout the Site; and asbestos from demolition activities.

Analytes such as VHCs, OC/OPs, cyanide, sulphates and phenols were not considered to be contaminants of concern as previous results for these analytes were either below detection limits or below the relevant guidelines.

Based on the investigation results, the main contaminants of concern specific to the Part 2 remediation works were lead and asbestos.

## 2.5 Summary of Remediation Works

#### 2.5.1 Asbestos Remediation

Manual collection of asbestos containing materials (ACM) from identified locations was undertaken under the supervision of Hibbs & Associates Pty Ltd (Hibbs). The manual collection of the ACM was undertaken by placing the fragments in 200-micron thick polythene bags or were wrapped in polythene sheeting and placed in trucks or waste bins for off-Site disposal to a licensed landfill facility.

Excavation of demolition fill materials containing ACM was completed to a depth of between 300 mm and 1.0 m along the eastern boundary adjacent to Danger Street. The volume excavated was approximately 600 m<sup>3</sup>. The excavation continued to below the fill materials and was terminated on the natural Botany sands.

## **Site Description and Background**

Following the excavation of asbestos fill materials, Hibbs completed a detailed visual inspection and verified the successful removal of ACM. URS conducted surface soil sampling on a 20 m by 20 m sampling grid on the base of the excavation footprint and submitted the soil samples for laboratory examination of asbestos fibres. No asbestos fibres were identified by the laboratory in the soil samples submitted.

## 2.5.2 Preparation of Containment Cells

Prior to the commencement of remediation works within the lead impacted area, two containment cells were prepared to contain the lead impacted fill material. Preparation of the containment cell areas was completed by lining the excavation with a geofabric marker layer for easy identification in the future. Refer Appendix B for photographic plates of the cell under construction. Following placement of lead impacted material in the cells, another layer of the geofabric marker layer was used to mark the top of the containment cell.

The cells were located in the south western comer and eastern side of the site (refer Figure 2). The cell locations were recorded by survey (attached as Appendix C). The cells were strategically located in areas proposed for building footprints or for paving, thus minimising opportunity for uncontrolled access to contaminated soil. The impacted material was placed at an elevation above the zone of measured groundwater influence. Groundwater level monitoring was completed by Jeffery and Katauskas (J&K). The J&K letter reports (4 August 2003 and 18 July 2003) indicated that the standing groundwater level in the south-east of Lot 202 was approximately 38 m AHD, although it is noted that a 1-2 m fluctuation in level is possible in the Botany Basin. The impacted materials were placed above 40 m AHD to allow for possible fluctuations in the groundwater level.

## 2.5.3 Lead Remediation Excavation

The impacted fill material was excavated and loaded directly from the bucket of the excavator to the bucket of the front-end loader, which then transported it directly to one of the containment cells. A water cart was present throughout the excavation and loading processes, to provide a fine spray of water to control the generation of dust, which may have contained elevated concentrations of lead.

Validation sampling was undertaken of the excavation to verify that all lead impacted material had been removed. Where a validation sample indicated that concentrations of lead in the fill materials exceeded the adopted site remediation guideline level further excavation was undertaken to remove this material.

The final excavation was generally rectangular in shape, with some protrusions due to the "chasing" of contamination within the excavation (Figure 3). The excavation was approximately 49 m in length, with a maximum width of 9 m. The depth of the excavation, which was extended to encounter the natural Botany sands, ranged from approximately 0.9 m to 1.3 m. Survey data defining the location of the remediation excavation has been included in Appendix C.

「中国市場の事品業は最近な市場のないないないではなどのです」」の

All material excavated from the lead impacted area was retained on Site, in the containment cells. The analytical results for the samples taken to characterise the impacted material placed in the containment cells is presented in Table 3.

#### 2.5.4 Lead Impacted Validation Sampling

The base of the excavation in the lead impacted area was sampled on an approximate 10 m by 10 m grid. Samples were also collected from the walls of the excavation, where the grid lines intersected the walls. All samples collected were analysed for total lead concentrations. The analytical results for the excavation validation and characterisation samples are presented in **Table 4**.

URS validated all final excavation surfaces in accordance with the RAP modification (URS, 26 August 2003).

#### 2.5.5 Site Surface Cover

A series of development works associated with the construction of the aged care facility are proposed for the site, including construction of paved landscaping, installation of footpaths and grassed areas and placement of garden beds. In the areas immediately above the containment cells, concrete slabs will be placed to restrict access to the material contained within the cells. The concrete slabs will provide a physical barrier to prevent inadvertent contact between users of the site and the impacted materials.

The slabs consist of 120 mm thick reinforced concrete. It is understood that these slabs will eventually be paved on completion of the landscaping works in Part 2.

The proposed future site layout, including the locations of the concrete slabs and containment cells, is shown in Figure 2. No soft landscaping with access to soil is expected in the areas directly above the containment cells.

2-5

# **Risk Management Considerations**

This section provides summary information on the physical form and some of the health risk considerations for the chemical substances that may be present in the residual soil / fill materials which remain on site in the containment cells.

#### 3.1 Nature of Contaminants

The main chemical of concern for the site is lead. The primary potential exposure routes for lead are inhalation of contaminated dusts or ingestion of contaminated soils.

## 3.2 Identification of Contamination

The investigations and remediation works undertaken at the site have indicated that the potentially contaminated residual fill and soil materials that remain in-situ may generally be identified by the presence of sandy fill.

#### 3.2.1 Exposure Pathways

Workers involved in disturbing potentially contaminated fill / soil material could be exposed to the chemicals of concern by:

- Respiration/inhalation of dust generated from fill/soil materials;
- Ingestion of airborne fill/soil materials through poor hygiene practices (i.e. eating or drinking during work activities, not washing hands before eating, etc.); and
- Possible secondary exposure from potentially contaminated equipment and clothing.

Planning of intrusive works involving exposure and/or disturbance of potentially contaminated fill / soil materials and implementation of appropriate health and safety measures will minimise the potential for worker contact with potentially impacted materials through the above listed exposure pathways.

## 3.3 Location of Contaminated Materials

Based on the analytical results of the characterisation samples collected during the remediation validation works, some residual impacted materials remain in-situ along :

- the northern portion of the eastern boundary of the site; and
- the middle section of the western boundary of the site.

The locations of the characterisation samples that exceeded the adopted remediation guideline levels are shown in Figure 3. Additionally impacted materials have been placed within the containment cells located on the western boundary of the site.

The locations of the containment cells are shown in Figure 2.



#### SECTION 4

This section describes the processes that should be considered prior to undertaking any intrusive site works in the vicinity of the containment cells. It addresses scenarios where minor intrusive works may be required in the vicinity of the containment cells such as for installation or repair of underground services or for landscaping works. It also includes recommendations on the ongoing maintenance and monitoring of the concrete cap located above the containment cells.

It does not address more substantial intrusive works, which may be required for construction activities. Where demolition/construction works are required, these works will be undertaken in accordance with the conditions of a Development Consent issued by Randwick City Council.

All works should be undertaken in accordance with current regulatory requirements. A summary of the main legislation, planning instruments and guidelines that relate to the management of contaminated land in NSW at the time of preparation of the EMP is provided in Appendix D.

The advice of suitably qualified environmental consultant and/or Council should be sought where there is any uncertainty as to the regulatory requirements.

#### 4.1 Maintenance of the Concrete Capping

Periodic inspections of the containment cells should be undertaken to ensure that the integrity of the capping is maintained. It is anticipated that this inspection would take place once every 12 months.

During the proposed construction works for the development of the site care should be taken to ensure that the works do not result in any damage to the capping slabs. If damage or deterioration does occur the slabs should be repaired to meet the original specifications.

Until completion of construction works on site, it is anticipated that the concrete capping may either be exposed at the surface or alternatively, buried underneath soft landscaping.

If exposed, it will be directly inspected by the designated Montefiore representative.

If buried beneath soft landscaping, the nominated Montefiore representative will inspect that the ground in this area has not been disturbed.

Inspections shall involve recording the condition of the capping areas by making observations of the appearance of cracks, subsidence or unauthorised excavations. Photographic evidence will be documented for each inspection. Findings of the inspections will be communicated to the Montefiore Board on an annual basis.

In the long term, the concrete capping will be below hard landscaping.

## 4.2 Intrusive Site Works - Preliminaries

Prior to exposing and/or disturbing site fill materials in the vicinity of the containment cells by intrusive excavation works, the following activities should be carried out and implemented:



#### SECTION 4

Scope detail of proposed works so as to minimise the requirement to expose and/or excavate site fill
 /soil materials;

Ensure workers are aware of the potential for contaminated fill / soil materials on site;

Preparation of a health, safety and environmental plan and safe work method statements to address
the proposed activities/works, including the provision of personal protective equipment and
environmental controls to minimise potential for exposure; and

Preparation of a methodology for managing excavated materials (eg waste disposal requirements).

Should the works involve substantial excavation, the advice of Council and /or a suitably qualified environmental consultant should be sought.

## 4.3 Intrusive Site Works - Earthworks

During excavation works, care should be taken to separate the fill/soil materials that are suspected to be contaminated. Decontamination of equipment that has been in contact with contaminated materials may be required.

Excess spoil generated from intrusive works that is either contaminated or cannot be reinstated within the original excavation area should be assessed for off-site disposal in accordance with NSW DEC requirements. This may require the materials to be assessed, classified and disposed off-site in accordance with the NSW DEC's *Environmental Guidelines: Assessment, Classification & Management of Liquid & Non-Liquid Wastes* (2004) (the 'Waste Guidelines'). This may require collection of samples for laboratory analysis by a suitably qualified environmental consultant. The responsibility for ensuring such material has been assessed and disposed correctly, in accordance with current regulations and guidelines, remains with Montefiore as the site owner.

## 4.4 Reinstatement

Following completion of intrusive excavations, the excavation shall be backfilled with excavated materials or with virgin excavated natural materials (VENM) sourced from off-site.

If removal of part of the concrete capping is required to undertake the intrusive works, this should be reinstated in accordance with the original specifications in order to maintain the integrity of cap and prevent access to the impacted materials.

## 4.4.1 Environmental Controls

Intrusive excavations should be appropriately managed such that there are no off-site impacts associated with contaminated stormwater, which may come into contact with contaminated soil. In addition surface



water should be appropriately managed such that the health and safety considerations of handling potentially contaminated water are considered.

If required, provisions for temporary storage of excess spoil in an environmentally responsible manner prior to disposal must be arranged. This may include:

- Placement of material on a sealed or plastic lined surface with appropriate environmental controls including control of sediment (eg silt fences, bunds) and dust (eg covering surfaces and/or wetting of material); or
- Placement in suitably labelled secure drums/skip bins.

Surface water shall generally be controlled and prevented from being impacted by contaminated soils, by intercepting and re-directing runoff in a controlled manner by any appropriate means, including but not limited to the use of temporary bunds, diversion drains, ditches, straw bales and siltation fences.

Groundwater is not expected to be encountered during intrusive excavations as it is expected to be at a depth of approximately 38 m AHD, however if encountered it should be assessed, removed and disposed of in accordance with relevant regulatory requirements.

All relevant approvals, including but not limited to Trade Waste Agreements, waste facility disposal approvals etc, should be obtained prior to disposal of materials.

#### 4.5 Health and Safety

An Occupational Health, Safety and Environment Plan (HSEP) should be prepared by the party conducting intrusive works on site. This would be expected to cover all aspects of the management of health and safety issues associated with the works to be undertaken.

Some of the main issues that require consideration during the preparation of the plan include, but are not limited to:

- Nature of the work being undertaken;
- Hazards associated with the task including potential chemicals of concern;
- Control measures which may include managing potential human and environmental exposure to the chemicals of concern;
- Personal protective equipment;
- Approvals and guidelines; and
- Work methods.

#### **SECTION 4**

#### 4.6 Reporting Intrusive Site Works Activities

The following documentation should be maintained during intrusive excavation works:

- Daily logs documenting the location and quantities of materials excavated, the amount reinstated and the amount removed for off-site disposal and/or treatment;
- Records of the assessment and classification of materials and all relevant approvals and dockets for off-site disposal and/or treatment;
- Documentation relating to the volumes and source of reinstatement materials including, where required, relevant documentation relating to the verification of virgin excavated natural material (VENM) used for backfill;
- Details, including testing where required, of compaction of backfill; and
- Details of reinstatement of surface cover.

Contraction of the second s

This information should be incorporated by Montefiore into the EMP for future site management.

## 4.7 Contingency and Emergency Response Planning

Conditions encountered during the intrusive works on site are uncertain. Should unexpected conditionsoccur at the site, the contingency and emergency measures listed in **Table 2** should be followed.

ANTICIPATED PROBLEM	CORRECTIVE ACTION
Excessive Dust	Use water sprays, or cease dust-generating activities until better dust control can be achieved.
Suspected Asbestos Containing Materials Encountered	Stop works. Handle and dispose in accordance with WorkCover requirements by appropriately licensed contractor. Controlled wetting and/or covering may be employed to reduce asbestos dust emission by suitably trained personnel.
Release of fuel/oil from equipment	Remove source, use absorbent booms to remove oil, make any repairs as required.
Excessive Odour	Reduce the area of contamination exposed by covering the excavation face, or application of odour suppressing agents or cease work until better odour control can be achieved. Reschedule the work program to take into account prevailing wind conditions.

#### Table 2. Contingency Planning

おいたは世界になるないないので、「ない」のないでは、

12411111

l

#### **SECTION 4**

ANTICIPATED PROBLEM	CORRECTIVE ACTION		
Potentially Contaminated Fill / Soil Encountered	Undertake works in accordance with Health, Safety and Environment Plan (HSEP).		
	Replace materials following completion of works or dispose off site		
	If previously unidentified conditions are encountered assessment and classification by a suitably qualified environmental consultant may be required.		
Oily materials	If oily materials are encountered, the workers should cease work immediately and await direction from the site supervisor or a suitably qualified environmental consultant as to the appropriate course of action.		
Contaminated Water Encountered	Contain water and following assessment, as necessary, on site treatment and/or off site treatment or disposal to a suitably licensed facility or disposal to stormwater/sewer in accordance with relevant regulatory requirement The advice of a suitably qualified environmental consultant may be required		
Emergency Maintenance Works Required	Ensure that no direct contact is made with the soil and that no dust is generated during works. Notify Montefiore and Council of situation as soon as possible.		

If conditions encountered differ from those anticipated the site owner should be notified. Occurrence of excessive odours or worker discomfort should be immediately reported and works discontinued.

4-5

UR

ANZECC/ARMANC 2000, Australian Water Quality Guidelines for the protection of Aquatic
 Ecosystems

# ::

- Dames & Moore, EPA Compliance Report, Initial Site Validation Program, Randwick Bus Depot for State Transit Authority (22 July 1994).
- Dames & Moore, EPA Compliance Report (Volumes 1 and 2) Final Site Validation Program, Randwick Bus Depot for State Transit Authority (4 July 1995, Revision 3).
- Doring, C. 1990. History of the Former Randwick Tramway Workshops. Volumes 1 and 2. Prepared for the Urban Transit Authority of NSW and The Heritage Council of NSW and The Heritage Council of NSW.
- Jeffrey & Katauskas Pty Ltd, Groundwater Investigation letter reports, 18 July 2003 and 4 August 2003;
- National Environmental Protection Council Service Corporation, December 1999, National Environmental Protection (Assessment of Site Contamination) Measure (NEPM)
- NSW DEC 2004, Environmental Guidelines : Assessment, Classification & Management of Non-Liquid and Liquid Wastes

NSW EPA 1994, Contaminated Sites: Guidelines for Assessing Service Station Sites

- NSW EPA 1995, Contaminated Sites: Sampling Design Guidelines
- NSW EPA 1997, Contaminated Sites: Guidelines for Consultants Reporting on Contaminated Sites
- NSW DEC 2006, Guidelines for the NSW Site Auditor Scheme, 2<sup>nd</sup> edition.
- URS, 5 December 2002, Data Assessment Report, Lot 202 King St, Randwick (Final Report)
- URS, 7 February 2003, Remediation Action Plan, Lot 202 King St, Randwick (Final)
- URS, 26 August 2003, RAP Modification Proposed Lead Remediation (letter)
- URS, 18 September 2003. "Remediation Validation, Part 1 Area Lot 202 King Street, Randwick," No: 51072-001-558.
- US EPA Contract Laboratory Program 1999 National Functional Guidelines for inorganic Data Review, Office of Emergency and Remedial Response, US EPA
- US EPA Contract Laboratory Program 2001 National Functional Guidelines for Organic Data Review, Office of Emergency and Remedial Response, US EPA

## Limitations

「日本のないない」というないで、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」の

記録をになったのできる

#### SECTION 6

URS Australia Pty Ltd (URS) has prepared this report for the use of Montefiore in accordance with the usual care and thoroughness of the consulting profession. It is based on generally accepted practices and standards at the time it was prepared. No other warranty, expressed or implied, is made as to the professional advice included in this report. It is prepared in accordance with the scope of work and for the purpose outlined in the Proposal dated 5 June 2006.

The methodology adopted and sources of information used by URS are outlined in this report. URS has made no independent verification of this information beyond the agreed scope of works and URS assumes no responsibility for any inaccuracies or omissions. No indications were found during our investigations that information contained in this report as provided to URS was false.

This report was prepared between June and September 2006 and is based on the conditions encountered and information reviewed at the time of preparation. URS disclaims responsibility for any changes that may have occurred after this time.

This report should be read in full. No responsibility is accepted for use of any part of this report in any other context or for any other purpose or by third parties. This report does not purport to give legal advice. Legal advice can only be given by qualified legal practitioners.



.

. .

. .

.

J:\JOBS\43346065\EMP\RANDWICK EMP FINAL\_DOC\20-SEP-06



#### Table 3 Soil Analytical Results - Lead Containment Cell Characterisation Results Lot 202 - Part 2 Validation

Sample ID	Sample Date	- Laboratory	Batch No	QC Sample ID	
				LOR	
TP02_0.2-0.4	24/07/2002	ALS	ES34741	DUP02	1880
TP02_1.0-1.1	24/07/2002	ALS	ES41010	QC10/QC11	6600
HA201_0.2-0.3	30/04/2003	ALS	ES39497		819
HA201_0.4-0.5	30/04/2003	ALS	ES39497		484
HA202_0.2-0.3	30/04/2003	ALS	ES39497		749
HA202_0.4-0.5	30/04/2003	ALS	ES39497		2940
HA203_0.4-0.5	30/04/2003	ALS .	ES39497	_{	181
HA204_0.2-0.3	11/07/2003	ALS	ES41010		1430
HA204_0.6-0.7	11/07/2003	ALS	ES41010		2460
HA204_1.0-1.1	11/07/2003	ALS	ES41010		5100
HA205_0.2-0.3	11/07/2003	ALS	ES41010		263
HA205_0.6-0.7	11/07/2003	ALS	ES41010		344
HA205_1.0-1.1	11/07/2003	ALS	ES41010		999
HA206_0.2-0.3	11/07/2003	ALS	ES41010		430
HA206_0.6-0.7	11/07/2003	ALS	ES41010	QC12	764
HA206 1.0-1.1	11/07/2003	ALS	ES41010		169
HA207_0.2-0.3	11/07/2003	ALS	ES41010		1460
HA207_D.6-0.7	11/07/2003	ALS	ES41010		1340
HA207_1.0-1.1	11/07/2003	ALS	ES41010		5130
HA208_0.6-0.7	11/07/2003	ALS	ES41010		3370
HA208_1.0-1.1	11/07/2003	ALS	ES41010		716
TP02_1.0-1.1	11/07/2003	ALS	ES41010	QC10/QC11	6600
TP204_0.5-0.6	23/07/2003	ALS	ES41189		1660
TP204_1.0-1.1	23/07/2003	· ALS	ES41189		1080
TP204_2.2-2.3	23/07/2003	ALS	ES41189		8 .
TP205_0.5-0.6	23/07/2003	ALS	ES41189		614
TP205_1.0-1.1	23/07/2003	ALS	ES41189	QC01	506
TP205_2.2-2.3	23/07/2003	ALS	ES41189		3
TP207_0.5-0.6 *1	23/07/2003	ALS	ES41189	QC02/QC03	6460
TP207_1.0-1.1 *1	23/07/2003	ALS	ES41189		1900
TP207_1.5-1.6 *1	23/07/2003	ALS	ES41189		.1
TP212 0.5-0.6	23/07/2003	ALS	ES41189		500
TP212 1.0-1.1	23/07/2003	ALS	ES41189		3730
TP212 1.7-1.8	23/07/2003	ALS	ES41189		2
VAL200	31/09/2003	ALS	ES42626		2420
VAL203	31/09/2003	ALS	ES42626		1470
VAL209	31/09/2003	ALS	ES42626		2270
VAL210	1/10/2003	ALS	ES42645		650
VAL220	2/10/2003	ALS	ES42668		1440
LSP-03 *2	12/08/2003	ALS	ES41573		13
LSP-04 *2	12/08/2003	ALS	ES41573		2200

Notes:

<del>\*</del>1

\*2

Material from TP207 (located in Part 1) was excavated and relocated to the containment cells located in Part 2.

Stockpile LSP was generated from excavations completed in Part 1 and was relocated to the containment cells located in Part 2.

竹田と、小小山子 あ

#### Table 4 Soil Analytical Results - Part 2 Lead Validation Lot 202 - Part 2 Validation

日本での方法になる。これである。「日本での方法」である。

Sample ID	- Sample Date	Laboratory	Batch No	QC Sample ID	Lead (mg/kg)	Comment
				LOR		
· Z7-2	1995	Environnel	9505153	· · · · · · · · · · · · · · · · · · ·	~5	Previous Investigation
Z7-3	1995	Enviromet	9505153		<5	Previous Investigation
Z7=4	1995	Enviromet	9505153		6	- Previous Investigation
Z7-8	1995	Enviromet	9505153	. et e na star e e e e e e e e e e e e e e e e e e e	5	Previous Investigation
HA209_0.6-0.7		ALS	ES41010		230	Delineation-Investigation
HA209_1.0-1.1	11/07/2003	ALS	E\$41010		573	Delineation Investigation
TP201_0.5-0.6	23/07/2003	ALS	ES41189	· · · · · · · · · · · · · · · · · · ·	263	Delineation Investigation
TP201_1.0-1.1	23/07/2003	ALS	ES41189		484 -	-Delineation Investigation
TP201_2.0-2.1	- 23/07/2003 -	ALS	ES41189		1	Delineation Investigation
TP202_2.5-2.6	23/07/2003	ALS	ES41189		<1	Delineation Investigation
TP203_0.5-0.5	23/07/2003	ALS	ES41189		505	Delineation Investigation
TP203_1.0-1.1	23/07/2003	ALS	ES41189		346	Delineation Investigation
TP203_2.2-2.3	23/07/2003	ALS	ES41189		<1	Delineation Investigation
TP309_0.5-0.6	31/07/2003	ALS	ES41346		357	Delineation Investigation
	31/07/2003	ALS	ES41346		6	Delineation Investigation
TP309_1.5-1.6	31/07/2003	ALS	E\$41346		15	Delineation Investigation
VAL202	31/09/2003	ALS	ES42626		2	Excavation Validation
VAL205	31/09/2003	ALS	ES42626	QC02	11	Excavation Validation
VAL206	31/09/2003	ALS	ES42626	QC01	》的《中602 <del>第三章</del>	Excavation Validation
VAL208	31/09/2003	ALS	ES42626		66	Excavation Validation
VAL211	1/10/2003	ALS	ES42645		655 2674	Excavation Validation
VAL212	1/10/2003	ALS	ES42645		<	Excavation Validation
VAL213	1/10/2003	ALS	ES42645		281	Excavation Validation
VAL215	1/10/2003	ALS	ES42645		3	Excevation Validation
VAL216	1/10/2003	ALS	ES42645		535	Excavation Validation
VAL218	2/10/2003	ALS	ES42668		26	Excavation Validation
VAL219	2/10/2003	ALS	ES42668	QC06	144	Excavation Validation
VAL221	9/10/2003	ALS	ES42743		151	Excavation Validation
VAL-222	9/10/2003	ALS	ES42743	+++++	<b>第4字》902</b> 字《读	Excavation Validation
VAL223	9/10/2003	ALS	ES42743		276	Excevation Validation
VAL224	9/10/2003	ALS	ES42743		561	Excavation Validation
Boundary Chara	cterisation	· - ·				
TP202_0.5-0.6	23/07/2003	ALS	ES41189		進一 12330 5 9 年	Boundary Characterisation (West)
TP202_1.0-1.1	23/07/2003	ALS	ES41189		631	Boundary Characterisation (West)
TP206_0.5-0.6	23/07/2003	ALS	ES41189		ar 1.1390 - P	Boundary Characterisation (West)
TP206_1.0-1.1	23/07/2003	ALS	ES41189		2	Boundary Characterisation (West)
TP206 2.0-2.1	23/07/2003	ALS	ES41189	·····	<1	Boundary Characterisation (West)
VAL201	31/09/2003	ALS	ES42626		4	Boundary Characterisation (West)
VAL204	31/09/2003	ALS	ES42626		945 <b>3790</b> 4 40	Boundary Characterisation (East)
VAL204	31/09/2003	ALS	ES42626		3990	Boundary Characterisation (East)
VAL207	1/10/2003	ALS	ES42645	QC05	557	Boundary Characterisation (East)
VAL214 VAL217	2/10/2003	ALS	ES42668		10	Boundary Characterisation (East)
1 */11-51/	2/10/2003	<u></u>	LOAYOOO		600	Doundary onalablensation (East)

600 Exceeds NEPM HIL "E" - Investigation levels for land use as open space


. . .

J:UOBS\43346065\EMP\RANDWICK EMP FINAL.DOC\20-SEP-06

URS







## Appendix A Bulk Earthworks Design Drawing





•						· .
·						
	·	ETOUT TAB	·····	-111	,23 vorie skali be	EDENT_CONTROL_H suively conted cut in acce
	POINT	EASTING	NORTHIN		(λ) Local ณหิดส) (6) DPA – Parib	hy nephanesla, ian anskral manural far unbor:
	2	321903,15 321986.61	1245811.78		Urben Erman	l sanumation and land mar n à Secimant Control".
	3	321976.60 321975,31	1246734.17 1246732.35	<sup>2</sup>	the more of the	eal control drawings and ma works. Should the Contracto my require to be modified.
J.	5 8	321950.60 321950.91	1246735,17 1246738,15		Adding the second s	e le be approved by life rd actional control elon sheet b
,	7	321892,87 321993,03	1245747.08 1246797.88	1 1	scories to meet t Nontain of property	he mying silvoitons or ye in statement control devic
NECOSCIE DO DE CONTINUES	9 10	322001.93 \$21995.67	1246795.50 1246762,51	1.	the stormale a	al are considered entrologic
NAMES HOW AND ADDRESS OF ADDRESS HOW ADDRESS	11 12	322010.01 322005.33	1246760,44 1245730,20	5.	Ninimise the area of Project of standard	af shi being skrivebad al pa af shi being skrivebad al pa
AN ATH A MAX COM DAL HEADS	13 14	321983.09 322008,45 322032,69	1246733,58 1246795,49		alacipit koose mel Techniczanaus	leid in readways, near drai
-710	15 16 17	322032.59 322022.77 322011.85	1245791.64		We so and solar a De out al soch ve conditions.	anivel measures are to be exists day, and movilled to
	15 19	322017.54 (	1246729.19 1246765.96 1246758.02	11	Central water from enter the distanced	
	20 21	322016.91 321991.70	1246719.76	11	lensorory construct	ide shell enter and wit lh llen miny wit
	22 23	321987,38 322015,09	1246703,63 1246699,91	+ I	terra.	he sile shot be cleaned on her sport and pills clear of
	24 25	322012.58 321979.11	1246692.22 1246598.04		station to L. Inspect of Storm strend.	lenarativ system and dear
	25 27	321983,14 321952,40	1246723 <i>.5</i> 5 1246728.30	1 12	ીસ્ત્ર છો, તી સરસંસ ગોલ્યા શબ્દો.	a sond accolonge and a compared a constrained according to the constraint of the con
	26 29	321941.94 321963,82	1246661.25 1246660.87	1	<b>nace () Viela</b> Price la communición	mi of eccentrices the follow
Ĩ,	30 31	321966,48 322009,35	1246678.05 1246671.42	11	monogernent device Construct all fances	s much be installed. I belose life and across
	32 33 34	321895.64 321907.53	1245764.95 1245763.14	12	ભગવી ગીસ્ટ. દિલ્હાભદી ໂલાફલવાગુ મંત્રીલીર લ્લાફેલ ગુર્ગ	enshuction entry/cail on
	35	321904.76 321895.30 321905.02	1246745.24	111	Construct, measures Flormvolar system.	le divert upstraczn Jans in
;	37	321902,59	1246761.31 1246745.58	14.	Construct andmenia markets	lion trops/basin including o
•				- LL	Construct but land Provide sandbag sed Construct production	ander Men by State and Stat State and State and St
				11	es they are construe In completion all pon	ener pil sumaund eround of clied romonl provide singid bog ke
				_   4.8	trops croconel pille Trovide and momenta alter the construction	e strip of birt on holds sid
				-	•	
<u> </u>				LA	bult exclined and	aul ten uni lines 1020.
				1 2 11	squit e le zallad en laisleas beixeu	al 2 (H) : 1 (Y) U.H.O. If be tased as a tracking (1)
· .					ն) ես բիշատողերդ	he specification requirement adduct control, compliant will termine, and allows filling
					prostructed is not	andonce with the specifical
				1		abynde te noi less there
;					ecolian	Slanderil d
·.			•		nder þuijdfing slobs i	(AS 1285
·				1 4	nder mods and eary andreased or eas	atr S
CEDADARE FINA Fined to Finite-	~	-508.)	<b>1</b> 21	,1 Be	kee placing fill, proo	l tol exposed subgrade will subgrade and thest remove
3 x 2.5 mm x 1 100 canes		Ĥ	1	(a	res with more than A spois to be repla	Jana movement under (ali end with select DIU).
		1	\$	1 10	cardonce with Idena	salely barriers ancumi execu al salely regulations, lapart prepared by ~
	<u> </u>			- Jel;	wy and Kalanston J 1716752m/22-oct	Ply Lid
	1	1	1 2	HUK	EARTHNORKS	 IFTEND
and General Land	200 200	۷ <u> </u>	1		<u>, , , , , , , , , , , , , , , , , , , </u>	Boller
אבאר אין באפא אלארביר אין אנארביר אין ג אנאראר אין אנארביר אין אנאר		4		Piate	H Hinte I nuo	hute Eurthmorke Sleep [Sleep horn low side lo
SILTATION	EEUro	INCT IN				Bulk worthworks continuer Bulk worthworks continuer
	LE L 24			Đ	22.00	bulk earthworks pholloca
				ER0S	- On and seden	ent control, 1997)
						Biller
				x		" Silation fonce Slammator pit with .
				.		Slammaler pit silk Geolexile filer sumeand Noy bole komiers
•					Å.	sandhag sectment itop
				+		Calch drain
:				4 4	······································	Sociel dilar Linit of eartherarist
					DUAR HEILICED IS	okme A
,	•				09403	्म म.
•				A DOLE	NIK DECER	<u>г рж</u> г
.•				180-101		·
						NTEFIORE JEXT . NTIAL AGED C/
Th PC PCID PL				FACE	<u>. YTL</u>	
TO BE READ IN CO TREE PROTECTION	JUN .	painuu –			EARTHIN	
LAVIANCE LA LA	Č	2 002			ært and Rol plan	
			'	het fleel		
				une tr	ISP REFE PONT P	ALKOS AROTTI
FOR CON	- <u></u>	1107101	{	TAVL		ARCHI ENGLEN
				×	)55 LTD, hereforeted	)語 われる ACH, N1 21
WORKS OF	SITE SH	וסטנם אמי		PHONE S	CHAPOUS STRUCT	TELEBAL & CELE ME
COMMENCE CONTAMINATION IN ACCORDANC	WINDOWS	L REMEDIAL	- L - Ł	icaic AS SK		The w
			-			

## Appendix B Plates of Containment Cell

J:\JOBS\43346065\EMP\RANDWICK EMP FINAL.DOC\20-SEP-06



URS

# Appendix C Registered Survey Drawings

J:\JOBS\43346065\EMP\RANDWICK EMP FINAL.DOC\20-SEP-06



•		*		•	•		
		· · · · · · · · · ·	·····	••••	· · · · ìì	-017/59	140
		•		4			*
					<u> </u>		• •
-	· ·			-	•		
						·	
•		POINT 1 2 3 4 5 6 7 4 5 6 7 8 C D E	3370 3372 3372 3370 3370 3370 3370 3370	TING 58,68 31,98 08,58 31,60 48,90 52,30 55,82 27,21 16,19 03,50 05,32 95,38 94,56 58;24 58;24 58,49 50,55	6246 6246 6246 6246 6246 6246 6246 6246	28 THING 924.26 899.03 736.20 5761.69 880.90 904.73 904.23 865.86 867.35 774.42 763.29 764.10 759.12 762.24 765.29 766.68 743.54	
E] 1613 1638 2726	45	C	4911 P	DIATIO	N BOUN	3576	



J. A. BRADSHAW CIVIL CONTRACTING

A. BRADSHAW

- .<u>...</u>

Pty Etd ABN 33 106 837 736 7 / 20 Foundry Road Seven Hills NSW 2147 PO Box-224, Seven-Hills NSW 1730 Telephone: (02) 9674 1055 Facsimile: (02) 9674 1099 Email: admin@jabradshaw.com.au

CIVILCAD 5 Job 03030 Date 19/02/04 File C:\CCAD5\TEMP\03030.OUT

Job details 09/04/03 KING STREET RANDWICK

•				
Number	Easting	Northing	RL	
178	322020.26	1246691.24	44.300	
179	322017.32	1246691.57	44.395	· · · · · · · · · · · · · · · · · · ·
180	322015.83	1246694.05	44.340	·
181	322014.00	1246699.18	44,230	
182	322013.52	1246705.46	44.161	
183	322016.52	1246706.99	44.181	
184	322020.55	1246706.31	44.400	
185	322021,33	1246699.06	44.400	
186	322021.04	1246692.05	44.281	
187	322018.62 .	1246695.58	44.450	
188	322017.73	1246704.31	44.220	
189	322026.34	1246718.83	-44.600	
190	322020.99	1246719.22	44.530	
_191	322021.61	1246732.76	44.600	
	322024.24	1246733.03	44.450	
193	322026.18	1246740.87	44.700	
194	322022.68	1246748.97	44.600	
195	322023.75	1246756.07	44.600	•
196	322029.90	1246769.95	44.556	
197	322034.88	1246769.64	45.160	

- Trickow

# Appendix D Legislation and Guidelines

J:WOBS\43346065\EMP\RANDWICK EMP FINAL.DOC\20-SEP-06

山市市は南部市市の市市市市市の市市市市市

### Appendix D Legislation and Guidelines

The Acts and guidelines listed below are current at the time of reporting.

Contaminated Land Management Act 1997 (CLM Act): contains a comprehensive legislative regime for the identification, assessment and management of contaminated sites.

Contaminated Land Management Regulation 1998 (CLM Regulation) contains minor matters incidental to the operation of the CLM Act such as; EPA rates of recovery and the EPA notification form.

State Environmental Planning Policy No 55 Remediation of Land (SEPP 55) provides a framework for planning decisions relating to the remediation of contaminated land in NSW.

Managing Land Contamination – Planning Guidelines Prepared by the Planning NSW and Department of Environment and Conservation (DEC) in conjunction with SEPP 55 provides guidance for planning authorities in NSW to make decisions in relation to contaminated land.

Guidelines on Significant Risk of Harm from Contaminated Land and the Duty to Report (the Significant Risk Guidelines) provides guidance to stakeholders and the general public on the processes the EPA uses to assess risk as well as the duty to report when contamination presents a significant risk of harm to human health or the environment as defined in the CLM Act.

*Protection of the Environment Operations Act* 1997 grants the DEC a number of powers in which to regulate and licence pollution.

Local Government Act 1993 Section 124 empowers local councils to order owners, occupiers and other persons conducting activities on land to 'not conduct' or 'cease conducting' an activity if it constitutes or is likely to constitute a life threatening hazard or threat to public health or safety.

Australian and New Zealand Guidelines for Fresh and Marine Water Quality 2000 Australian and New Zealand Environment and Conservation Council and Agriculture and Resource Management Council of Australia and New Zealand (ANZECC and ARMCANZ). This document provides a summary of the water quality guidelines designed to protect and manage the environmental values supported by water resources and gives advice on designing and implementing water quality monitoring and assessment programs.

National Environmental Protection (Assessment of Site Contamination) Measure 1999, National Environmental Protection Council (NEPC, 1999) establishes a nationally consistent approach to the assessment of site contamination to provide protection to human health and the environment.

NSW EPA (now incorporated within the Department of Environment and Conservation (DEC)) Guidelines provide assistance for landowners, developers, site auditors or the general public, in interpreting various aspects, obligations and duties associated with the contamination investigation and remediation process. Some of the main guidelines that relate to contaminated sites include:

- Contaminated Sites: Guidelines for Assessing Service Station Sites (1994);
- Contaminated Sites: Sampling Design Guidelines (1995);

### Appendix D Legislation and Guidelines

- a contra c

.....

22

- Contaminated Sites: Guidelines for Consultants Reporting on Contaminated Sites (1997);
- Contaminated Sites: Guidelines for the NSW Site Auditor Scheme (2006), and
- Environmental Guidelines: Assessment, Classification & Management of Liquid & Non-liquid
  Wastes (2004)

J:\JOB\$\43346065\EMP\RANDWICK EMP FINAL.DOC\20-SEP-06

an ann ar cliaig

URS Australia Pty Ltd ABN 46 000 691 690 Level 3, 116 Miller Street North Sydney, NSW 2060 Australia Tel: 61 2 8925 5500 • Fax: 61 2 8925 5555

規算に引きる

諸国語語の意味に知る語

時に変更したななななないなどである